Regency Energy Partners LP Form SC 13D/A May 13, 2010

UNITED STATES
SECURITIES AND EXCHANGE COMMISSION
Washington, D.C. 20549
SCHEDULE 13D
Under the Securities Exchange Act of 1934
(Amendment No. 6)*
Regency Energy Partners LP

(Name of Issuer) **Common Units**

(Title of Class of Securities) 75885Y 10 7

(CUSIP Number)
Regency LP Acquirer, L.P.
c/o GE Energy Financial Services
General Electric Capital Corporation
General Electric Company
800 Long Ridge Road
Stamford, Connecticut 06927
Telephone: (203) 961-5963
Attn: General Counsel
Attn: Portfolio Manager

(Name, Address and Telephone Number of Persons Authorized to Receive Notices and Communications)

May 10, 2010

(Date of Event which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§ 240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box, o

*The remainder of this cover page shall be filled out for a reporting person s initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter the disclosures provided in a prior cover page.

The information required in the remainder of this cover page shall not be deemed to be filed for the purpose of Section 18 of the Securities Exchange Act of 1934 (<u>Act</u>) or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

CUSIP No. 75885Y 10 7

NAMES OF REPORTING PERSONS

1.R.S. Identification Nos. of Above Persons (Entities Only)

Regency LP Acquirer, L.P.

CHECK THE APPROPRIATE BOX IF A MEMBER OF A GROUP

2

(a) þ

(b) o

SEC USE ONLY

3

SOURCE OF FUNDS

4

NA

CHECK IF DISCLOSURE OF LEGAL PROCEEDINGS IS REQUIRED PURSUANT TO ITEMS $2(\mbox{d})$ OR $2(\mbox{e})$

o

CITIZENSHIP OR PLACE OF ORGANIZATION

6

Delaware

SOLE VOTING POWER

7

NUMBER OF - 0 -

SHARES SHARED VOTING POWER

BENEFICIALLY 8

OWNED BY 24,679,577

EACH SOLE DISPOSITIVE POWER

9

Table of Contents

3

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A REPORTING **PERSON** - 0 -WITH SHARED DISPOSITIVE POWER 10 24,679,577 AGGREGATE AMOUNT BENEFICIALLY OWNED BY EACH REPORTING PERSON 11 24,679,577 CHECK IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES 12 o PERCENT OF CLASS REPRESENTED BY AMOUNT IN ROW (11) 13 26.48% TYPE OF REPORTING PERSON 14 PN (Limited Partnership)

2

CUSIP No. 75885Y 10 7

NAMES OF REPORTING PERSONS

1.R.S. Identification Nos. of Above Persons (Entities Only)

General Electric Capital Corporation

CHECK THE APPROPRIATE BOX IF A MEMBER OF A GROUP

2

(a) þ

(b) o

SEC USE ONLY

3

SOURCE OF FUNDS

4

NA

CHECK IF DISCLOSURE OF LEGAL PROCEEDINGS IS REQUIRED PURSUANT TO ITEMS $2(\mbox{d})$ OR $2(\mbox{e})$

О

CITIZENSHIP OR PLACE OF ORGANIZATION

6

Delaware

SOLE VOTING POWER

7

NUMBER OF - 0 -

SHARES SHARED VOTING POWER

BENEFICIALLY 8

OWNED BY 24,679,577

EACH SOLE DISPOSITIVE POWER

9

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A REPORTING **PERSON** - 0 -WITH SHARED DISPOSITIVE POWER 10 24,679,577 AGGREGATE AMOUNT BENEFICIALLY OWNED BY EACH REPORTING PERSON 11 24,679,577 CHECK IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES 12 o PERCENT OF CLASS REPRESENTED BY AMOUNT IN ROW (11) 13 26.48% TYPE OF REPORTING PERSON 14 CO

3

CUSIP No. 75885Y 10 7

NAMES OF REPORTING PERSONS

1.R.S. Identification Nos. of Above Persons (Entities Only)

General Electric Company

CHECK THE APPROPRIATE BOX IF A MEMBER OF A GROUP

2

(a) þ

(b) o

SEC USE ONLY

3

SOURCE OF FUNDS

4

NA

CHECK IF DISCLOSURE OF LEGAL PROCEEDINGS IS REQUIRED PURSUANT TO ITEMS $2(\mbox{d})$ OR $2(\mbox{e})$

o

CITIZENSHIP OR PLACE OF ORGANIZATION

6

NEW YORK

SOLE VOTING POWER

7

NUMBER OF - 0 -

SHARES SHARED VOTING POWER

BENEFICIALLY 8

OWNED BY 24,679,577

EACH SOLE DISPOSITIVE POWER

9

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A REPORTING **PERSON** - 0 -WITH SHARED DISPOSITIVE POWER 10 24,679,577 AGGREGATE AMOUNT BENEFICIALLY OWNED BY EACH REPORTING PERSON 11 24,679,577 CHECK IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES 12 o PERCENT OF CLASS REPRESENTED BY AMOUNT IN ROW (11) 13 26.48% TYPE OF REPORTING PERSON 14 CO

4

TABLE OF CONTENTS

Item 4. Purpose of Transaction

Item 6. Contracts, Arrangements, Understandings or Relationships With Respect to Securities of the

[ssuer

Item 7. Material to be filed as Exhibits

SIGNATURES

Table of Contents

This Amendment No. 6 to Schedule 13D (this **Amendment**) amends the Schedule 13D initially filed on June 28, 2007, the Amendment No. 1 to Schedule 13D filed on December 14, 2007, the Amendment No. 2 to Schedule 13D filed on January 10, 2008, the Amendment No. 3 to Schedule 13D filed on April 21, 2008, the Amendment No. 4 to Schedule 13D filed on March 11, 2009 and the Amendment No. 5 to Schedule 13D filed on September 4, 2009 (collectively, the **Schedule 13D**).

General Electric Company, a New York corporation (GE), General Electric Capital Corporation, a Delaware corporation (GECC) and Regency LP Acquirer, L.P., a Delaware limited partnership (LP Holdings , and together with GE and GECC, the Reporting Persons) are filing this Amendment to disclose that Regency GP Acquirer LP, a Delaware limited partnership (GP Holdings), Energy Transfer Equity, L.P. a Delaware limited partnership (ETE) and ETE GP Acquirer LLC, a Delaware limited liability company (ETE GP Buyer) have entered into an agreement for the sale by GP Holdings of (i) 100% of the membership interests in Regency GP LLC, a Delaware limited liability company (RGPLLC); and (ii) 99.999% of the limited partner interest in Regency GP LP, a Delaware limited partnership (RGPLP), to ETE GP Buyer, in exchange for approximately \$300 million in convertible preferred units of ETE (the GP Transfer), with the rights and preferences set forth in Amendment No. 3 to the Third Amended and Restated Agreement of Limited Partnership of ETE. GP Holdings is an indirect subsidiary of GECC although current and former management of Regency Energy Partners LP, a Delaware limited partnership (the Issuer) own a minority of its equity. RGPLP is the General Partner of Issuer.

The closing of the GP Transfer is subject to amendments to the credit agreements of both Issuer and ETE as well as customary closing conditions. Also, expected to close immediately after the GP Transfer is the Contribution Agreement, dated May 10, 2010 (the **Contribution Agreement**), among Issuer, Regency Midcontinent Express LLC, a wholly owned subsidiary of Issuer (**Regency Midcon**), and ETE, pursuant to which Regency Midcon will acquire from ETE membership interests in Midcontinent Express Pipeline LLC in exchange for Common Units. On May 11, 2010, Issuer filed a Current Report on Form 8-K to describe the transactions contemplated by the Contribution Agreement, which is also filed as an exhibit thereto.

The Reporting Persons are also filing this Amendment to disclose the sale on April 30, 2010 by EFS Haynesville LLC, a Delaware limited liability company and an indirect, wholly owned subsidiary of GECC (**EFS Partner**), of 76,989 units representing general partner interests in RIGS Haynesville Partnership Co., a Delaware general partnership (the **Haynesville General Partnership**), to Regency Haynesville Intrastate Gas LLC, a Delaware limited liability company (**RHIG**), an indirect, wholly owned subsidiary of Issuer, in exchange for \$92,086,857 in cash.

EFS Partner will continue to own 151 units representing general partner interests in the Haynesville General Partnership, which represents 0.0137% of the Haynesville General Partnership interests. EFS Partner has agreed to vote the general partner interests as directed by RHIG.

5

Table of Contents

Except as provided herein, this Amendment does not modify any of the information previously reported on the Schedule 13D. Capitalized terms used and not otherwise defined have the meaning given to them in the Schedule 13D.

The Reporting Persons have entered into a Joint Filing Agreement, dated the date hereof, a copy of which is filed with this Amendment as Exhibit A (which is hereby incorporated by reference) pursuant to which the Reporting Persons have agreed to file this statement jointly in accordance with the provisions of Rule 13d-1(k)(1) under the Act. Information with respect to each Reporting Person is given solely by such Reporting Person, and no Reporting Person assumes responsibility for the accuracy or completeness of the information furnished by another Reporting Person.

Item 4. Purpose of Transaction.

Item 4 is hereby amended by adding the following:

The following describes plans or proposals, including those relating to the contemplated GP Transfer, that the Reporting Persons may have with respect to the matters set forth in Item 4(a)-(j) of Schedule 13D:

- (a) Although the consummation of the GP Transfer will result in a change of control of Issuer, the 24,679,577 Common Units are not part of the transaction and upon closing, the Reporting Persons will continue to be the beneficial owners of such Common Units. LP Holdings and Issuer intend to enter into a registration rights agreement at the closing of the GP Transfer pursuant to which LP Holdings will have the right to cause Issuer to prepare and file a registration statement to permit the resale of Common Units held by LP Holdings from time to time as permitted by Rule 415 promulgated under the Securities Act of 1933, as amended (the **Securities Act**). If the closing of the GP Transfer occurs and LP Holdings either exercises its right to have a registration statement filed or does so pursuant to an exemption from registration, it may, from time to time, dispose of a portion or all of its Common Units based on market and general economic conditions, the business affairs and financial condition of Issuer, the market price of the Common Units and other factors deemed relevant by LP Holdings.
 - (b) None.
 - (c) None.
- (d) Upon the closing of the GP Transfer, the owners of RGPLP will have the right to designate the board of directors of RGPLLC (the **RGPLLC Board**); provided however, at the closing of the GP Transfer, LP Holdings, RGPLP, RGPLLC and ETE intend to enter into an Investor Rights Agreement pursuant to which LP Holdings would be granted the contractual right, but not the obligation, to either (i) designate up to two directors to the RGPLLC Board or (ii) designate up to two persons to attend all meetings of the RPGLLC Board, solely in the capacity of a non-voting observer. These rights would terminate if specific ownership thresholds are not maintained by LP Holdings.

(e) None.

6

Table of Contents

- (f) None.
- (g) None.
- (h) None.
- (i) None.
- (j) Except as described in this Item 4, the Reporting Persons do not have, as of the date of this Amendment, any plans or proposals that relate to or result in any of the actions or events specified in clauses (a) through (i) of Item 4 of Schedule 13D.

$Item\ 6.\ Contracts, Arrangements, Understandings\ or\ Relationships\ With\ Respect\ to\ Securities\ of\ the\ Issuer.$

Item 6 is hereby amended to add the following to the end thereof:

On May 10, 2010, GP Holdings, ETE and ETE GP Buyer entered into an agreement for the sale by GP Holdings of (i) 100% of the membership interests in RGPLLC; and (ii) 99.999% of the limited partner interest in RGPLP, to ETE GP Buyer, in exchange for approximately \$300 million in convertible preferred units of ETE, with the rights and preferences set forth in Amendment No. 3 to Third Amended and Restated Agreement of Limited Partnership of ETE.

Upon the closing of the GP Transfer, LP Holdings and Issuer expect to enter into a registration rights agreement pursuant to which LP Holdings will have the right to cause Issuer to prepare and file a registration statement to permit the resale of any Common Units held by LP Holdings from time to time as permitted by Rule 415 promulgated under the Securities Act. If the closing of the GP Transfer occurs and LP Holdings either exercises its right to have a registration statement filed or does so pursuant to an exemption from registration, it may, from time to time, dispose of a portion or all of its Common Units based on market and general economic conditions, the business affairs and financial condition of Issuer, the market price of the Common Units and other factors deemed relevant by LP Holdings.

On April 30, 2010, EFS Partner, sold 76,989 units representing general partner interests in the Haynesville General Partnership, to Issuer in exchange for \$92,086,857 in cash and the entering into of an agreement pursuant to which EFS Partner has agreed to vote the units representing general partner interests as directed by RHIG.

Item 7. Material to be filed as Exhibits.

Exhibit A Joint Filing Agreement*

Exhibit B General Partner Purchase Agreement, dated as of May 10, 2010, by and among Regency GP Acquirer, L.P., Energy Transfer Equity, L.P. and ETE GP Acquirer LLC*

* Filed Herewith

7

SIGNATURES

After reasonable inquiry and to the best of my knowledge and belief, I certify that the information set forth in this statement is true, complete and correct.

Dated: May 13, 2010

Regency LP Acquirer, L.P.

By: EFS Regency GP Holdco II, LLC

Its: General Partner

By: Aircraft Services Corporation

its Managing Member

By: /s/ Tyson R. Yates

Name: Tyson R. Yates

Title: Vice President

General Electric Capital Corporation

By: /s/ J. Alex Urquhart

Name: J. Alex Urquhart

Title: Authorized Signatory

General Electric Company

By: /s/ J. Alex Urquhart

Name: J. Alex Urquhart

Title: Vice President, General Electric Company

Table of Contents

Execution Version

EXHIBIT A

AGREEMENT REGARDING THE JOINT FILING OF SCHEDULE 13D

The undersigned hereby agree as follows:

- (i) Each of them is individually eligible to use the Schedule 13D to which this Exhibit is attached, and such Schedule 13D is filed on behalf of each of them; and
- (ii) Each of them is responsible for the timely filing of such Schedule 13D and any amendments thereto, and for the completeness and accuracy of the information concerning such person contained therein; but none of them is responsible for the completeness or accuracy of the information concerning the other persons making the filing, unless such person knows or has reason to believe that such information is inaccurate.

Date: May 13, 2010

[Signature Pages Follow]

Regency LP Acquirer, L.P.

By: EFS Regency GP Holdco II, LLC

Its: General Partner

By: Aircraft Services Corporation

its Managing Member

By: /s/ Tyson R. Yates

Name: Tyson R. Yates Title: Vice President

General Electric Capital Corporation

By: /s/ J. Alex Urquhart

Name: J. Alex Urquhart Title: Authorized Signatory

General Electric Company

By: /s/ J. Alex Urquhart

Name: J. Alex Urquhart Title: Vice President, General

Electric Company

EXHIBIT B

GENERAL PARTNER PURCHASE AGREEMENT
BY AND AMONG
REGENCY GP ACQUIRER, L.P.
AND
ENERGY TRANSFER EQUITY, L.P.
AND
ETE GP ACQUIRER LLC
May 10, 2010

TABLE OF CONTENTS

	Page
ARTICLE I	
DEFINITIONS AND INTERPRETATIONS	
1.1 Definitions	2
1.2 Interpretations	2
ARTICLE II	
ACQUISITION OF THE ACQUIRED GP INTERESTS; CLOSING	
2.1 Acquisition of the Acquired GP Interests	2
2.2 Time and Place of Closing	2
2.3 Deliveries and Actions at Closing	3
2.4 Pro Ration of Closing Quarterly Distribution	4
ARTICLE III	
REPRESENTATIONS AND WARRANTIES OF SELLER	
REFRESERVITATIONS THAT WINDOWS SEEDER	
3.1 Organization; Qualification	4
3.2 Authority; Enforceability	5
3.3 Non-Contravention	5
3.4 Governmental Approvals	5
3.5 Capitalization	6
3.6 Ownership of Acquired GP Interests	7
3.7 Compliance with Law	8
3.8 Title to Properties and Assets 3.9 Rights-of-Way	8 8
3.10 Regency SEC Reports; Financial Statements; Operating Surplus	8
3.11 Absence of Certain Changes	9
3.12 Environmental Matters	9
3.13 Material Contracts	10
3.14 Legal Proceedings	11
3.15 Permits	12
3.16 Taxes	12
3.17 Employee Benefits; Employment and Labor Matters	13
3.18 Brokers Fee	15
3.19 Regulatory Matters; Compliance	15
3.20 Intellectual Property	15
3.21 Matters Relating to Acquisition of the Convertible Preferred Units	15
ARTICLE IV	
REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES	
4.1 Organization: Qualification	1.6
4.1 Organization; Qualification4.2 Authority; Enforceability; Valid Issuance	16 16
4.3 Non-Contravention	17
4.4 Governmental Approvals	17
r rr	-,

4.5 Capitalization 18

Table of Contents

Table of Contents

4.6.C. 11 14.1	1 age	
4.6 Compliance with Law	19	
4.7 Title to Properties and Assets	20	
4.8 Rights-of-Way	20	
4.9 ETE SEC Reports; Financial Statements; Operating Surplus	20	
4.10 Absence of Certain Changes	21	
4.11 Environmental Matters	21	
4.12 Legal Proceedings	22	
4.13 Permits	22	
4.14 Taxes	22	
4.15 Employee Benefits; Employment and Labor Matters	23	
4.16 Brokers Fee	24	
4.17 Regulatory Matters; Compliance	25	
4.18 Intellectual Property	25	
4.19 Matters Relating to Acquisition of the Acquired GP Interests	25	
ARTICLE V		
COVENANTS OF THE PARTIES		
5.1 Conduct of Business	26	
5.2 Notice of Certain Events	29	
5.3 Access to Information	29	
5.4 Governmental Approvals	30	
5.5 Expenses	31	
5.6 Further Assurances	31	
5.7 Public Statements	31	
5.8 Convertible Preferred Units	31	
5.9 No Solicitation	32	
5.10 Confidential Information	33	
5.11 No Hire	33	
5.12 Tax Matters	33	
5.13 Books and Records; Financial Statements; Litigation Support	35	
5.14 Commitment Regarding Indemnification Provisions; D&O Insurance Continuation	36	
5.15 GECC Names and Marks	37	
ARTICLE VI		
CONDITIONS TO CLOSING		
6.1 Conditions to Obligations of Each Party	37	
6.2 Conditions to Obligations of the Buyer Parties	38	
6.3 Conditions to Obligations of Seller	38	
ARTICLE VII		
TERMINATION RIGHTS		
7.1 Termination Rights	39	
7.2 Effect of Termination	40	

ii

Table of Contents

Table of Contents

	Page
ARTICLE VIII	
INDEMNIFICATION	
8.1 Indemnification by Seller	40
8.2 Indemnification by the Buyer Parties	41
8.3 Limitations and Other Indemnity Claim Matters	41
8.4 Indemnification Procedures	43
8.5 No Reliance	44
ARTICLE IX	
GOVERNING LAW AND CONSENT TO JURISDICTION	
9.1 Governing Law	45
9.2 Consent to Jurisdiction	45
9.3 Waiver of Jury Trial	46
ARTICLE X	
GENERAL PROVISIONS	
10.1 Amendment and Modification	46
10.2 Waiver of Compliance; Consents	46
10.3 Notices	46
10.4 Assignment	47
10.5 Third Party Beneficiaries	47
10.6 Entire Agreement	48
10.7 Severability	48
10.8 Representation by Counsel	48
10.9 Disclosure Schedules	48
10.10 Facsimiles; Counterparts iii	49
111	

Table of Contents

Exhibits

Exhibit A Definitions

Schedules

Annex B

Annex C

Annex D

Schedule 2.3(a)(iv)	Required Consents (Seller Parties)
Schedule 2.3(a)(v)	Amendments to Regency Credit Agreement
Schedule 2.3(b)(v)	Required Consents (Buyer Parties)
Schedule 3.3	Non-Contravention (Seller Parties)
Schedule 3.4	Governmental Approvals
Schedule 3.5(a)	Capitalization
Schedule 3.11	Absence of Certain Changes
Schedule 3.12	Environmental Matters
Schedule 3.13	Material Contracts
Schedule 3.14	Legal Proceedings
Schedule 3.17	Employee Matters
Schedule 3.19	Regulatory Status
Schedule 4.3	Non-Contravention (Buyer Parties)
Schedule 4.4	Governmental Approvals
Schedule 4.5	Employee Benefit Plan
Schedule 4.10	Absence of Certain Changes
Schedule 4.11	Environmental Matters
Schedule 4.12	Legal Proceedings
Schedule 4.15(b)	Employee Matters
Schedule 5.1	Conduct of Regency Entities Business
Schedule 5.11	No Hire
Schedule 5.14	Continuing D&O Insurance Policies
Schedule 6.1(a)	Approvals
Schedule 6.1(c)	Required Consents
Schedule 6.2(d)	Amendment to Regency Credit Agreement
Schedule 6.2(e)	Amendment to ETE Credit Agreement
<u>Annexes</u>	
Annex A Form of A	Amendment No. 3 to the ETE Partnership Agreement

Annex E Government Policies

Form of Assignment of Interests

Form of Opinion of Vinson & Elkins L.L.P.

Form of Registration Rights Agreement

Table of Contents 22

iv

GENERAL PARTNER PURCHASE AGREEMENT

This **GENERAL PARTNER PURCHASE AGREEMENT** (this *Agreement*), dated as of May 10, 2010 (the *Execution Date*), is made and entered into by and among Regency GP Acquirer LP, a Delaware limited partnership (*Seller*), Energy Transfer Equity, L.P. a Delaware limited partnership (*ETE*) and ETE GP Acquirer LLC, a Delaware limited liability company (*Buyer*).

ETE and Buyer are sometimes referred to individually in this Agreement as a *Buyer Party* and are sometimes collectively referred to in this Agreement as the *Buyer Parties*.

Each of the parties to this Agreement is sometimes referred to individually in this Agreement as a *Party* and all of the parties to this Agreement are sometimes collectively referred to in this Agreement as the *Parties*.

RECITALS

WHEREAS, Seller owns the following interests (collectively, the *Acquired GP Interests*): (i) 100% of the membership interests in Regency GP LLC, a Delaware limited liability company (*RGPLLC*); and (ii) the 99.999% limited partner interest in Regency GP LP, a Delaware limited partnership (*RGPLP*);

WHEREAS, RGPLLC owns a 0.001% general partner interest in RGPLP;

WHEREAS, RGPLP owns a 2.0% general partner interest in Regency Energy Partners, L.P., a Delaware limited partnership (*Regency*), and is the sole general partner of Regency;

WHEREAS, pursuant to that certain Redemption and Exchange Agreement dated as of the date hereof (the *ETP Redemption Agreement*), by and between ETE and Energy Transfer Partners, L.P., a Delaware limited partnership (*ETP*), subject to the terms and conditions contained therein, ETP has agreed to redeem (the *Redemption and Exchange*) certain limited partner interests of ETP held by ETE in exchange for (a) all of the outstanding membership interests in ETC Midcontinent Express Pipeline III, L.L.C., a Delaware limited liability company (*ETC III*) and (b) an option to acquire all of the outstanding membership interests in ETC Midcontinent Express Pipeline II, L.L.C., a Delaware limited liability company (*ETC III*);

WHEREAS, at the Closing, ETE, ETP and Regency intend to enter into a Master Services Agreement in substantially the form attached hereto as Exhibit H to the Contribution Agreement (as defined below) (the **Master Services Agreement**);

WHEREAS, pursuant to that certain Contribution Agreement dated as of the date hereof (the *MEP Contribution Agreement*), by and among ETE, Regency and Regency Midcontinent Express LLC, subject to the terms and conditions contained therein, ETE has agreed, in exchange for limited partner interests in Regency, to (a) contribute ETC III to Regency and (b) assign Regency an option to acquire ETC II (the *MEP Contribution*); and

WHEREAS, Seller desires to sell to the Buyer Parties, and the Buyer Parties desire to purchase from Seller, the Acquired GP Interests, and in exchange ETE desires to issue and sell to Seller 3,000,000 convertible preferred units of ETE (the *Convertible Preferred Units*), having the terms set forth in Amendment No. 3 to the Third Amended and Restated Agreement of Limited Partnership of Energy Transfer Equity, L.P. (the *Third Amendment*) a form of which is attached hereto as <u>Annex A</u>, all on the terms and subject to the conditions set forth herein.

1

ARTICLE I DEFINITIONS AND INTERPRETATIONS

- 1.1 *Definitions*. Capitalized terms used in this Agreement but not defined in the body of this Agreement shall have the meanings ascribed to them in <u>Exhibit A</u>. Capitalized terms defined in the body of this Agreement are listed in <u>Exhibit A</u> with reference to the location of the definitions of such terms in the body of this Agreement.
- 1.2 Interpretations. In this Agreement, unless a clear contrary intention appears: (a) the singular includes the plural and vice versa; (b) reference to a Person includes such Person s successors and assigns but, in the case of a Party, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity; (c) reference to any gender includes each other gender; (d) references to any Exhibit, Schedule, Section, Article, Annex, subsection and other subdivision refer to the corresponding Exhibits, Schedules, Sections, Articles, Annexes, subsections and other subdivisions of this Agreement unless expressly provided otherwise; (e) references in any Section or Article or definition to any clause means such clause of such Section, Article or definition; (f) hereunder, hereof, hereto and words of similar import are references to this Agreement as a whole and not to any particular provision of this Agreement; (g) the word or is not exclusive, and the word including (in its various forms) means including without limitation; (h) each accounting term not otherwise defined in this Agreement has the meaning commonly applied to it in accordance with GAAP; (i) references to days are to calendar days; and (j) all references to money refer to the lawful currency of the United States. The Table of Contents and the Article and Section titles and headings in this Agreement are inserted for convenience of reference only and are not intended to be a part of, or to affect the meaning or interpretation of, this Agreement.

ARTICLE II

ACQUISITION OF THE ACQUIRED GP INTERESTS; CLOSING

- 2.1 *Acquisition of the Acquired GP Interests*. Upon the terms and subject to the satisfaction or due waiver of the conditions contained in this Agreement, Seller shall sell, transfer and deliver to Buyer, and Buyer shall purchase from Seller the Acquired GP Interests. In exchange for the Acquired GP Interests, at the Closing, ETE shall issue and deliver to Seller the Convertible Preferred Units.
- 2.2 *Time and Place of Closing*. The closing of the transactions contemplated by this Agreement (the *Closing*) will take place at the offices of Vinson & Elkins L.L.P., 1001 Fannin Street, Suite 2500, Houston, Texas 77002 on the second Business Day after all of the conditions set forth in <u>Article VI</u> (other than those conditions which by their terms are only capable of being satisfied at the Closing, but subject to the satisfaction or due waiver of those conditions) have been satisfied or waived by the Party or Parties entitled to waive such conditions, unless another time, date and place are agreed to in writing by the Parties. The date of the Closing is referred to in this Agreement as the *Closing Date*. The Closing will be deemed effective as of 12:01 a.m., Houston, Texas time, on the Closing Date.

2

- 2.3 Deliveries and Actions at Closing.
- (a) At the Closing, Seller shall deliver, or shall cause to be delivered, the following to the Buyer Parties:
- (i) <u>Assignment of Interests</u>. A counterpart of an assignment (the *Assignment of Interests*), a form of which is attached hereto as <u>Annex B</u>, evidencing the contribution, assignment, transfer and delivery to Buyer of the Acquired GP Interests, duly executed by Seller;
- (ii) <u>FIRPTA Certificate</u>. A certificate of Seller in the form specified in Treasury Regulation Section 1.1445-2(b)(2)(iv) that Seller is not a foreign person within the meaning of Section 1445 of the Code, duly executed by Seller;
 - (iii) Closing Certificate. The certificate contemplated by Section 6.2(c);
 - (iv) Required Consents. The consents, approvals and waivers set forth on Schedule 2.3(a)(iv);
- (v) <u>Amendment to Regency Credit Agreement</u>. A duly executed amended Regency Credit Agreement, which address the matters set forth on Schedule 2.3(a)(v); and
- (vi) <u>Registration Rights Agreement</u>. A counterpart of a registration rights agreement, a form of which attached hereto as <u>Annex D</u> (the *Registration Rights Agreement*) duly executed by Seller.
 - (b) At the Closing, the Buyer Parties shall deliver, or shall cause to be delivered, the following to Seller:
 - (i) <u>Preferred Unit Certificate</u>. Original unit certificates representing the Convertible Preferred Units;
 - (ii) <u>Assignment of Interests</u>. A counterpart of the Assignment of Interests duly executed by Buyer;
- (iii) <u>Third Amendment</u>. A copy of the Third Amendment duly executed by the general partner of ETE, as general partner of ETE and as attorney-in-fact for all limited partners pursuant to the powers of attorney granted pursuant to Section 2.6 of the ETE Partnership Agreement, a form of which is attached hereto as <u>Annex A</u>;
 - (iv) Closing Certificate. The certificate contemplated by Section 6.3(c);
 - (v) Required Consents. The consents, approvals and waivers set forth on Schedule 2.3(b)(v);
 - (vi) MEP Contribution. A duly executed copy of the MEP Contribution Agreement;

2

Table of Contents

- (vii) ETP Redemption Agreement. A duly executed copy of the ETP Redemption Agreement;
- (viii) <u>Legal Opinion</u>. An opinion from Vinson & Elkins L.L.P., counsel to Buyer Parties, dated as of the Closing Date and reasonably satisfactory to Seller, a form of which is attached hereto as <u>Annex C</u>;
- (ix) <u>Registration Rights Agreement</u>. A counterpart of the Registration Rights Agreement, duly executed by ETE; and
- (x) <u>Governance Policies</u>. A copy of the Governance Policies adopted by the general partner of ETE, a form of which is attached hereto as Annex E.
 - 2.4 Pro Ration of Closing Quarterly Distribution.
- (a) Within three (3) Business Days after Buyer receives its regular quarterly cash distribution in respect of the quarter in which the Closing occurs, Buyer shall pay to Seller an amount in cash equal to the Pro Rata Distribution Amount. The *Pro Rata Distribution Amount* shall equal the product of (a) \$1,574,073.87 multiplied by (b) a fraction, the numerator of which is the number of days in the quarter preceding the date of the Closing Date, plus the Closing Date, and the denominator of which is the number of days in such quarter.
- (b) In the event that the Closing Date occurs on or before the record date relating to the distribution for the quarter immediately prior to the quarter in which the Closing Date occurs (the *Preceding Quarter*), in addition to any amounts to be paid by Buyer to Seller pursuant to Section 2.4(a). Buyer shall pay to Seller 100% of the applicable cash distribution for the Regency GP LP Interest for the Preceding Quarter. The payment shall be made no later than three (3) Business Days following the receipt by Buyer of such distribution from Regency.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF SELLER

Seller hereby represents and warrants to the Buyer Parties as follows:

3.1 *Organization; Qualification*. Seller and each Regency Entity are entities duly formed, validly existing and in good standing under the laws of the State of Delaware and have all requisite corporate, limited partnership or limited liability company power and authority to own, lease and operate their properties and to carry on their business as it is now being conducted, and are duly qualified, registered or licensed to do business as a foreign entity and are in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification necessary, except where the failure to be so duly qualified, registered or licensed and in good standing would not reasonably be expected to have a Regency Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair Seller s ability to perform its obligations under this Agreement. Seller has made available to the Buyer Parties true and complete copies of the Organizational Documents of each Regency Entity, as in effect on the Execution Date.

4

Table of Contents

3.2 Authority; Enforceability.

- (a) Seller has the requisite corporate, partnership or limited liability company power and authority to execute and deliver this Agreement and to consummate the transactions contemplated by this Agreement. The execution and delivery by Seller of this Agreement and the consummation by Seller of the transactions contemplated by this Agreement have been duly and validly authorized by Seller, and no other corporate, partnership or limited liability company proceedings on the part of Seller is necessary to authorize this Agreement or to consummate the transactions contemplated by this Agreement.
- (b) This Agreement has been duly executed and delivered by Seller, and, assuming the due authorization, execution and delivery by the Buyer Parties, this Agreement constitutes the valid and binding agreement of Seller, enforceable against Seller in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws relating to or affecting creditors—rights generally and subject, as to enforceability, to legal principles of general applicability governing the availability of equitable remedies, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether such enforceability is considered in a proceeding in equity or at law) (collectively, *Creditors Rights*).
- 3.3 Non-Contravention. Except as set forth on Schedule 3.3 of the Seller Disclosure Schedule, the execution, delivery and performance of this Agreement and the consummation by Seller of the transactions contemplated by this Agreement does not and will not: (a) result in any breach of any provision of the Organizational Documents of Seller or any Regency Entity; (b) constitute a default (or an event that with notice or passage of time or both would give rise to a default) under, or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both) under any of the terms, conditions or provisions of any Contract to which Seller or any Regency Entity is a party or by which any property or asset of Seller or any Regency Entity is bound or affected; or (c) assuming compliance with the matters referred to in Section 3.4, violate any Law to which Seller or any Regency Entity is subject or by which any of Seller s or any Regency Entity s properties or assets is bound; or (d) constitute (with or without the giving of notice or the passage of time or both) an event which would result in the creation of any Lien (other than Permitted Liens) on any asset of any Regency Entity, except, in the cases of clauses (b), (c) and (d) for such defaults or rights of termination, cancellation, amendment, acceleration, violations or Liens, as would not reasonably be expected to have a Regency Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair Seller s ability to perform their obligations under this Agreement.
- 3.4 *Governmental Approvals*. Except as set forth on <u>Schedule 3.4</u> of the Seller Disclosure Schedule, no declaration, filing or registration with, or notice to, or authorization, consent or approval of, any Governmental Authority is necessary for the consummation by Seller of the transactions contemplated by this Agreement, other than such declarations, filings, registrations, notices, authorizations, consents or approvals which, if not obtained or made, would not reasonably be expected to have a Regency Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair Seller s ability to perform their obligations under this Agreement.

5

Table of Contents

3.5 Capitalization.

- (a) <u>Schedule 3.5(a)</u> of the Seller Disclosure Schedule sets forth a correct and complete description of the following: (i) all of the issued and outstanding equity interests in each of the Regency GP Entities; and (ii) the record owners of each of the outstanding equity interests in each of the Regency GP Entities. Except as set forth on <u>Schedule 3.5(a)</u> of the Seller Disclosure Schedule, there are no other outstanding equity interests of any Regency GP Entity. All of the issued and outstanding equity interests in each of the Regency GP Entities have been duly authorized, validly issued and fully paid and are nonassessable (except as such nonassessability may be affected by Sections 17-303 and 17-607 of the Delaware LP Act or Section 18-607 of the Delaware LLC Act) and have not been issued in violation of any preemptive rights, rights of first refusal or other similar rights of any Person. All of the issued and outstanding equity interests in each of the Regency GP Entities are owned by the Persons set forth on <u>Schedule 3.5(a)</u> of the Seller Disclosure Schedule named as owning such interests free and clear of all Liens other than (A) transfer restrictions imposed by federal and state securities laws and (B) any transfer restrictions contained in the Organizational Documents of the Regency GP Entities.
- (b) At the Closing, (i) the Acquired GP Interests will constitute 100% of the issued and outstanding membership interests in RGPLLC and a 99.999% limited partner interest in RGPLP and (ii) RGPLLC will own a 0.001% general partner interest in, and serve as the sole general partner of, RGPLP.
- (c) There are no preemptive rights or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, subscription agreements, commitments or rights of any kind that obligate any of the Regency GP Entities to issue or sell any equity interests or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any equity interests in any of the Regency GP Entities, and no securities or obligations evidencing such rights are authorized, issued or outstanding.
- (d) No Regency GP Entity has outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the holders of equity interests in any Regency GP Entity on any matter.
- (e) RGPLP is the sole general partner of Regency with a 2.0% general partner interest in Regency (the *Regency GP Interest*) and owns 100% of the Regency Incentive Distribution Rights (collectively with the Regency GP Interest, the *Regency GP LP Interests*). The Regency GP LP Interests are owned by RGPLP free and clear of all Liens, other than (i) transfer restrictions imposed by federal and state securities laws and (ii) any transfer restrictions contained in the Regency Partnership Agreement.
- (f) Except as set forth in <u>Schedule 3.5(a)</u> of the Seller Disclosure Schedule or <u>Section 3.5(e)</u>, no Regency GP Entity owns any equity interest in any other Person.

6

Table of Contents

- (g) As of the Execution Date: (i) 93,191,602 Regency Common Units were issued and outstanding, (ii) 4,371,586 Series A Cumulative Convertible Preferred Units of Regency (*Regency Series A Units*), which Regency Series A Units are convertible into Regency Common Units at an initial conversion price of \$18.30 per unit, subject to adjustment, were issued and outstanding and (iii) 1,155,129 Regency Common Units were available for issuance under Regency s employee benefit plans, of which 297,651 Regency Common Units were subject to issuance upon exercise of outstanding Regency options, 267,135 Regency Common Units were subject to issuance upon the vesting of outstanding phantom units and 355,609 Regency Common Units were subject to issuance upon the vesting of outstanding un-vested restricted units.
- (h) All of the limited partner interests in Regency are duly authorized and validly issued in accordance with the Organizational Documents of Regency, and are fully paid (to the extent required under the Organizational Documents of Regency) and nonassessable (except as nonassessability may be affected by Sections 17-303 and 17-607 of the Delaware LP Act) and were not issued in violation of any preemptive rights, rights of first refusal or other similar rights of any Person.
- (i) Except as set forth in the Organizational Documents of Regency and except as otherwise provided in Section 3.5(g), there are no preemptive rights or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, subscription agreements, commitments or rights of any kind that obligate any of the Regency Entities to issue or sell any equity interests of Regency or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any equity interests in Regency, and no securities or obligations evidencing such rights are authorized, issued or outstanding.
- (j) Except for the Regency Series A Units, none of the Regency Entities has outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the holders of equity interests in Regency on any matter.

3.6 Ownership of Acquired GP Interests.

- (a) Upon the consummation of the transactions contemplated by this Agreement, Seller will assign, convey, transfer and deliver to Buyer good and valid title to the Acquired GP Interests free and clear of all Liens other than (i) any transfer restrictions imposed by federal and state securities laws, (ii) any transfer restrictions contained in the Organizational Documents of Regency GP Acquirer and (iii) any Liens on the Acquired GP Interests as a result of actions by the Buyer Parties.
- (b) Seller is not a party to any agreements, arrangements or commitments obligating Seller to grant, deliver or sell, or cause to be granted, delivered or sold, the Acquired GP Interests, by sale, lease, license or otherwise, other than this Agreement.
- (c) There are no voting trusts, proxies or other agreements or understandings to which Seller is bound with respect to the voting of the Acquired GP Interests.

7

Table of Contents

- 3.7 Compliance with Law. Except as to specific matters disclosed in the Regency SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such Regency SEC Documents or any other disclosures in such Regency SEC Documents to the extent they are predictive or forward looking and general in nature), and except for Environmental Laws, Laws requiring the obtaining or maintenance of a Permit, Tax matters, Laws relating to employee benefits, employment and labor matters, and Laws relating to regulatory and compliance matters, which are the subject of Sections 3.12, 3.15, 3.16, 3.17, and 3.18, respectively, and except as to matters that would not reasonably be expected to have a Regency Material Adverse Effect, (a) each Regency Entity is in compliance with all applicable Laws, (b) none of Seller, or, to the Knowledge of Seller any Regency Entity, has received written notice of any violation of any applicable Law, and (c) to the Knowledge of Seller, none of the Regency Entities is under investigation by any Governmental Authority for potential non-compliance with any Law.
- 3.8 *Title to Properties and Assets*. To the Knowledge of Seller, except as to matters that would not reasonably be expected to have a Regency Material Adverse Effect, each Regency Entity has title to or rights or interests in its real property and personal property, free and clear of all Liens (subject to Permitted Liens), sufficient to allow it to conduct its business as currently being conducted.
- 3.9 *Rights-of-Way*. To the Knowledge of Seller, except as to matters that would not reasonably be expected to have a Regency Material Adverse Effect, (a) each Regency Entity has such Rights-of-Way from each Person as are necessary to use, own and operate each Regency Entity s assets in the manner such assets are currently used, owned and operated by each Regency Entity, (b) each Regency Entity has fulfilled and performed all of its obligations with respect to such Rights-of-Way and (c) no event has occurred that allows, or after the giving of notice or the passage of time, or both, would allow, revocation or termination thereof or would result in any impairment of the rights of the holder of any such Rights-of-Way.
 - 3.10 Regency SEC Reports; Financial Statements; Operating Surplus.
- (a) Regency has furnished or filed all reports, schedules, forms, statements and other documents (including exhibits and other information incorporated therein) required to be furnished or filed by Regency with the Securities and Exchange Commission (*SEC*) since January 1, 2009 (such documents being collectively referred to as the *Regency SEC Documents*).
- (b) Each Regency SEC Document (i) at the time filed, complied in all material respects with the requirements of the Exchange Act and the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Regency SEC Document and (ii) did not at the time it was filed (or if amended or superseded by a filing or amendment prior to the Execution Date, then at the time of such filing or amendment) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

8

Table of Contents

- (c) Each of the financial statements of Regency or RGPLP included in the Regency SEC Documents (*Regency Financial Statements*) complied at the time it was filed as to form in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, have been prepared in accordance with GAAP, applied on a consistent basis throughout the periods presented thereby and fairly present in all material respects the consolidated financial position and operating results, equity and cash flows of Regency as of, and for the periods ended on, the respective dates thereof, subject, however, in the case of unaudited financial statements, to normal year-end audit adjustments and accruals and the absence of notes and other textual disclosures as permitted by Form 10-Q of the SEC.
- (d) To the Knowledge of Seller, none of the Regency Entities has any liability, whether accrued, contingent, absolute or otherwise, except for (i) liabilities set forth on the consolidated balance sheet of Regency dated as of March 31, 2010 or the notes thereto, (ii) liabilities that have arisen since March 31, 2010 in the ordinary course of business, and (iii) liabilities which would not reasonably be expected to have a Regency Material Adverse Effect.
- (e) The Regency GP Entities do not have assets or liabilities other than in connection with their direct or indirect ownership of the 2.0% general partner interest in Regency and the Regency Incentive Distribution Rights.
- (f) Seller does not have assets or liabilities other than in connection with its ownership of the Acquired GP Interests and the relationships of the parties under its partnership agreement with respect thereto.
- (g) All distributions made by Regency since its initial public offering have been made from Operating Surplus (as defined in the Regency Partnership Agreement) and in accordance with the terms of the Regency Partnership Agreement.
- 3.11 Absence of Certain Changes. Except as to specific matters disclosed in the Regency SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such Regency SEC Documents or any other disclosures in such Regency SEC Documents to the extent they are predictive or forward looking and general in nature), and except as set forth on Schedule 3.11 of the Seller Disclosure Schedule or as expressly contemplated by this Agreement, since December 31, 2009 (a) to the Knowledge of Seller, the business of the Regency Entities has been conducted in the ordinary course and in a manner consistent with past practice; and (b) there has not been (i) any event, occurrence or development which has had, or would be reasonably expected to have, a Regency Material Adverse Effect or (ii) to the Knowledge of Seller, the occurrence of any of the transactions or matters described in Section 5.1(b).
- 3.12 *Environmental Matters*. Except as to specific matters disclosed in the Regency SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such Regency SEC Documents or any other disclosures in such Regency SEC Documents to the extent they are predictive or forward looking and general in nature), and except for matters set forth on

9

Table of Contents

<u>Schedule 3.12</u> of the Seller Disclosure Schedule and except for matters that would not reasonably be expected to have a Regency Material Adverse Effect:

- (a) each of the Regency Entities is in compliance with all applicable Environmental Laws;
- (b) each of the Regency Entities possesses all Permits required under Environmental Laws for its operations as currently conducted and is in compliance with the terms of such Permits, and such Permits are in full force and effect;
- (c) none of the Regency Entities nor any of their properties or operations are subject to any pending or, to the Knowledge of Seller, threatened Proceeding arising under any Environmental Law, nor has any of the Regency Entities received any written and pending notice, order or complaint from any Governmental Authority alleging a violation of or liability arising under any Environmental Law; and
- (d) to the Knowledge of Seller, there has been no Release of Hazardous Substances on, at, under, to, or from any of the properties of the Regency Entities, or from or in connection with the Regency Entities operations in a manner that would reasonably be expected to give rise to any liability pursuant to any Environmental Law.

3.13 Material Contracts.

- (a) Except as set forth on <u>Schedule 3.13</u> of the Seller Disclosure Schedule or filed with any Regency SEC Document (including incorporation by reference) filed with the SEC on or after January 1, 2010 and prior to the date hereof, as of the Execution Date, to the Knowledge of Seller, none of the Regency Entities is a party to or bound by any Contract that:
- (i) is of a type that would be required to be included as an exhibit to a Registration Statement on Form S-1 pursuant to Items 601(b)(2), (4), (9) or (10) of Regulation S-K of the SEC if such a registration statement was filed by Regency on the Execution Date;
- (ii) includes Seller or any Affiliate of Seller (other than Regency or its Subsidiaries) as a counter party or third party beneficiary;
- (iii) contains any provision or covenant which materially restricts any Regency Entity or any Affiliate thereof from engaging in any lawful business activity or competing with any Person;
- (iv) (A) relates to the creation, incurrence, assumption, or guarantee of any indebtedness for borrowed money by any Regency Entity or (B) creates a capitalized lease obligation (except, in the cases of clauses (A) and (B), any such Contract with an aggregate principal amount not exceeding \$10,000,000);
- (v) is in respect of the formation of any partnership or joint venture or otherwise relates to the joint ownership or operation of the assets owned by any of the Regency Entities involving assets or obligations in excess of \$75,000,000;

10

Table of Contents

- (vi) includes the acquisition or sale of assets with a book value in excess of \$50,000,000 (whether by merger, sale of stock, sale of assets or otherwise);
- (vii) any Contract or commitment that involves a sharing of profits, losses, costs or liabilities by any Regency Entity with any other Person other than gas processing contracts; or
- (viii) otherwise involves the annual payment by or to any of the Regency Entities of more than \$10,000,000 and cannot be terminated by the Regency Entities on 90 days or less notice without payment by the Regency Entities of any material penalty.
- (b) Except as set forth on Schedule 3.13 of the Seller Disclosure Schedule, each Contract required to be disclosed pursuant to Section 3.13(a) and each Contract to which any of the Regency Entities is bound as of the Execution Date that relates to (A) the purchase of materials, supplies, goods, services or other assets, (B) the purchase, sale, transporting, treatment, gathering, processing or storing of, or gas compression services rendered in connection with, natural gas, condensate or other liquid or gaseous hydrocarbons or the products therefrom, or the provision of services related thereto or (C) the construction of capital assets, in the cases of clauses (A), (B) and (C) that (i) provides for either (1) annual payments by or to any of the Regency Entities in excess of \$10,000,000 (collectively, the Regency Material Contracts) has been made available to the Buyer Parties and to the Knowledge of Seller is a valid and binding obligation of the applicable Regency Entity, and is in full force and effect and enforceable in accordance with its terms against such Regency Entity and, to the Knowledge of Seller, the other parties thereto, except, in each case, as enforcement may be limited by Creditors Rights.
- (c) To the Knowledge of Seller, none of the Regency Entities nor any other party to any Regency Material Contract is in default or breach in any material respect under the terms of any Regency Material Contract and no event has occurred that with the giving of notice or the passage of time or both would constitute a breach or default in any material respect by such Regency Entity or, to the Knowledge of Seller, any other party to any Regency Material Contract, or would permit termination, modification or acceleration under any Regency Material Contract.
- 3.14 *Legal Proceedings*. Except as to specific matters disclosed in the Regency SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such Regency SEC Documents or any other disclosures in such Regency SEC Documents to the extent they are predictive or forward looking and general in nature), and other than with respect to Proceedings arising under Environmental Laws which are the subject of Section 3.12 or as is set forth on Schedule 3.14 of Seller Disclosure Schedule, there are no Proceedings pending or, to the Knowledge of Seller, threatened against the Regency Entities, except such Proceedings as would not reasonably be expected to have a Regency Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair Seller s ability to perform their obligations under this Agreement.

11

Table of Contents

3.15 *Permits*. To the Knowledge of Seller, other than with respect to Permits issued pursuant to or required under Environmental Laws which are the subject of <u>Section 3.12</u>, the Regency Entities have all Permits as are necessary to use, own and operate their assets in the manner such assets are currently used, owned and operated by the Regency Entities, except where the failure to have such Permits would not reasonably be expected to have a Regency Material Adverse Effect.

3.16 *Taxes*.

- (a) All material Tax Returns required to be filed with respect to the Regency GP Entities or, to the Knowledge of Seller, the other Regency Entities have been filed and all Tax Returns with respect to the Regency GP Entities and, to the Knowledge of Seller, the other Regency Entities are complete and correct in all material respects and all material Taxes due relating to the Regency GP Entities and, to the Knowledge of Seller, the other Regency Entities have been paid in full. Except as disclosed on Schedule 3.16 of Seller Disclosure Schedule, there is no claim (other than claims being contested in good faith through appropriate proceedings and for which adequate reserves have been made in accordance with GAAP) against any Regency GP Entity or, to the Knowledge of Seller, any other Regency Entity for any material Taxes, and no material assessment, deficiency, or adjustment has been asserted or proposed in writing with respect to any material Taxes or material Tax Returns of or with respect to any Regency GP Entity or, to the Knowledge of Seller, any other Regency Entity.
- (b) Except as set forth on <u>Schedule 3.16</u> of the Seller Disclosure Schedule, no material Tax audits or administrative or judicial proceedings are being conducted or are pending with respect to any Regency GP Entity or, to the Knowledge of Seller, any other Regency Entity.
- (c) All material Taxes required to be withheld, collected or deposited by or with respect to any Regency GP Entity or, to the Knowledge of Seller, any other Regency Entity have been timely withheld, collected or deposited as the case may be, and to the extent required, have been paid to the relevant taxing authority.
- (d) There are no outstanding agreements or waivers extending the applicable statutory periods of limitation for any material Tax of, or any material Taxes associated with the ownership or operation of the assets of, any Regency GP Entity or, to the Knowledge of Seller, any other Regency Entity.
- (e) No Regency GP Entity nor, to the Knowledge of Seller, any other Regency Entity is a party to any Tax sharing agreement.
- (f) No Regency GP Entity nor, to the Knowledge of Seller, any other Regency Entity, has engaged in a transaction that would be reportable by or with respect to any Regency Entity pursuant to Treasury Regulation § 1.6011-4 or any predecessor thereto.
- (g) Immediately prior to the consummation of the transactions contemplated by this Agreement, each Regency GP Entity is an entity disregarded as separate from its owner for United States federal income tax purposes and neither of the Regency GP Entities has elected to be treated as a corporation for federal Tax purposes.

12

Table of Contents

- (h) Regency has not elected to be treated as a corporation for federal Tax purposes, and Regency qualifies as a publicly traded partnership within the meaning of Section 7704(b) of the Code and has met, and continues to meet, the gross income requirements (within the meaning of Section 7704(c) of the Code) in each Tax year since its formation. Regency has filed a federal income tax return that has in effect an election pursuant to Section 754 of the Code.
 - 3.17 Employee Benefits; Employment and Labor Matters. To the Knowledge of Seller:
- (a) Except as set forth on <u>Schedule 3.17(a)</u> of the Seller Disclosure Schedule or filed with any Regency SEC Documents (including by incorporation by reference) filed with the SEC on or after January 1, 2010 and prior to the date hereof, no Regency Entity, nor any ERISA Affiliate of any Regency Entity, sponsors, maintains or contributes to, or has sponsored, maintained or contributed to within six years prior to the Closing Date any of the following:
- (i) any employee benefit plan, as such term is defined in Section 3(3) of ERISA (including, but not limited to, employee benefit plans, such as foreign plans, which are not subject to the provisions of ERISA, but excluding any multiemployer plan within the meaning of Section 3(37) of ERISA or multiple employer plan within the meaning of Section 4063(a) of ERISA); or
- (ii) any material personnel policy, equity-based plans (including, but not limited to, stock option plans, stock purchase plans, stock appreciation rights and phantom stock plans), collective bargaining agreement, bonus plan or arrangement, incentive award plan or arrangement, vacation policy, severance pay plan or arrangements, change in control policies or agreements, deferred compensation agreement or arrangement, executive compensation or supplemental income arrangement, consulting agreement, employment agreement and each other employee benefit plan, agreement, arrangement, program, practice or understanding which is not described in <u>Section 3.17(a)(i)</u> (collectively, along with any plan described in <u>Section 3.17(a)(i)</u> above, the *Regency Benefit Plans*).
- (b) True and correct and complete copies of each of the Regency Benefit Plans, related trusts, insurance or group annuity contracts and each other funding or financing arrangement relating to any Plan, including all amendments thereto, have been made available to the Buyer Parties and there has been made available to the Buyer Parties, with respect to each Regency Benefit Plan required to file such report and description, the most recent report on Form 5500 and the summary plan description. Additionally, the most recent determination letter or opinion letter from the Internal Revenue Service for each of the Regency Benefit Plans intended to be qualified under Section 401 of the Code, and any outstanding determination letter application for such plans has been furnished.
- (c) Except as disclosed on <u>Schedule 3.17(c)</u> of the Seller Disclosure Schedule and except for matters that would not reasonably be expected to have a Regency Material Adverse Effect:

13

Table of Contents

- (i) each Regency Benefit Plan has been administered in compliance with its terms, the applicable provisions of ERISA, the Code and all other applicable laws and the terms of all applicable collective bargaining agreements;
- (ii) there are no actions, suits or claims pending (other than routine claims for benefits) or threatened, with respect to any Regency Benefit Plan and no Regency Benefit Plan is under audit or is subject to an investigation by the Internal Revenue Service, the Department of Labor or any other federal or state governmental agency nor is any such audit or investigation pending;
 - (iii) no Regency Benefit Plan is subject to Title IV of ERISA;
- (iv) all contributions and payments required to be made by any Regency Entity or an ERISA Affiliate of any Regency Entity to or under each Regency Benefit Plan have been timely made;
- (v) as to any Regency Benefit Plan intended to be qualified under Section 401 of the Code, there has been no termination or partial termination of such plan within the meaning of Section 411(d)(3) of the Code; and
- (vi) none of the Regency Entities or any of their ERISA Affiliates has any liability with respect to any multiemployer plan within the meaning of Section 3(37) of ERISA or multiple employer plan with the meaning of Section 4063(a) of ERISA.
- (d) In connection with the consummation of the transaction contemplated by this Agreement, no payments have or will be made under the Regency Benefit Plans which, in the aggregate, would result in imposition of the sanctions imposed under Sections 280G and 4999 of the Code.
- (e) No Regency Benefits Plan provides retiree medical or retiree life insurance benefits to any person and neither the Regency Entities nor any ERISA Affiliate of a Regency Entity is contractually or otherwise obligated (whether or not in writing) to provide any person with life insurance or medical benefits upon retirement or termination of employment, other than as required by the provisions of Section 601 through 608 of ERISA and Section 4980B of the Code. Additionally, each Regency Benefits Plan which is an employee welfare benefit plan, as such term is defined in Section 3(1) of ERISA, may be unilaterally amended or terminated in its entirety without liability except as to benefits accrued thereunder prior to such amendment or termination.
- (f) Except as would not reasonably be expected to have a Regency Material Adverse Effect, (a) each of the Regency Entities is in compliance with all applicable labor and employment Laws including, without limitation, all Laws, rules, regulations, orders, rulings, decrees, judgments and awards relating to employment discrimination, payment of wages, overtime compensation, immigration, occupational health and safety, and wrongful discharge; (b) no action, suit, complaint, charge, arbitration, inquiry, proceeding or investigation by or before any Governmental Authority, brought by or on behalf of any employee, prospective or former employee or labor organization or other representative of the employees or of any prospective or former employees of any of the Regency Entities is pending or threatened against

14

Table of Contents

any of the Regency Entities, any present or former director or employee (including with respect to alleged sexual harassment, unfair labor practices or discrimination); (c) no grievance is pending or threatened against any of the Regency Entities; and (d) none of the Regency Entities is subject to or otherwise bound by, any material consent decree, order, or agreement with, any Governmental Authority relating to employees or former employees of any of the Regency Entities. To the Knowledge of Seller, none of the Regency Entities is a signatory party to or otherwise subject to any collective bargaining agreements, and none of the employees of the Regency Entities is represented by a labor union; and there is no labor dispute, strike, work stoppage or other labor trouble (including any organizational drive) against any of the Regency Entities pending or threatened.

3.18 *Brokers Fee.* Except for the fee payable to Morgan Stanley & Co. Incorporated which shall be paid by Seller, no broker, investment banker, financial advisor or other Person is entitled to any broker s, finder s, financial advisor s or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Seller.

3.19 Regulatory Matters; Compliance.

- (a) To the Knowledge of Seller, except as set forth on <u>Schedule 3.19</u> of the Seller Disclosure Schedule, there are no currently effective tariffs authorized and approved by the FERC as of the Execution Date applicable to the Regency Entities, or currently pending material rate filings, certificate applications, or other filings that relate to any of the Regency Entities made with FERC prior to the Execution Date.
- (b) The Regency Entities (i) have all necessary approvals from FERC to provide service to customers pursuant to the Natural Gas Act and the Natural Gas Policy Act of 1978, as amended, and (ii) have made all required FERC filings necessary to offer such service, except where the failure to have any such approval or to have made any such filing would not reasonably be expected to have a Regency Material Adverse Effect.
- (c) Regency is not an investment company or a company controlled by an investment company within the meaning of the Investment Company Act of 1940, as amended.
- 3.20 *Intellectual Property*. To the Knowledge of Seller (a) the Regency Entities own or have the right to use pursuant to license, sublicense, agreement or otherwise all material items of Intellectual Property required in the operation of the business as presently conducted; (b) no third party has asserted in writing delivered to the Regency Entities an unresolved claim that any of the Regency Entities is infringing on the Intellectual Property of such third party and (c) no third party is infringing on the Intellectual Property owned by the Regency Entities.

3.21 Matters Relating to Acquisition of the Convertible Preferred Units.

(a) Seller has such knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of their investment in the Convertible Preferred Units and is capable of bearing the economic risk of such investment. Seller is an accredited investor as that term is defined in Rule 501 of Regulation D (without regard to Rule 501(a)(4))

15

Table of Contents

promulgated under the Securities Act. Seller is acquiring the Convertible Preferred Units for investment for its own account and not with a view toward or for sale in connection with any distribution thereof, or with any present intention of distributing or selling the Convertible Preferred Units. Seller is not a party to any Contract or arrangement with any Person to sell, transfer or grant participations to such Person or to any third Person, with respect to the Convertible Preferred Units. Seller acknowledges and understands that (i) the acquisition of the Convertible Preferred Units has not been registered under the Securities Act in reliance on an exemption therefrom and (ii) that the Convertible Preferred Units will, upon their sale by Seller, be characterized as restricted securities under state and federal securities laws. Seller agrees that the Convertible Preferred Units may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of except pursuant to an effective registration statement under the Securities Act or pursuant to an available exemption from the registration requirements of the Securities Act, and in compliance with other applicable state and federal securities laws.

(b) Seller has undertaken such investigation as they deemed necessary to enable them to make an informed and intelligent decision with respect to the execution, delivery and performance of this Agreement and the acquisition of the Convertible Preferred Units. Seller has had an opportunity to ask questions and receive answers from the Buyer Parties regarding the terms and conditions of the offering of the Convertible Preferred Units and the business, properties, prospects, and financial condition of ETE. The foregoing, however, does not modify the representations and warranties of the Buyer Parties in Article IV and such representations and warranties constitute the sole and exclusive representations and warranties of the Buyer Parties to Seller in connection with the transactions contemplated by this Agreement.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES

The Buyer Parties hereby jointly and severally represent and warrant to Seller as follows:

- 4.1 *Organization; Qualification*. Each of the Buyer Parties is an entity duly formed, validly existing and in good standing under the laws of the state of Delaware and has all requisite partnership or limited liability company power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, and are duly qualified, registered or licensed to do business as a foreign entity and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification necessary, except where the failure to be so duly qualified, registered or licensed and in good standing would not reasonably be expected to have an ETE Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair the ability of the Buyer Parties to perform their obligations under this Agreement. The Buyer Parties have made available to Seller true and complete copies of the Organizational Documents of each Buyer Party, as in effect on the Execution Date.
 - 4.2 Authority; Enforceability; Valid Issuance.
- (a) Each Buyer Party has the requisite partnership or limited liability company power and authority to execute and deliver this Agreement and to consummate the transactions contemplated by this Agreement. The execution and delivery by the Buyer Parties of this

16

Table of Contents

Agreement and the consummation by the Buyer Parties of the transactions contemplated by this Agreement have been duly and validly authorized by the Buyer Parties, and no other partnership or limited liability company proceedings on the part of the Buyer Parties are necessary to authorize this Agreement or to consummate the transactions contemplated by this Agreement.

- (b) This Agreement has been duly executed and delivered by the Buyer Parties, and, assuming the due authorization, execution and delivery by Seller, this Agreement constitutes the valid and binding agreement of the Buyer Parties, enforceable against the Buyer Parties in accordance with its terms, except as such enforceability may be limited by Creditors Rights.
- (c) The issuance of the Convertible Preferred Units and the ETE Common Units issuable upon conversion of the Convertible Preferred Units have been duly authorized in accordance with the Organizational Documents of ETE. The Convertible Preferred Units, when issued and delivered to Seller in accordance with the terms of this Agreement, and the ETE Common Units issuable upon conversion of the Convertible Preferred Units, when issued upon conversion of the Convertible Preferred Units, in each case will be validly issued, fully paid (to the extent required under the ETE Partnership Agreement), nonassessable (except to the extent nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act) and free of any restrictions upon voting or transfer thereof pursuant to the Organizational Documents of ETE or any Contract to which any of the ETE Entities is a party or by which any property or asset of any such Person is bound or affected. Upon issuance and delivery of the Convertible Preferred Units, Seller will be duly admitted to ETE as an additional limited partner.
- 4.3 Non-Contravention. Except as set forth on Schedule 4.3 of the Buyers Disclosure Schedule, the execution, delivery and performance of this Agreement and the consummation by the Buyer Parties of the transactions contemplated by this Agreement does not and will not: (a) result in any breach of any provision of the Organizational Documents of the Buyer Parties; (b) constitute a default (or an event that with notice or passage of time or both would give rise to a default) under, or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both) under any of the terms, conditions or provisions of any Contract to which any of the ETE Entities is a party or by which any property or asset of any ETE Entity is bound or affected; (c) assuming compliance with the matters referred to in Section 4.4, violate any Law to which any ETE Entity is subject or by which any of the ETE Entities properties or assets is bound or (d) constitute (with or without the giving of notice or the passage of time or both) an event which would result in the creation of any Lien (other than Permitted Liens) on any asset of any ETE Entity, except, in the cases of clauses (b), (c) and (d), for such defaults or rights of termination, cancellation, amendment, acceleration, violations or Liens as would not reasonably be expected to have an ETE Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair the Buyer Parties ability to perform their obligations under this Agreement. Except as set forth on Schedule 4.3 of the Buyers Disclosure Schedule, the ETE Entities are in material compliance with all Contracts evidencing indebtedness, including the ETE Credit Agreement.
- 4.4 *Governmental Approvals*. Except as set forth on <u>Schedule 4.4</u> of the Buyers Disclosure Schedule, no declaration, filing or registration with, or notice to, or authorization, consent or approval of, any Governmental Authority is necessary for the consummation by the

17

Table of Contents

Buyer Parties of the transactions contemplated by this Agreement, other than such declarations, filings, registrations, notices, authorizations, consents or approvals which, if not obtained or made, would not reasonably be expected to have an ETE Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair the Buyer Parties ability to perform their obligations under this Agreement.

4.5 Capitalization.

- (a) As of the Execution Date: (i) 222,941,172 ETE Common Units were issued and outstanding and (ii) 3,000,000 ETE Common Units were reserved for issuance under ETE s employee benefit plans, of which 0 ETE Common Units were subject to issuance upon exercise of outstanding ETE options and 0 ETE Common Units were subject to issuance upon the vesting of outstanding phantom units.
- (b) All of the limited partner interests in ETE are duly authorized and validly issued in accordance with the Organizational Documents of ETE, and are fully paid (to the extent required under the Organizational Documents of ETE) and nonassessable (except as nonassessability may be affected by Sections 17-303 and 17-607 of the Delaware LP Act) and were not issued in violation of any preemptive rights, rights of first refusal or other similar rights of any Person.
- (c) Except as set forth in the Organizational Documents of ETE and except as otherwise provided in Section 4.5(a), there are no preemptive rights or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, subscription agreements, commitments or rights of any kind that obligate ETE to issue or sell any equity interests of ETE or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any equity interests in ETE, and no securities or obligations evidencing such rights are authorized, issued or outstanding.
- (d) ETE does not have any outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the holders of equity interests in ETE on any matter.
- (e) L.E. GP, LLC, a Delaware limited liability company (*LEGPLLC*) is the sole general partner of ETE with a 0.3% general partner interest in ETE (the *LEGP Interest*) as of the Execution Date. The LEGP Interest has been duly authorized and validly issued in accordance with the ETE Partnership Agreement and has not been issued in violation of any preemptive rights, rights of first refusal or other similar rights of any Person. The LEGP Interest is owned by LEGPLLC free and clear of all Liens, other than (i) transfer restrictions imposed by federal and state securities laws and (ii) any transfer restrictions contained in the ETE Partnership Agreement.
- (f) As of the Execution Date, ETE owns 100% of the issued and outstanding membership interests in ETP GP LLC. Such membership interests have been duly authorized and validly issued in accordance with the Organizational Documents of ETP GP LLC and are fully paid (to the extent required under the Organizational Documents of ETP GP LLC) and non-

18

Table of Contents

assessable (except as such non-assessability may be affected by matters described in Sections 18-303 and 18-607 of the Delaware LLC Act). ETE owns such membership interests free and clear of all Liens other than (i) Liens arising under the ETE Credit Agreement, (ii) transfer restrictions imposed by federal and state securities laws and (iii) any transfer restrictions contained in the Organizational Documents of ETP GP LLC.

- (g) As of the Execution Date, ETP GP LLC is the sole general partner of ETP GP LP, with a 0.01% general partner interest in ETP GP LP; (ii) such general partner interest has been duly authorized and validly issued in accordance with the Organizational Documents of ETP GP LP; (iii) ETP GP LLC owns such general partner interest free and clear of all Liens, other than (a) Liens arising under the ETE Credit Agreement, (b) transfer restrictions imposed by federal and state securities laws and (c) any transfer restrictions contained in the Organizational Documents of ETP GP LP; (iv) ETE owns 100% of the Class A limited partner interests of ETP GP LP and 100% of the Class B limited partner interests of ETP GP LP; (v) such limited partner interests have been duly authorized and validly issued in accordance with the Organizational Documents of ETP GP LP and are fully paid (to the extent required under the Organizational Documents of ETP GP LP) and non-assessable (except as such non-assessability may be affected by Sections 17-303 and 17-607 of the Delaware LP Act); and (vi) ETE owns its limited partner interests in ETP GP LP free and clear of all Liens other than (1) Liens arising under the ETE Credit Agreement, (2) transfer restrictions imposed by federal and state securities laws and (3) any transfer restrictions contained in the Organizational Documents of ETP GP LP.
- (h) As of the Execution Date, ETP GP LP is the sole general partner of ETP with a 1.8% general partner interest in ETP (the *ETP GP Interest*) and owns the ETP Incentive Distribution Rights (collectively with the ETP GP Interest, the *ETP GP LP Interests*). The ETP GP LP Interests have been duly authorized and validly issued in accordance with the ETP Partnership Agreement and have not been issued in violation of any preemptive rights, rights of first refusal or other similar rights of any Person. As of the Execution Date, the ETP GP LP Interests are owned by ETP GP LP free and clear of all Liens, other than (i) Liens arising under the ETE Credit Agreement, (ii) transfer restrictions imposed by federal and state securities laws and (iii) any transfer restrictions contained in the ETP Partnership Agreement.
- (i) As of the Execution Date, ETE owns 62,500,797 ETP Common Units, representing approximately a 32.7% limited partner interest (collectively, the *Owned Units*), in each case free and clear of all Liens, other than (i) Liens arising under the ETE Credit Agreement, (ii) transfer restrictions imposed by federal and state securities laws and (iii) any transfer restrictions contained in the ETP Partnership Agreement. All of the Owned Units have been duly authorized and validly issued in accordance with the ETP Partnership Agreement and are fully paid (to the extent required under the ETP Partnership Agreement) and non-assessable (except as such non-assessability may be affected by Sections 17-303 and 17-607 of the Delaware LP Act).
- 4.6 *Compliance with Law*. Except as to specific matters disclosed in the ETE SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such ETE SEC Documents or any other disclosures in such ETE SEC Documents to the extent they are predictive or forward looking and general in nature), and except for Environmental Laws, Laws requiring the obtaining or

19

Table of Contents

maintenance of a Permit, Tax matters, Laws relating to employee benefits, employment and labor matters, and Laws relating to regulatory compliance matters, which are the subject of <u>Sections 4.11</u>, <u>4.13</u>, <u>4.14</u>, <u>4.15</u> and <u>4.17</u>, respectively, and except as to matters that would not reasonably be expected to have an ETE Material Adverse Effect, (a) each ETE Entity is in compliance with all applicable Laws, (b) none of the ETE Entities has received written notice of any violation of any applicable Law and (c) to the Knowledge of the Buyer Parties, none of the ETE Entities is under investigation by any Governmental Authority for potential non-compliance with any Law.

- 4.7 *Title to Properties and Assets*. To the Knowledge of the Buyer Parties, except as to matters that would not reasonably be expected to have an ETE Material Adverse Effect, each ETE Entity has title to or rights or interests in its real property and personal property free and clear of all Liens (subject to Permitted Liens), sufficient to allow it to conduct its business as currently being conducted.
- 4.8 *Rights-of-Way*. To the Knowledge of the Buyer Parties, except as to matters that would not reasonably be expected to have an ETE Material Adverse Effect, (a) each ETE Entity has such Rights-of-Way from each Person as are necessary to use, own and operate each ETE Entity s assets in the manner such assets are currently used, owned and operated by each ETE Entity, (b) each ETE Entity has fulfilled and performed all of its obligations with respect to such Rights-of-Way and (c) no event has occurred that allows, or after the giving of notice or the passage of time, or both, would allow, revocation or termination thereof or would result in any impairment of the rights of the holder of any such Rights-of-Way.
 - 4.9 ETE SEC Reports; Financial Statements; Operating Surplus.
- (a) ETE has furnished or filed all reports, schedules, forms, statements and other documents (including exhibits and other information incorporated therein) required to be furnished or filed by ETE with the SEC since January 1, 2009 (such documents being collectively referred to as the *ETE SEC Documents*).
- (b) Each ETE SEC Document (i) at the time filed, complied in all material respects with the requirements of the Exchange Act and the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such ETE SEC Document and (ii) did not at the time it was filed (or if amended or superseded by a filing or amendment prior to the Execution Date, then at the time of such filing or amendment) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (c) Each of the financial statements of ETE or LEGPLLC included in the ETE SEC Documents (*ETE Financial Statements*) complied at the time it was filed as to form in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, has been prepared in accordance with GAAP, applied on a consistent basis throughout the periods presented thereby and fairly present in all material respects the consolidated financial position and operating results, equity and cash flows of ETE as of, and for the periods ended on, the respective dates thereof, subject, however, in the

20

Table of Contents

case of unaudited financial statements, to normal year-end audit adjustments and accruals and the absence of notes and other textual disclosures as permitted by Form 10-Q of the SEC.

- (d) To the Knowledge of the Buyer Parties, none of the ETE Entities has any liability, whether accrued, contingent, absolute or otherwise, except for (i) liabilities set forth on the consolidated balance sheet of ETE dated as of March 31, 2010 or the notes thereto, and (ii) liabilities that have arisen since March 31, 2010 in the ordinary course of business.
- (e) All distributions made by ETP since its initial public offering have been made from Operating Surplus (as defined in the ETP Partnership Agreement) and in accordance with the terms of the ETP Partnership Agreement.
- (f) A true and correct copy of an ownership schedule setting forth the owners of record of the membership interests in LEGPLLC has been provided to Seller.
- 4.10 *Absence of Certain Changes*. Except as to specific matters disclosed in the ETE SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such ETE SEC Documents or any other disclosures in such ETE SEC Documents to the extent they are predictive or forward looking and general in nature), except as set forth on <u>Schedule 4.10</u> of the Buyers Disclosure Schedule or as expressly contemplated by this Agreement, since December 31, 2009, there has not been any event, occurrence or development which has had, or would be reasonably expected to have, an ETE Material Adverse Effect.
- 4.11 *Environmental Matters*. Except as to specific matters disclosed in the ETE SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such ETE SEC Documents or any other disclosures in such ETE SEC Documents to the extent they are predictive or forward looking and general in nature), except as to matters set forth on <u>Schedule 4.11</u> of the Buyers Disclosure Schedule and except as to matters that would not reasonably be expected to have an ETE Material Adverse Effect:
 - (a) each of the ETE Entities is in compliance with all applicable Environmental Laws;
- (b) each of the ETE Entities possesses all Permits required under Environmental Laws for their operations as currently conducted and is in compliance with the terms of such Permits, and such Permits are in full force and effect;
- (c) none of the ETE Entities nor any of their properties or operations are subject to any pending or, to the Knowledge of the Buyer Parties, threatened Proceeding arising under any Environmental Law, nor has any of the ETE Entities received any written and pending notice, order or complaint from any Governmental Authority alleging a violation of or liability arising under any Environmental Law; and
- (d) to the Knowledge of the Buyer Parties, there has been no Release of Hazardous Substances on, at, under, to, or from any of the properties of the ETE Entities, or from or in

21

Table of Contents

connection with the ETE Entities operations in a manner that would reasonably be expected to give rise to any liability pursuant to any Environmental Law.

- 4.12 *Legal Proceedings*. Except as to specific matters disclosed in the ETE SEC Documents filed or furnished on or after January 1, 2010 and prior to the date hereof (excluding any disclosures included in any risk factor section of such ETE SEC Documents or any other disclosures in such ETE SEC Documents to the extent they are predictive or forward looking and general in nature), and other than with respect to Proceedings arising under Environmental Laws which are the subject of Section 4.11 or as is set forth on Schedule 4.12 of Buyers Disclosure Schedule, there are no Proceedings pending or, to the Knowledge of the Buyer Parties, threatened against the ETE Entities, except such Proceedings as would not reasonably be expected to have an ETE Material Adverse Effect or to prevent or materially delay the consummation of the transactions contemplated by this Agreement or to materially impair the Buyer Parties ability to perform their obligations under this Agreement.
- 4.13 *Permits*. To the Knowledge of the Buyer Parties, other than with respect to Permits issued pursuant to or required under Environmental Laws which are the subject of <u>Section 4.11</u>, the ETE Entities have all Permits as are necessary to use, own and operate their assets in the manner such assets are currently used, owned and operated by the ETE Entities, except where the failure to have such Permits would not reasonably be expected to have an ETE Material Adverse Effect.

4.14 *Taxes*.

- (a) All material Tax Returns required to be filed with respect to ETE and, to the Knowledge of Buyer, the other ETE Entities have been filed and all the Tax Returns of ETE and, to the Knowledge of Buyer, the other ETE Entities are complete and correct in all material respects and all material Taxes due relating to ETE and, to the Knowledge of Buyer, the other ETE Entities have been paid in full. Except as disclosed on Schedule 4.14 of the Buyers Disclosure Schedule, there is no claim (other than claims being contested in good faith through appropriate proceedings and for which adequate reserves have been made in accordance with GAAP) against ETE or, to the Knowledge of Buyer, the other ETE Entities for any material Taxes, and no material assessment, deficiency, or adjustment has been asserted, proposed in writing with respect to any material Taxes or material Tax Returns of or with respect to ETE or, to the Knowledge of Buyer, the other ETE Entities.
- (b) Except as set forth on Schedule 4.14 of the Buyer Disclosure Schedule, no material Tax audits or administrative or judicial proceedings are being conducted or are pending with respect to ETE or, to the Knowledge of Buyer, the other ETE Entities.
- (c) All material Taxes required to be withheld, collected or deposited by or with respect to ETE or, to the Knowledge of Buyer, the other ETE Entities have been timely withheld, collected or deposited as the case may be, and to the extent required, have been paid to the relevant taxing authority.
- (d) There are no outstanding agreements or waivers extending the applicable statutory periods of limitations for any material Tax of, or any material Taxes associated with the

22

Table of Contents

ownership or operation of the assets of, ETE or, to the Knowledge of Buyer, any other ETE Entities.

- (e) Neither ETE nor, to the Knowledge of Buyer, any other ETE Entities is a party to any Tax sharing agreement.
- (f) Neither ETE nor, to the Knowledge of Buyer, the other ETE Entities has engaged in a transaction that would be reportable by or with respect to any ETE Entity pursuant to Treasury Regulation § 1.6011-4 or any predecessor thereto.
- (g) Neither ETE nor ETP has elected to be treated as a corporation for federal Tax purposes. ETE and ETP each qualify as a publicly traded partnership within the meaning of Section 7704(b) of the Code and each have met, and continue to meet, the gross income requirements (within the meaning of Section 7704(c) of the Code) in each Tax year since its formation. Both of ETE and ETP have filed a federal income tax return that has in effect an election pursuant to Section 754 of the Code.
 - 4.15 Employee Benefits; Employment and Labor Matters. To the Knowledge of the Buyer Parties:
- (a) Except as set forth on Schedule 4.15(a) of the Buyers Disclosure Schedule or filed with any ETE SEC Documents (including by incorporation by reference) filed with the SEC on or after January 1, 2010 and prior to the date hereof, no ETE Entity, nor any ERISA Affiliate of any ETE Entity, sponsors, maintains or contributes to, or has sponsored, maintained or contributed to within six years prior to the Closing Date any of the following:
- (i) any employee benefit plan, as such term is defined in Section 3(3) of ERISA (including, but not limited to, employee benefit plans, such as foreign plans, which are not subject to the provisions of ERISA, but excluding any multiemployer plan within the meaning of Section 3(37) of ERISA or multiple employer plan within the meaning of Section 4063(a) of ERISA); or
- (ii) any material personnel policy, equity-based plans (including, but not limited to, stock option plans, stock purchase plans, stock appreciation rights and phantom stock plans), collective bargaining agreement, bonus plan or arrangement, incentive award plan or arrangement, vacation policy, severance pay plan or arrangements, change in control policies or agreements, deferred compensation agreement or arrangement, executive compensation or supplemental income arrangement, consulting agreement, employment agreement and each other employee benefit plan, agreement, arrangement, program, practice or understanding which is not described in Section 4.15(a)(i) (collectively, along with any plan described in Section 4.15(a)(i) above, the ETE Benefit Plans).
- (b) Except as disclosed on <u>Schedule 4.15(b)</u> of the Buyers Disclosure Schedule and except for matters that would not reasonably be expected to have a ETE Material Adverse Effect;

23

Table of Contents

- (i) each ETE Benefit Plan has been administered in compliance with its terms, the applicable provisions of ERISA, the Code and all other applicable laws and the terms of all applicable collective bargaining agreements;
- (ii) there are no actions, suits or claims pending (other than routine claims for benefits) or threatened, with respect to any ETE Benefit Plan and no ETE Benefit Plan is under audit or is subject to an investigation by the Internal Revenue Service, the Department of Labor or any other federal or state governmental agency nor is any such audit or investigation pending;
 - (iii) no ETE Benefit Plan is subject to Title IV of ERISA;
- (iv) all contributions and payments required to be made by any ETE Entity or an ERISA Affiliate of any ETE Entity to or under each ETE Benefit Plan have been timely made;
- (v) as to any ETE Benefit Plan intended to be qualified under Section 401 of the Code, there has been no termination or partial termination of such plan within the meaning of Section 411(d)(3) of the Code;
- (vi) none of the ETE Entities or any of their ERISA Affiliates have any liability with respect to any multiemployer plan within the meaning of Section 3(37) of ERISA or multiple employer plan with the meaning of Section 4063(a) of ERISA.
- (c) Except as would not reasonably be expected to have an ETE Material Adverse Effect, (a) each of the ETE Entities is in compliance with all applicable labor and employment Laws including, without limitation, all Laws, rules, regulations, orders, rulings, decrees, judgments and awards relating to employment discrimination, payment of wages, overtime compensation, immigration, occupational health and safety, and wrongful discharge; (b) no action, suit, complaint, charge, arbitration, inquiry, proceeding or investigation by or before any Governmental Authority, brought by or on behalf of any employee, prospective or former employee or labor organization or other representative of the employees or of any prospective or former employees of any of the ETE Entities is pending or threatened against any of the ETE Entities, any present or former director or employee (including with respect to alleged sexual harassment, unfair labor practices or discrimination); (c) no grievance is pending or threatened against any of the ETE Entities; and (d) none of the ETE Entities is subject to or otherwise bound by, any material consent decree, order, or agreement with, any Governmental Authority relating to employees or former employees of any of the ETE Entities.
- 4.16 **Brokers Fee.** Except for the fee payable to Credit Suisse Securities (USA) LLC, which shall be paid by the Buyer Parties, no broker, investment banker, financial advisor or other Person is entitled to any broker s, finder s, financial advisor s or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Buyer Parties.

24

Table of Contents

4.17 Regulatory Matters; Compliance.

- (a) The ETE Entities (i) have all necessary approvals from FERC to provide service to customers pursuant to the Natural Gas Act and the Natural Gas Policy Act of 1978, as amended, and (ii) have made all required FERC filings necessary to offer such service, except where the failure to have any such approval or to have made any such filing would not reasonably be expected to have an ETE Material Adverse Effect.
- (b) Neither ETE nor ETP is an investment company or a company controlled by an investment company within the meaning of the Investment Company Act of 1940, as amended.
- 4.18 *Intellectual Property*. To the Knowledge of the Buyer Parties (a) the ETE Entities own or have the right to use pursuant to license, sublicense, agreement or otherwise all material items of Intellectual Property required in the operation of the business as presently conducted; (b) no third party has asserted in writing delivered to the ETE Entities an unresolved claim that any of the ETE Entities is infringing on the Intellectual Property of such third party; and (c) no third party is infringing on the Intellectual Property owned by the ETE Entities.

4.19 Matters Relating to Acquisition of the Acquired GP Interests.

- (a) The Buyer Parties have such knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of their investment in the Acquired GP Interests and are capable of bearing the economic risk of such investment. The Buyer Parties are accredited investors as that term is defined in Rule 501 of Regulation D (without regard to Rule 501(a)(4)) promulgated under the Securities Act. The Buyer Parties are acquiring the Acquired GP Interests for investment for their own account and not with a view toward or for sale in connection with any distribution thereof, or with any present intention of distributing or selling the Acquired GP Interests. The Buyer Parties do not have any Contract or arrangement with any Person to sell, transfer or grant participations to such Person or to any third Person, with respect to the Acquired GP Interests. The Buyer Parties acknowledge and understand that (i) the acquisition of the Acquired GP Interests has not been registered under the Securities Act in reliance on an exemption therefrom and (ii) that the Acquired GP Interests will, upon its sale by Seller, be characterized as restricted securities under state and federal securities laws. The Buyer Parties agree that the Acquired GP Interests may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of except pursuant to an effective registration statement under the Securities Act or pursuant to an available exemption from the registration requirements of the Securities Act, and in compliance with other applicable state and federal securities laws.
- (b) The Buyer Parties have undertaken such investigation as they have deemed necessary to enable them to make an informed and intelligent decision with respect to the execution, delivery and performance of this Agreement and the acquisition of the Acquired GP Interests. The Buyer Parties have had an opportunity to ask questions and receive answers from Seller regarding the terms and conditions of the offering of the Acquired GP Interests and the business, properties, prospects, and financial condition of Regency. The foregoing, however, does not modify the representations and warranties of Seller in Article III and such

25

Table of Contents

representations and warranties constitute the sole and exclusive representations and warranties of Seller to the Buyer Parties in connection with the transactions contemplated by this Agreement.

ARTICLE V COVENANTS OF THE PARTIES

5.1 Conduct of Business.

- (a) From the Execution Date through the Closing, except as described in <u>Schedule 5.1</u> of the Seller Disclosure Schedule, and except as required by this Agreement or consented to or approved in writing by the Buyer Parties (which shall not be unreasonably withheld, conditioned or delayed), Seller shall cause the Regency GP Entities to, and shall use reasonable best efforts to cause each other Regency Entity to:
 - (i) conduct its business and activities in the ordinary course of business consistent with past practice;
- (ii) use reasonable best efforts to preserve intact their goodwill and relationships with customers, suppliers and others having business dealings with them with respect thereto;
 - (iii) comply in all material respects with all applicable Laws relating to them; and
- (iv) use reasonable best efforts to maintain in full force without interruption its present insurance policies or comparable insurance coverage of the Regency Entities.
- (b) Without limiting the generality of Section 5.1(a), and, except as described in Schedule 5.1(b) of the Seller Disclosure Schedule, as required by this Agreement or consented to or approved in writing by the Buyer Parties (which shall not be unreasonably withheld, conditioned or delayed), Seller shall cause the Regency GP Entities not to, and shall use reasonable best efforts to cause each other Regency Entity not to:
 - (i) make any material change or amendment to its Organizational Documents;
- (ii) purchase any securities or ownership interests of, or make any investment in any Person, other than in respect of the Regency Entities (excluding the Regency GP Entities) (A) ordinary course overnight investments consistent with the cash management policies of such Person and (B) purchases and investments in addition to those contemplated by clause (A) not in excess of \$50,000,000 in the aggregate;
- (iii) make any capital expenditure; *provided* that the Regency Entities (excluding the Regency GP Entities) may make capital expenditures that are not in excess of \$50,000,000 in the aggregate or as required on an emergency basis or for the safety of individuals or the environment;

(iv) make any material change to its tax methods, principles or elections;

26

Table of Contents

- (v) except as required under its Organizational Documents, declare or pay any distributions in respect of any of its equity securities or partnership units except (A) in the case of Regency, the declaration and payment of regular quarterly cash distributions of Available Cash from Operating Surplus (each as defined in the Regency Partnership Agreement) not in excess of \$0.46 per Regency Common Unit per quarter, plus any corresponding distribution on the general partner interest and Regency Incentive Distribution Rights, (B) in the case of Regency, regular quarterly distributions of Available Cash from Operating Surplus, not in excess of \$0.445 per quarter in respect of the Regency Series A Units and (C) the declaration and payment of distributions from any direct or indirect wholly owned Subsidiary of Regency;
- (vi) split, combine or reclassify any of its equity securities or partnership units or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for, its equity securities or partnership units, except for any such transaction by a direct or indirect wholly owned Subsidiary of Regency that remains a direct or indirect wholly owned Subsidiary of Regency after consummation of such transaction;
- (vii) repurchase, redeem or otherwise acquire any of its equity securities or partnership units or any securities convertible into or exercisable for any equity securities or partnership units other than redemptions to satisfy federal income tax withholding obligations (calculated using the applicable federal income tax rates) in connection with the vesting of units under Regency s long-term incentive plan;
- (viii) issue, deliver, sell, pledge or dispose of, or authorize the issuance, delivery, sale, pledge or disposition of, any (A) equity securities or partnership units of any class, (B) debt securities having the right to vote on any matters on which holders of capital stock or members or partners of the same issuer may vote or (C) securities convertible into or exercisable for, or any rights, warrants, calls or options to acquire, any such securities, other than issuances (1) by a direct or indirect wholly owned Subsidiary of Regency of equity securities or partnership units to such Person s parent or any other direct or indirect wholly owned Subsidiary of Regency and (2) pursuant to awards outstanding prior to the Execution Date under Regency Benefit Plans which are reflected on Schedule 3.17(a) of the Seller Disclosure Schedule;
- (ix) purchase or sell assets (including any general partner or limited partner interest or any other equity interests in any other Person) or waive any rights or benefits held by either of the Regency GP Entities attributable to RGPLP s ownership of the general partner interest in Regency and the Regency Incentive Distribution Rights, other than purchases or sales of inventory in the ordinary course by any of the Regency Entities (excluding the Regency GP Entities), with a value not exceeding \$50,000,000 individually or \$50,000,000 in the aggregate;
- (x) create, incur, guarantee or assume any indebtedness for borrowed money other than borrowings by the Regency Entities (excluding the Regency GP Entities) of less than \$100,000,000 in the aggregate;

27

Table of Contents

- (xi) enter into any joint venture or similar arrangement with a third party other than joint ventures or similar arrangements entered into by any of the Regency Entities (excluding the Regency GP Entities) in the ordinary course of business including assets or obligations of less than \$75,000,000;
- (xii) (A) settle any claims, demands, lawsuits or state or federal regulatory proceedings for damages to the extent such settlements assess damages in excess of \$10,000,000 in the aggregate (other than any claims, demands, lawsuits or proceedings to the extent insured (net of deductibles), reserved against in the Regency Financial Statements or covered by an indemnity obligation not subject to dispute or adjustment from a solvent indemnitor) or (B) settle any claims, demands, lawsuits or state or federal regulatory proceedings seeking an injunction or other equitable relief where such settlements would have or would reasonably be expected to have a Regency Material Adverse Effect;
- (xiii) except as otherwise expressly permitted under this <u>Section 5.1</u>, merge with or into, or consolidate with, any other Person or acquire all or substantially all of the business or assets of any other Person;
- (xiv) take any action with respect to or in contemplation of any liquidation, dissolution, recapitalization, reorganization, or other winding up;
- (xv) change or modify any accounting policies, except as required by applicable regulatory authorities or independent accountants;
- (xvi) approve or make material modifications of the salaries, bonuses or other compensation (including incentive compensation) payable to any individual whose base salary exceeds \$200,000 per annum or adopt or make any material amendment to any employee compensation, benefit or incentive plans;
- (xvii) modify, make any material amendment to or voluntarily terminate, prior to the expiration date thereof, any Regency Material Contracts or waive any default by, or release, settle or compromise any claim against, any other party thereto; or
 - (xviii) agree, or commit to take any of the actions described above.

Notwithstanding anything in this Agreement to the contrary, nothing in <u>Section 5.1</u> shall prohibit Regency or any of its Subsidiaries from taking any action, or Seller from approving the taking by Regency or any of its Subsidiaries of any action, in each case that would otherwise be prohibited by this <u>Section 5.1</u> without consent or approval of the Buyer Parties, if, prior to taking such action, or approving the taking of such action, Seller or Regency, as applicable, determines in good faith, after consultation with outside legal counsel, that failure to take such action, or to approve the taking of such action, would be reasonably likely to be a breach of the implied contractual covenant of good faith and fair dealing owed to an unaffiliated third party and imposed on Seller, Regency or such Subsidiary, as applicable, in its capacity as the general partner of Regency and a party to the Regency Partnership Agreement or a partner in the RIGS JV and a party to the RIGS JV Agreement under the Delaware Revised Uniform Partnership Act, as applicable.

28

Table of Contents

- (c) From the Execution Date though the Closing, ETE shall not: (i) amend the ETP Redemption Agreement in any manner that would reasonably be expected to adversely affect Seller s rights under this Agreement or (ii) exercise its rights under Section 7.1(a) of the ETP Redemption Agreement to terminate the ETP Redemption Agreement,
- (d) From the Execution Date through the Closing, except as described in <u>Schedule 5.1(d)</u> of the Buyers Disclosure Schedule, or consented to or approved in writing by Seller (which shall not be unreasonably withheld, conditioned or delayed), the Buyer Parties shall not take any action that would require the consent of Seller pursuant to <u>Section 5.13(b)(v)</u> of the Third Amendment (which for purposes of this <u>Section 5.1(d)</u> shall be deemed to be in full force and effect).

5.2 Notice of Certain Events.

- (a) Subject to applicable Law, each Party shall promptly notify the other Parties of:
- (i) any event, condition or development that has resulted in the inaccuracy or breach of any representation or warranty, covenant or agreement contained in this Agreement made by or to be complied with by such notifying Party at any time during the term hereof and that would reasonably be expected to result in any of the conditions set forth in Article VI not to be satisfied and which notice shall identify the applicable representation or warranty, covenant or agreement and disclosure schedule, if any, for which such breach or inaccuracy relates; *provided*, *however*, that no such notification shall be deemed to cure any such breach of or inaccuracy in such notifying Party s representations and warranties or covenants and agreements or in the Seller Disclosure Schedule or the Buyers Disclosure Schedule for any purpose under this Agreement and no such notification shall limit or otherwise affect the remedies available to the other Parties;
- (ii) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement;
- (iii) any notice or other communication from any Governmental Authority in connection with the transactions contemplated by this Agreement; or
- (iv) any Proceedings commenced that would be reasonably expected to prevent or materially delay the consummation of the transactions contemplated by this Agreement or materially impair the notifying Party s ability to perform its obligations under this Agreement.
- (b) Seller shall promptly notify the Buyer Parties of the occurrence of any of the events described in <u>Section 5.1(b)</u> regardless of whether Seller used reasonable best efforts to prevent the occurrence of such event.
- 5.3 *Access to Information*. From the Execution Date until the Closing Date, Seller on the one hand, and Buyer Parties, on the other hand, will subject to compliance with Law governing the use of such information, (a) give the other party and their counsel, financial

29

Table of Contents

advisors, auditors and other authorized representatives (collectively, *Representatives*) reasonable access to the offices, properties, books and records of the Regency Entities or ETE Entities, as the case may be, and permit such party to make copies thereof, in each case during normal business hours and (b) furnish such financial and operating data and other information relating to the Regency Entities or ETE Entities, as the case may be, as such Persons may reasonably request. For two (2) years after the Closing Date, Seller, its Affiliates and its Representatives shall have reasonable access during normal business hours to the offices, properties, books and records of the Buyer Entities with respect to information of the nature and scope described on Schedule 5.3, and which the Buyer Parties acknowledge and agree Seller may share with its Representatives; provided that Seller and its applicable Affiliates and Representatives shall have entered into a confidentiality agreement relating to such information reasonably acceptable to the Buyer Parties. Any investigation pursuant to this Section 5.3 shall be conducted in such manner as not to interfere with the conduct of the business of any Party. Notwithstanding the foregoing, no Party shall be entitled to perform any intrusive or subsurface investigation or other sampling of, on or under any of the properties of a Party without the prior written consent of the other Party. Notwithstanding the foregoing provisions of this Section 5.3, no Party shall be required to grant access or furnish information to the extent that such information is subject to an attorney/client or attorney work product privilege or that such access or the furnishing of such information is prohibited by Law or an existing Contract. To the extent practicable, such Party shall make reasonable and appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply, including the execution of a joint defense agreement to allow the parties to exchange information protected by the attorney client privilege or work product doctrine. To the fullest extent permitted by Law, (1) no party nor any of their respective Representatives or Affiliates shall be responsible or liable to another party for personal injuries sustained in connection with the access provided pursuant to this Section 5.3 and (2) shall be indemnified and held harmless by the visiting party for any losses suffered by any such Persons in connection with any such personal injuries; provided such personal injuries are not caused by the gross negligence or willful misconduct of the hosting party. The Parties agree that they will not, and will cause their Representatives not to, use any information obtained pursuant to this Section 5.3 for any purpose unrelated to the consummation of the transactions contemplated by this Agreement.

5.4 Governmental Approvals.

- (a) The Parties will cooperate with each other and use reasonable best efforts to obtain from any Governmental Authorities any consents, licenses, permits, waivers, approvals, authorizations or orders required to be obtained and to make any filings with or notifications or submissions to any Governmental Authority that are necessary in order to consummate the transactions contemplated by the Transaction Agreements and shall diligently and expeditiously prosecute, and shall cooperate fully with each other in the prosecution of, such matters.
- (b) The Parties agree to cooperate with each other and use reasonable best efforts to contest and resist, any Proceeding, and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order (whether temporary, preliminary or permanent) of any Governmental Authority that is in effect and that restricts, prevents or prohibits the consummation of the transactions contemplated by the Transaction Agreements.

30

Table of Contents

- 5.5 *Expenses*. All costs and expenses incurred by Seller in connection with this Agreement and the transactions contemplated thereby, and any documented, out-of-pocket expenses reasonably incurred by the other Regency Entities (not to exceed \$50,000), shall be paid by Seller, and all costs and expenses incurred by the Buyer Parties or the ETE Entities in connection with this Agreement and the transactions contemplated thereby shall be paid by the Buyer Parties; *provided, however*, that if any action at law or equity is necessary to enforce or interpret the terms of this Agreement, the prevailing Party shall be entitled to reasonable attorneys fees and expenses in addition to any other relief to which such Party may be entitled.
- 5.6 Further Assurances. Subject to the terms and conditions of this Agreement, each of the Parties shall use its reasonable best efforts to take, or cause to be taken, all action, and to do, or cause to be done, all things necessary, proper or advisable under applicable Law to consummate the transactions contemplated by the Transaction Agreements. Without limiting the generality of the foregoing, each Party will use its reasonable best efforts to obtain timely all authorizations, consents and approvals of all third parties necessary in connection with the consummation of the transactions contemplated by the Transaction Agreements prior to the Closing. The Parties will coordinate and cooperate with each other in exchanging such information and assistance as any of the Parties hereto may reasonably request in connection with the foregoing.
- 5.7 *Public Statements*. The Parties shall consult with each other prior to issuing any public announcement, statement or other disclosure with respect to this Agreement or the transactions contemplated thereby and neither Seller nor its Affiliates, on one hand, nor the Buyer Parties and their Affiliates, on the other hand, shall issue any such public announcement, statement or other disclosure without having first notified Seller, on one hand, or the Buyer Parties on the other; *provided, however*, that any of Seller and its Affiliates, on one hand, and any of the Buyer Parties and their Affiliates, on the other hand, may make any public disclosure without first so consulting with or notifying the other Party or Parties if such disclosing party believes that it is required to do so by Law or by any stock exchange listing requirement or trading agreement concerning the publicly traded securities of Seller or its Affiliates, on one hand, or the Buyer Parties or any of their Affiliates, on the other hand.
- 5.8 *Convertible Preferred Units*. Seller agrees to the imprinting, so long as the restrictions described in the legend are applicable, of the following legend on any certificates evidencing all or any portion of the Convertible Preferred Units or any ETE Common Units issuable upon conversion thereof:

THESE SECURITIES HAVE NOT BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE SECURITIES ACT) AND ARE SUBJECT TO THE TERMS OF THE THIRD AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT OF ENERGY TRANSFER EQUITY, L.P., AS AMENDED, THE HOLDER OF THIS SECURITY ACKNOWLEDGES FOR THE BENEFIT OF ENERGY TRANSFER EQUITY, L.P. THAT THIS SECURITY MAY NOT BE SOLD, OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED IF SUCH TRANSFER WOULD (A) VIOLATE THE THEN APPLICABLE FEDERAL OR STATE SECURITIES LAWS OR

31

Table of Contents

RULES AND REGULATIONS OF THE SECURITIES AND EXCHANGE COMMISSION, ANY STATE SECURITIES COMMISSION OR ANY OTHER GOVERNMENTAL AUTHORITY WITH JURISDICTION OVER SUCH TRANSFER, (B) TERMINATE THE EXISTENCE OR QUALIFICATION OF ENERGY TRANSFER EQUITY, L.P. UNDER THE LAWS OF THE STATE OF DELAWARE, OR (C) CAUSE ENERGY TRANSFER EQUITY, L.P. TO BE TREATED AS AN ASSOCIATION TAXABLE AS A CORPORATION OR OTHERWISE TO BE TAXED AS AN ENTITY FOR FEDERAL INCOME TAX PURPOSES (TO THE EXTENT NOT ALREADY SO TREATED OR TAXED). L.E. GP, THE GENERAL PARTNER OF ENERGY TRANSFER EQUITY, L.P., MAY IMPOSE ADDITIONAL RESTRICTIONS ON THE TRANSFER OF THIS SECURITY IF IT RECEIVES AN OPINION OF COUNSEL THAT SUCH RESTRICTIONS ARE NECESSARY TO AVOID A SIGNIFICANT RISK OF ENERGY TRANSFER EQUITY, L.P. BECOMING TAXABLE AS A CORPORATION OR OTHERWISE BECOMING TAXABLE AS AN ENTITY FOR FEDERAL INCOME TAX PURPOSES. THE RESTRICTIONS SET FORTH ABOVE SHALL NOT PRECLUDE THE SETTLEMENT OF ANY TRANSACTIONS INVOLVING THIS SECURITY ENTERED INTO THROUGH THE FACILITIES OF ANY NATIONAL SECURITIES EXCHANGE ON WHICH THIS SECURITY IS LISTED OR ADMITTED TO TRADING.

5.9 No Solicitation.

- (a) Neither Seller nor the Regency GP Entities or their respective officers, directors, employees, stockholders, representatives, agents, or anyone acting on behalf of them, shall, directly or indirectly, (i) encourage, solicit, engage in discussions or negotiations with, or provide any information to, any Person (other than the Buyer Parties or their Representatives) concerning any Competing Proposal, (ii) accept any offer or respond to any indications of interest from any Person concerning any Competing Proposal, (iii) enter into an agreement, arrangement or understanding with any Person other than the Buyer Parties or their Affiliates concerning any Competing Proposal or (iv) make or authorize any statement, recommendation or solicitation in support of or concerning or otherwise facilitate any purchase or sale of securities or similar transaction involving any Competing Proposal; *provided*, *however*, that this Section 5.9 shall in no way prohibit the board of directors of RGPLLC, in its capacity as the general partner of RGPLP, in its capacity as the general partner of Regency, from taking any action required by its fiduciary duty.
- (b) At every meeting of limited partners of Regency called with respect to any of the following, and at every adjournment or postponement thereof, and on every action or approval by written consent of limited partners of Regency with respect to any of the following, Seller shall and shall cause the Regency GP Entities and any of its other Affiliates to vote:
- (i) against approval of any proposal made in opposition to, or in competition with, consummation of the transactions contemplated by this Agreement, including, without limitation, any proposal to remove RGPLP as the general partner of Regency; and

(ii) except as otherwise agreed to in writing in advance by the Buyer Parties, against any Competing Proposal.

32

Table of Contents

(c) For purposes of this Agreement, *Competing Proposal* means any contract, proposal, offer or indication of interest relating to any transaction or series of transactions involving: (i) any merger, amalgamation, unit exchange, recapitalization, consolidation, liquidation or dissolution of any of the Regency Entities, (ii) any direct or indirect acquisition of beneficial ownership (as defined under Section 13(d) of the Exchange Act) by any Person or group of any equity interests in any of the Regency GP Entities, (iii) any direct or indirect acquisition of beneficial ownership (as defined under Section 13(d) of the Exchange Act) by any Person or group of all or any portion of the general partner interest in Regency, (iv) any direct or indirect acquisition of beneficial ownership (as defined under Section 13(d) of the Exchange Act) by any Person or group of 15% or more of the limited partner interests in Regency or any tender or exchange offer that if consummated would result in any Person or group beneficially owning 15% or more of the limited partner interests in Regency, (v) any direct or indirect acquisition (by asset purchase, unit purchase, merger or otherwise) by any Person or group (as defined under Section 13(d) of the Exchange Act) of any business or material amount of assets of the Regency Entities (including capital stock of or ownership interest in any Subsidiary of Regency or any of its Subsidiaries) (other than sales of assets by the Regency Entities in the ordinary course of business consistent with past practice) or (vi) any transaction that would compete with or serve to interfere with, delay, discourage, adversely affect or inhibit the timely consummation of the transactions contemplated by this Agreement.

5.10 Confidential Information.

- (a) For a period of two (2) years after the Closing, Seller and its Affiliates shall not, directly or indirectly, disclose to any Person any secret, confidential or proprietary business information, data or material developed by, or on behalf of, any Regency Entity relating to the business and operations of the Regency Entities, whether acquired prior to or after the Closing Date, which has not been disclosed to the public.
- (b) Notwithstanding the foregoing, Seller or its Affiliates may disclose any information relating to the business and operations of the Regency Entities, if required by Law or applicable stock exchange rule, or to such other Persons if, at the time such information is provided, such Person is already in the possession of such information.
- 5.11 *No Hire*. Except for the individuals listed on <u>Schedule 5.11</u>, for a period of one year from and after the Execution Date, neither Seller nor the Energy Financial Service business unit of GECC shall solicit for employment or hire any executive officers or other management level employees of any of the Regency Entities who were employed by the Regency Entities within six months prior to the Closing Date. The restrictions in this Section 5.11 regarding the prohibition on solicitations (as opposed to hires) shall not apply to any solicitation directed at the general public.

5.12 Tax Matters.

(a) <u>Post-Closing Tax Returns</u>. Buyer shall cause the Regency Entities to prepare all Tax Returns relating to the Regency Entities for periods beginning on or before the Closing Date and ending after the Closing Date. With respect to any such Tax Returns for the Regency GP Entities, Buyer shall determine (by an interim closing of the books as of the Closing Date except

33

Table of Contents

for ad valorem and property taxes owed or owing by the Regency GP Entities, which shall be prorated on a daily basis) the Taxes that would have been due with respect to the period covered by such Tax Return if such taxable period ended on and included the Closing Date (the *Pre-Closing Tax*).

- (i) Not later than 20 days prior to the due date of any estimated Tax payment relating to any Pre-Closing Tax, Buyer shall deliver to Seller for its review a statement calculating the excess, if any, of the Pre-Closing Tax included in such payment over the amount set up as a liability for such Tax on the financial statements of Regency Entities. Buyer shall make or cause to be made such changes in such statement as Seller may reasonably request, which changes shall be subject to Buyer s approval, which shall not be unreasonably withheld. Thereafter, and not later than 5 days prior to the due date of such estimated Tax payment, Seller shall pay to Buyer the amount of such excess.
- (ii) Not later than 20 days prior to the due date of any Tax Return covering a Pre-Closing Tax, Buyer shall deliver to Seller for its review a copy of such Tax Return and a statement calculating the amount by which the Pre-Closing Tax reflected on such Tax Return is greater than or less than the amount set up as a liability for such Tax on the financial statements of Regency Entities and the amount of any payments paid by Seller to Buyer with respect to estimated Tax payments of such Pre-Closing Tax pursuant to Section 5.12(a)(i), which amount of estimated Tax payments shall be treated as a credit against Pre-Closing Tax owed to Buyer by Seller. Buyer shall make or cause to be made such changes in such Tax Returns or such statement as Seller may reasonably request, which changes shall be subject to Buyer s approval, which shall not be unreasonably withheld. Not later than 5 days prior to the due date of such Tax Return, Seller shall pay to Buyer (or Buyer pay to Seller, if appropriate) the amount of such difference. Upon receipt thereof, Buyer shall file or cause to be filed such Tax Return and shall pay all Taxes shown to be due thereon.
- (b) <u>Transfer Taxes</u>. All excise, sales, use, transfer (including real property transfer or gains), stamp, documentary, filing, recordation and other similar taxes, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties, resulting directly from the transactions contemplated by this Agreement (the **Transfer Taxes**), shall be borne 50% by Seller and 50% by the Buyer. Notwithstanding anything to the contrary in this <u>Section 5.12</u>, any Tax Returns that must be filed in connection with Transfer Taxes shall be prepared and filed when due by the party primarily or customarily responsible under the applicable local law for filing such Tax Returns, and such party will use reasonable best efforts to provide such Tax Returns to the other party at least ten days prior to the due date for such Tax Returns. Upon the filing of Tax Returns in connection with Transfer Taxes, the filing party shall provide the other party with evidence satisfactory to the other party that such Transfer Taxes have been filed and paid.
 - (c) Cooperation on Tax Matters.
- (i) Buyer and Seller Parties shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes for taxable periods

34

Table of Contents

beginning on or before the Closing Date. Such cooperation shall include the retention until the expiration of the relevant statute of limitations and (upon the other Party's request) the provision of records and information in such Party's possession that are reasonably relevant to any such audit, litigation or other proceeding and making employees available on the basis of reasonable best efforts to provide additional information and explanation of any material provided hereunder. Prior to the destruction or discarding of any books and records with respect to Tax matters pertinent to the Regency Entities relating to any taxable period beginning on or before the Closing Date, each Party shall give the other Party reasonable written notice and, if the other Party so requests, shall itself allow, or cause the Regency Entities to allow the other Party to take, possession of such books and records. In connection with any audit, litigation or other proceeding with respect to Taxes for taxable periods beginning on or before the Closing Date, Buyer and Seller shall promptly notify each other upon receipt by such party of written notice of any inquiries, claims, assessments, audits, or similar events. Buyer shall have sole control of the conduct of all such audit, litigation or other proceedings with respect to Taxes for periods beginning on or before the Closing Date, including any settlement or compromise thereof, *provided, however*, Buyer shall keep Seller reasonably informed of the progress of any such audit, litigation or other proceeding and shall not effect any such settlement or compromise with respect to which Seller is liable without obtaining Seller s prior written consent thereto, which shall not be unreasonably withheld.

- (ii) Buyer and Seller Parties further agree, upon request, to use their reasonable best efforts to obtain any certificate or other document from any Governmental Authority or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed as a result of the transactions contemplated hereby or for any taxable period beginning on or before the Closing Date.
 - 5.13 Books and Records; Financial Statements; Litigation Support.
- (a) Seller shall provide the Buyer Parties access to Seller s books and records relating to the Regency Entities to the extent reasonably necessary to enable the Buyer Parties to prepare financial statements of the Regency Entities and such other financial statement of the Buyer Parties and their Affiliates in such forms and covering such periods as may be required by any applicable securities laws to be filed with the SEC by ETE as a result of the transactions contemplated by this Agreement. Seller shall use reasonable best efforts to cause their independent accountants and the Regency Entities independent accountants to provide any consent necessary to the filing of such financial statements with the SEC and to provide such customary representation letters as are necessary in connection therewith.
- (b) Seller hereby consents to the inclusion or incorporation by reference of the financial statements of the Regency Entities in any registration statement, report or other filing of the Buyer Parties or any of their Affiliates as to which the Buyer Parties or any of their Affiliates reasonably determines that such financial statements are required to be included or incorporated by reference to satisfy any rule or regulation of the SEC or to satisfy relevant disclosure obligations under the Securities Act or the Exchange Act. Seller shall use reasonable best efforts to cause the Audit Firm to consent to the inclusion or incorporation by reference of its audit

35

Table of Contents

opinion with respect to any of the financial statements of the Regency Parties in any such registration statement, report or other filing of the Buyer Parties or their Affiliates, and Seller shall cause representation letters, in form and substance reasonably satisfactory to the Audit Firm, to be executed and delivered to the Audit Firm in connection with obtaining any such consent from the Audit Firm.

- (c) Seller shall cooperate with the Buyer Parties in connection with the preparation of any proforma financial statements of the Buyer Parties or any of their Affiliates that are derived in part from the financial statements of Seller that Buyer or their Affiliates reasonably determines are required to be included or incorporated by reference in any registration statement, report or other filing of the Buyer Parties or their Affiliates to satisfy any rule or regulation of the SEC or to satisfy relevant disclosure obligations under the Securities Act or the Exchange Act.
- (d) Seller shall provide access to its books and records as may be reasonably necessary for the Buyer Parties or any of their Affiliates, or any of their respective advisors or representatives, to conduct customary due diligence with respect to the financial statements of Seller in connection with any offering of securities by the Buyer Parties or any of their Affiliates or to enable an accounting firm to prepare and deliver a customary comfort letter with respect to financial information relating to Seller.
- (e) In the event and for so long as any Party actively is contesting or defending against any third-party Proceeding (other than any Proceedings in which the Buyer Parties or any of their Affiliates and the Seller or any of its Affiliates are adverse parties) in connection with (i) the transactions contemplated by this Agreement or (ii) any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act, or transaction on or prior to the Closing Date involving the Regency Entities, each of the other Parties will cooperate with it and its counsel in the contest or defense, make available their personnel, and provide such testimony and access to their books and records as shall be reasonably requested and necessary in connection with the contest or defense, all at the sole cost and expense of the contesting or defending Party; *provided, however*, that nothing in this Section 5.13(e) shall limit in any respect any rights a Party may have with respect to discovery or the production of documents or other information in connection with any such litigation..
 - 5.14 Commitment Regarding Indemnification Provisions; D&O Insurance Continuation.
- (a) Buyer Parties covenant and agree that during the period that commences on the Closing Date and ends on the sixth (6th) anniversary of the Closing Date, Buyer Parties (i) shall not cause any amendment, modification, waiver or termination to Section 7.7 or Section 7.8 of the Regency Partnership Agreement and (ii) shall not amend, modify, waive or terminate Section 9.01 or Section 9.02 of the RGPLLC Agreement, the effect of which would be to affect adversely the rights of any person serving as a member of the Board of Directors or officer of RGPLLC existing as of the date of this Agreement under such provisions; *provided*, *however*, that the foregoing restriction shall not apply to any such amendment, modification, waiver or termination to the extent required to cause such provisions (or any portion thereof) to comply with applicable law.

36

Table of Contents

- (b) Buyer Parties covenant and agree that, during the period that commences on the Closing Date and ends on the sixth (6th) anniversary of the Closing Date, with respect to any person serving as a member of the Board of Directors or officer of RGPLLC as of the date of this Agreement and any former member of the Board of Directors or officer of RGPLLC appointed by Seller, Buyer Parties shall cause RGPLLC (i) to continue in effect the current director and officer liability insurance policy or policies that RGPLLC has as of the date of this Agreement, as reflected on Schedule 5.14 hereto, or (ii) upon the termination or cancellation of any such policy or policies, (A) to provide director and officer liability insurance in substitution for, or in replacement of, such cancelled or terminated policy or policies or (B) to provide a tail or run-off policy, in each case so that any person serving as a member of the Board of Directors of RGPLLC as of the date of this Agreement, any former member of the Board of Directors appointed by Seller and any officer has coverage thereunder for acts, events, occurrences or omissions occurring or arising at or prior to the Closing to the same extent (including, without limitation, policy limits, exclusions and scope) as such person has coverage for such acts, events, occurrences or omissions under the director and officer insurance policy maintained by RGPLLC as of the date of this Agreement, as reflected on Schedule 5.14 hereto.
- 5.15 *GECC Names and Marks*. From and after the Closing Date, the Buyer Parties shall not use any of the GECC Names and Marks.

ARTICLE VI CONDITIONS TO CLOSING

- 6.1 *Conditions to Obligations of Each Party*. The respective obligation of each Party to consummate the Closing is subject to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any one or more of which may be waived in writing, in whole or in part, as to a Party by such Party (in such Party s sole discretion):
- (a) <u>Approvals</u>. All authorizations, consents, orders or approvals of, or declarations or filings with, or as set forth on Schedule 6.1(a) shall have been obtained or made.
- (b) <u>Governmental Restraints</u>. No order, decree or injunction of any Governmental Authority shall be in effect, and no Law shall have been enacted or adopted that enjoins, prohibits or makes illegal the consummation of the transactions contemplated by this Agreement and no Proceeding by any Governmental Authority with respect to the transactions contemplated by this Agreement shall be pending that seeks to restrain, enjoin, prohibit or delay the transactions contemplated by this Agreement.
- (c) <u>Amendment to Regency Credit Agreement</u>. The Regency Credit Agreement shall have been amended in a form reasonably acceptable to the Buyer Parties to address the matters set forth on <u>Schedule 6.1(c)</u>.
- (d) <u>Amendment to ETE Credit Agreement</u>. The ETE Credit Agreement shall have been amended in a form reasonably acceptable to the Buyer Parties to address the matters set forth on <u>Schedule 6.1(d)</u>.
- (e) <u>Redemption and Exchange</u>. All conditions to consummation of the Redemption and Exchange shall have been satisfied or validly waived pursuant to the terms of the ETP

37

Table of Contents

Redemption Agreement and each of the parties thereto shall have executed and delivered to the Parties a certificate stating that such parties thereto will consummate the Redemption and Exchange Agreement immediately following the Closing.

- (f) <u>MEP Contribution</u>. All conditions to consummation of the MEP Contribution shall have been satisfied and validly waived pursuant to the terms of the MEP Contribution Agreement and each of the parties thereto shall have executed and delivered to the Parties a certificate stating that such parties thereto will consummate the MEP Contribution Agreement immediately following the Closing.
- 6.2 *Conditions to Obligations of the Buyer Parties*. The obligation of the Buyer Parties to consummate the Closing is subject to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any one or more of which may be waived in writing, in whole or in part, by the Buyer Parties (in the Buyer Parties sole discretion):
- (a) <u>Representations and Warranties of Seller</u>. The representations and warranties of Seller (i) in <u>Article III</u> (other than those contained in <u>Sections 3.5</u> and <u>3.6</u>) shall be true and correct in all respects as of the Closing Date as if remade on the Closing Date (except for representations and warranties made as of a specific date, which shall be true and correct in all respects as of such specific date), with only such failures to be so true and correct as had not had, and would not reasonably be expected to have, a Regency Material Adverse Effect and (ii) in <u>Sections 3.5</u> and <u>3.6</u> shall be true and correct in all material respects as of the Closing Date as if remade on the Closing Date (except for representations and warranties contained therein made as of a specific date, which shall be true and correct in all material respects as of such specific date).
- (b) <u>Performance</u>. Seller shall have performed and complied in all material respects with all covenants and agreements required by this Agreement to be performed or complied with by Seller on or prior to the Closing Date.
- (c) <u>Closing Certificate</u>. The Buyer Parties shall have received a certificate, dated as of the Closing Date, signed by a Responsible Officer of Seller certifying that, to the best of such Responsible Officer s knowledge, the conditions set forth in <u>Sections 6.2(a)</u> and <u>6.2(b)</u> have been satisfied.
- (d) <u>Closing Deliverables</u>. Seller shall have delivered or caused to be delivered all of the closing deliveries set forth in <u>Section 2.3(a)</u> and in the other documents contemplated by this Agreement.
- 6.3 *Conditions to Obligations of Seller*. The obligation of Seller to consummate the Closing is subject to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any one or more of which may be waived in writing, in whole or in part, by Seller (in the Seller s sole discretion):
- (a) <u>Representations and Warranties of the Buyer Parties</u>. The representations and warranties of the Buyer Parties (i) in <u>Article IV</u> (other than those contained in <u>Section 4.5</u>) shall be true and correct in all respects as of the Closing Date as if remade on the Closing Date (except for representations and warranties made as of a specific date, which shall be true and correct in all respects as of such specific date), with only such failures to be so true and correct as had not

38

Table of Contents

had, and would not reasonably be expected to have, an ETE Material Adverse Effect and (ii) in <u>Sections 4.5</u> shall be true and correct in all material respects as of the Closing Date as if remade on the Closing Date (except for representations and warranties contained therein made as of a specific date, which shall be true and correct in all material respects as of such specific date).

- (b) <u>Performance</u>. The Buyer Parties shall have performed and complied in all material respects with all covenants and agreements required by this Agreement to be performed or complied with by the Buyer Parties on or prior to the Closing Date.
- (c) <u>Closing Certificate</u>. Seller shall have received a certificate, dated as of the Closing Date, signed by a Responsible Officer of the Buyer Parties certifying that, to the best of such Responsible Officer s knowledge, the conditions set forth in <u>Sections 6.3(a)</u> and <u>6.3(b)</u> have been satisfied.
- (d) <u>Closing Deliverables</u>. The Buyer Parties shall have delivered or caused to be delivered all of the closing deliveries set forth in <u>Section 2.3(b)</u> and in the other documents contemplated by this Agreement.

ARTICLE VII TERMINATION RIGHTS

- 7.1 *Termination Rights*. This Agreement may be terminated at any time prior to the Closing as follows:
- (a) By mutual written consent of the Parties:
- (b) By either Seller or the Buyer Parties if any Governmental Authority of competent jurisdiction shall have issued a final and non-appealable order, decree or judgment prohibiting the consummation of the transactions contemplated by this Agreement;
- (c) By either Seller or the Buyer Parties in the event that the Closing has not occurred on or prior to June 9, 2010 (the *Termination Date*); *provided, however*, that (i) Seller may not terminate this Agreement pursuant to this Section 7.1(c) if such failure of the Closing to occur is due to the failure of Seller to perform and comply in all material respects with the covenants and agreements to be performed or complied with by Seller and (ii) the Buyer Parties may not terminate this Agreement pursuant to this Section 7.1(c) if such failure of the Closing to occur is due to the failure of either Buyer Party to perform and comply in all material respects with the covenants and agreements to be performed or complied with by such Buyer Party;
- (d) By the Buyer Parties if there shall have been a breach or inaccuracy of Seller's representations and warranties in this Agreement or a failure by Seller to perform its covenants and agreements in this Agreement, in any such case in a manner that would result in, if occurring and continuing on the Closing Date, the failure of the conditions to the Closing set forth in Section 6.2(a) or Section 6.2(b), unless such failure is reasonably capable of being cured, and Seller is using all reasonable efforts to cure such failure by the Termination Date; provided, however, that the Buyer Parties may not terminate this Agreement pursuant to this Section 7.1(d) if (i) any of the Buyer Parties representations and warranties shall have become and continue to be untrue in a manner that would cause the condition set forth in Section 6.3(a) not to be satisfied

39

Table of Contents

- or (ii) there has been, and continues to be, a failure by either Buyer Party to perform its covenants and agreements in such a manner as would cause the condition set forth in <u>Section 6.3(b)</u> not to be satisfied;
- (e) By Seller if there shall have been a breach or inaccuracy of the Buyer Parties representations and warranties in this Agreement or a failure by either Buyer Party to perform its covenants and agreements in this Agreement, in any such case in a manner that would result in, if occurring and continuing on the Closing Date, the failure of the conditions to the Closing set forth in Section 6.3(a) or Section 6.3(b), unless such failure is reasonably capable of being cured, and such Buyer Party is using all reasonable efforts to cure such failure by the Termination Date; provided, however, that Seller may not terminate this Agreement pursuant to this Section 7.1(e) if (i) Seller s representations and warranties shall have become and continue to be untrue in a manner that would cause the condition set forth in Section 6.2(a) not to be satisfied or (ii) there has been, and continues to be, a failure by Seller to perform its covenants and agreements in such a manner as would cause the condition set forth in Section 6.2(b) not to be satisfied; or
- (f) By the Buyer Parties if either of the ETP Redemption Agreement or the MEP Contribution Agreement has been terminated by the counterparty to such agreement pursuant to its terms.
- 7.2 Effect of Termination. In the event of the termination of this Agreement pursuant to Section 7.1, all rights and obligations of the Parties under this Agreement shall terminate, except for the provisions of this Section 7.2, Article IX and Sections 5.5, 5.7, 10.1, 10.3, 10.6, 10.8 and 10.9, the last sentence of Section 5.3; provided, however, that no termination of this Agreement shall relieve any Party from any liability for any willful and intentional breach of this Agreement by such Party or for Fraud by such Party and all rights and remedies of a non-breaching Party under this Agreement in the case of any such willful and intentional breach or Fraud, at law and in equity, shall be preserved, including the right to recover reasonable attorneys fees and expenses. In the event of the termination of this Agreement, pursuant to Section 7.1, the Parties agree that for a period of one year from and after the Execution Date, neither Seller, on the one hand, nor the Buyer Parties, on the other hand, shall solicit for employment or hire any executive officers or other management level employees of the Regency Entities, in the case of the Buyer Parties, and the Buyer Parties, in the case of Seller, who were employed by such party within six months prior to the Execution Date. The restrictions in the preceding sentence regarding the prohibition on solicitations (as opposed to hires) shall not apply to any solicitation directed at the general public. Except to the extent otherwise provided in this Section 7.2, the Parties agree that, if this Agreement is terminated, the Parties shall have no liability to each other under or relating to this Agreement.

ARTICLE VIII INDEMNIFICATION

8.1 *Indemnification by Seller*. Subject to the terms of this <u>Article VIII</u>, from and after the Closing, Seller shall jointly and severally indemnify and hold harmless the Buyer Parties and their respective partners, members, managers, directors, officers, employees, consultants and permitted assigns (collectively, the *Buyer Indemnitees*), to the fullest extent permitted by Law, from and against any losses (excluding any loss in the value of the Regency

40

Table of Contents

Common Units issued to ETE pursuant to the Contribution Agreement), claims, damages, liabilities and costs and expenses (including reasonable attorneys fees and expenses) (collectively, *Losses*) incurred, arising out of or relating to:

- (a) any breach of any of the representations or warranties (in each case, when made) of Seller contained in <u>Article III</u> or of the certification of a Responsible Officer of Seller delivered to the Buyer Parties pursuant to Section 6.2(c); and
- (b) any breach of any of the covenants or agreements of Seller contained in this Agreement. *provided however*; that any liability of Seller under this <u>Section 8.1</u> shall be satisfied at Seller s option from (i) cash on hand; or (ii) by a redemption of the Convertible Preferred Units at a price based off the original issue price of the Convertible Preferred Units (subject to adjustment from time to time for stock dividends, stock splits, combinations of units, reorganizations, recapitalizations, reclassifications or other similar events occurring after the date hereof) plus all accrued and unpaid dividends (whether or not declared), in an amount equal to such Losses.
- 8.2 *Indemnification by the Buyer Parties*. Subject to the terms of this <u>Article VIII</u>, from and after the Closing, the Buyer Parties shall jointly and severally indemnify and hold harmless Seller and its directors, officers, employees, consultants and permitted assigns (collectively, the *Seller Indemnitees* and, together with the Buyer Indemnitees, the *Indemnitees*), to the fullest extent permitted by Law, from and against Losses incurred, arising out of or relating to:
- (a) any breach of any of the representations or warranties (in each case, when made) of the Buyer Parties contained in this Agreement or of the certification of a Responsible Officer of the Buyer Parties delivered to Seller pursuant to Section 6.3(c); and
 - (b) any breach of any of the covenants or agreements of the Buyer Parties contained in this Agreement.
- 8.3 *Limitations and Other Indemnity Claim Matters*. Notwithstanding anything to the contrary in this <u>Article VIII</u> or elsewhere in this Agreement, the following terms shall apply to any claim for monetary damages arising out of this Agreement or related to the transactions contemplated hereby:
- (a) <u>De Minimis</u>. No indemnifying party (an *Indemnifying Party*) will have any liability under this <u>Article VIII</u> in respect of any individual claim involving Losses arising under <u>Section 8.1(a)</u> or <u>Section 8.2(a)</u> to any single Buyer Indemnitee or Seller Indemnitee, as applicable, of less than \$125,000 (each, a *De Minimis Claim*). Notwithstanding the forgoing, this <u>Section 8.3(a)</u> shall not apply to Losses arising from any breach or inaccuracy of the representations or warranties set forth in <u>Section 3.16</u> or <u>Section 4.14</u>.
 - (b) Deductible.
- (i) Seller will not have any liability under <u>Section 8.1(a)</u> until the Buyer Indemnitees have suffered Losses in excess of in the aggregate \$3,000,000 (the

41

Table of Contents

Deductible) arising from Claims under Section 8.1(a) that are not De Minimis Claims, and then recoverable Losses claimed under Section 8.1(a) shall be limited to those that exceed the Deductible. Notwithstanding the forgoing, this Section 8.3(b)(i) shall not apply to Losses arising from any breach or inaccuracy of the representations or warranties set forth in Section 3.16.

- (ii) The Buyer Parties will not have any liability under <u>Section 8.2(a)</u> until the Seller Indemnitees have suffered Losses in excess of the Deductible arising from Claims under <u>Section 8.2(a)</u> that are not De Minimis Claims, and then recoverable Losses claimed under <u>Section 8.2(a)</u> shall be limited to those that exceed the Deductible. Notwithstanding the forgoing, this <u>Section 8.3(b)(ii)</u> shall not apply to Losses arising from any breach or inaccuracy of the representations or warranties set forth in <u>Section 4.14</u>.
 - (c) <u>Cap</u>.
- (i) Seller s aggregate liability under this Agreement and from the transactions contemplated hereby shall not exceed \$45,000,000 (the *Cap*); *provided* that the limitation set forth in this Section 8.3(c)(i) shall not apply to Losses arising out of or relating to (A) any breach or inaccuracy of the representations and warranties set forth in Sections 3.1, 3.2, 3.5, 3.6 or 3.18 or (B) any breach of any covenants or agreements of Seller set forth in this Agreement that by their terms are to be performed after the Closing Date; *provided*, *further*, that in no event shall Seller s aggregate liability arising under this Agreement and from the transactions contemplated hereby exceed \$300,000,000.
- (ii) The Buyer Parties aggregate liability under this Agreement and from the transactions contemplated hereby shall not exceed the Cap; *provided* that the limitation set forth in this Section 8.3(c)(ii) shall not apply to Losses arising out of or relating to (A) any breach or inaccuracy of the representations and warranties set forth in Sections 4.1, 4.2 or 4.16 or (B) any breach of any covenants or agreements of the Buyer Parties set forth in this Agreement that by their terms are to be performed after the Closing Date; *provided*, *further*, that in no event shall the Buyer Parties aggregate liability arising under this Agreement and from the transactions contemplated hereby exceed \$300,000,000.
 - (d) Survival; Claims Period.
- (i) The representations, warranties, covenants and agreements of the Parties under this Agreement shall survive the execution and delivery of this Agreement and shall continue in full force and effect until the one-year anniversary of the Closing Date (the *Expiration Date*); *provided* that (i) the representations and warranties set forth in Sections 3.1 (Organization; Qualification), 3.2 (Authority; Enforceability), 3.4 (Governmental Approvals), 3.5 (Capitalization), 3.6 (Ownership of Acquired GP Interests), 3.18 (Brokers Fee), 3.22 (Matters Relating to Acquisition of the Convertible Preferred Units), 4.1 (Organization; Qualification), 4.2 (Authority; Enforceability; Valid Issuance), 4.4 (Governmental Approvals), 4.5 (Capitalization), 4.16 (Brokers Fee) and 4.19 (Matters Relating to Acquisition of Acquired GP Interests) shall survive indefinitely,

42

Table of Contents

- (ii) the representations and warranties set forth in Section 3.16 and Section 4.14 shall survive the execution and delivery of this Agreement and shall continue in full force and effect until ninety (90) days after the expiration of the applicable statute of limitations (which shall be deemed to be the Expiration Date with respect to such representations and warranties) and (iii) any covenants or agreements contained in this Agreement that by their terms are to be performed after the Closing Date shall survive until fully discharged.
- (ii) No action for a breach of any representation or warranty contained herein (other than representations or warranties that survive indefinitely pursuant to <u>Section 8.3(d)(i)</u>) shall be brought after the Expiration Date, except for claims of which a Party has received a Claim Notice setting forth in reasonable detail the claimed misrepresentation or breach of warranty with reasonable detail, prior to the Expiration Date.
- (e) *Calculation of Losses*. In calculating amounts payable to any Seller Indemnitee or Buyer Indemnitee (each such person, an *Indemnified Party*) for a claim for indemnification hereunder, the amount of any indemnified Losses shall be determined without duplication of any other Loss for which an indemnification claim has been made or could be made under any other representation, warranty, covenant, or agreement and shall be computed net of (i) payments actually recovered by the Indemnified Party under any insurance policy with respect to such Losses and (ii) any prior or subsequent actual recovery by the Indemnified Party from any Person with respect to such Losses.
- (f) <u>Waiver of Certain Damages</u>. Notwithstanding any other provision of this Agreement, in no event shall any Party be liable for punitive, special, indirect, consequential, remote, speculative or lost profits damages of any kind or nature, regardless of the form of action through which such damages are sought, except for any such damages recovered by any third party against an Indemnified Party in respect of which such Indemnified Party would otherwise be entitled to indemnification pursuant to the terms hereof.
- (g) <u>Sole and Exclusive Remedy</u>. Except for the assertion of any Claim based on fraud with the intent to deceive or willful misconduct, the remedies provided in this <u>Article VIII</u> shall be the sole and exclusive legal remedies of the Parties, from and after the Closing, with respect to this Agreement and the transactions contemplated hereby.

8.4 Indemnification Procedures.

(a) Each Indemnitee agrees that promptly after it becomes aware of facts giving rise to a claim by it for indemnification pursuant to this Article VIII, such Indemnitee must assert its claim for indemnification under this Article VIII (each, a *Claim*) by providing a written notice (a *Claim Notice*) to the Indemnifying Party allegedly required to provide indemnification protection under this Article VIII specifying, in reasonable detail, the nature and basis for such Claim (e.g., the underlying representation, warranty, covenant or agreement alleged to have been breached) and the amount (to the extent that the nature and amount of such Claim is known or reasonably ascertainable at such time, *provided* that such amount or estimated amount shall not be conclusive of the final amount, if any, of such Claim). Notwithstanding the foregoing, an Indemnitee s failure to send or delay in sending a third party Claim Notice will not relieve the

43

Table of Contents

Indemnifying Party from liability hereunder with respect to such Claim except to the extent the Indemnifying Party is prejudiced by such failure or delay and except as is otherwise provided herein, including in <u>Section 8.3(d</u>).

- (b) In the event of the assertion of any third party Claim for which, by the terms hereof, an Indemnifying Party is obligated to indemnify an Indemnitee, the Indemnifying Party will have the right, at such Indemnifying Party s expense, to assume the defense of same including the appointment and selection of counsel on behalf of the Indemnitee so long as such counsel is reasonably acceptable to the Indemnitee. If the Indemnifying Party elects to assume the defense of any such third party Claim, it shall within 30 days of its receipt of the Claim Notice, notify the Indemnitee in writing of its intent to do so. The Indemnifying Party will have the right to settle or compromise or take any corrective or remediation action with respect to any such Claim by all appropriate proceedings, which proceedings will be diligently prosecuted by the Indemnifying Party to a final conclusion or settled at the discretion of the Indemnifying Party. The Indemnitee will be entitled, at its own cost, to participate with the Indemnifying Party in the defense of any such Claim. If the Indemnifying Party assumes the defense of any such third-party Claim but fails to diligently prosecute such Claim, or if the Indemnifying Party does not assume the defense of any such Claim, the Indemnitee may assume control of such defense and in the event it is determined pursuant to the procedures set forth in Article IX that the Claim was a matter for which the Indemnifying Party is required to provide indemnification under the terms of this Article VIII, the Indemnifying Party will bear the reasonable costs and expenses of such defense (including reasonable attorneys fees and expenses). Notwithstanding the foregoing, the Indemnifying Party may not assume the defense of the third-party Claim (but will be entitled at its own cost to participate with the Indemnified Party in the defense of any such Claim) if the potential damages under the third-party Claim could reasonably and in good faith be expected to exceed, in the aggregate when combined with all claims previously made by the Indemnified Party to the Indemnifying Party under this Article VIII, the maximum amount the Indemnifying Party may be liable pursuant to Section 8.3(c); provided, however, that to the extent the Parties are not in agreement with respect to the calculation of potential damages, the Indemnifying Party shall have the right to assume the defense of the third-party Claim in accordance herewith until the Parties have agreed or a final non-appealable judgment has been entered into, with respect to the determination of the potential damages.
- (c) Notwithstanding anything to the contrary in this Agreement, the Indemnifying Party will not be permitted to settle, compromise, take any corrective or remedial action or enter into an agreed judgment or consent decree, in each case, that subjects the Indemnified Party to any criminal liability, requires an admission of guilt or wrongdoing on the part of the Indemnified Party or imposes any continuing obligation on or requires any payment from the Indemnified Party without the Indemnified Party s prior written consent.

8.5 No Reliance.

(a) THE REPRESENTATIONS AND WARRANTIES OF SELLER CONTAINED IN <u>ARTICLE III</u> CONSTITUTE THE SOLE AND EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF SELLER TO THE BUYER PARTIES IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. THE REPRESENTATIONS OF THE BUYER PARTIES CONTAINED IN <u>ARTICLE IV</u> CONSTITUTE THE SOLE AND

44

Table of Contents

EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES TO SELLER IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EXCEPT FOR SUCH REPRESENTATIONS AND WARRANTIES, NO PARTY NOR ANY OTHER PERSON MAKES ANY OTHER EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY WITH RESPECT TO SUCH PARTY OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT, AND EACH PARTY DISCLAIMS ANY OTHER REPRESENTATIONS OR WARRANTIES. WHETHER MADE BY SUCH PARTY OR ANY OF ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS OR REPRESENTATIVES (INCLUDING WITH RESPECT TO THE DISTRIBUTION OF, OR ANY PERSON S RELIANCE ON, ANY INFORMATION, DISCLOSURE OR OTHER DOCUMENT OR OTHER MATERIAL MADE AVAILABLE TO THE ANY PARTY IN ANY DATA ROOM, ELECTRONIC DATA ROOM, MANAGEMENT PRESENTATION OR IN ANY OTHER FORM IN EXPECTATION OF, OR IN CONNECTION WITH, THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT). EXCEPT FOR SUCH REPRESENTATIONS AND WARRANTIES, EACH PARTY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, PROJECTION, FORECAST, STATEMENT, OR INFORMATION MADE, COMMUNICATED, OR FURNISHED (ORALLY OR IN WRITING) TO ANY OTHER PARTY OR ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS OR REPRESENTATIVES (INCLUDING OPINION, INFORMATION, PROJECTION, OR ADVICE THAT MAY HAVE BEEN OR MAY BE PROVIDED TO ANY PARTY OR ANY OFFICER, DIRECTOR, EMPLOYEE, AGENT OR REPRESENTATIVE OF SUCH PARTY OR ANY OF ITS AFFILIATES).

(b) Except as provided in <u>Sections 7.2</u>, <u>8.1</u> and <u>8.2</u>, no Party nor any Affiliate of a Party shall assert or threaten, and each Party hereby waives and shall cause such Affiliates to waive, any claim or other method of recovery, in contract, in tort or under applicable Law, against any Person that is not a Party (or a successor to a Party) relating to the transactions contemplated by this Agreement.

ARTICLE IX

GOVERNING LAW AND CONSENT TO JURISDICTION

- 9.1 *Governing Law*. This Agreement shall be governed by and construed and interpreted in accordance with the Laws of the State of Delaware, without giving effect to the conflicts of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware.
- 9.2 Consent to Jurisdiction. The Parties irrevocably submit to the exclusive jurisdiction of (a) the Delaware Court of Chancery, and (b) any state appellate court therefrom within the State of Delaware (or, only if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), for the purposes of any Proceeding arising out of this Agreement or the transactions contemplated hereby (and each agrees that no such Proceeding relating to this Agreement or the transactions contemplated hereby shall be brought by it except in such courts). The Parties irrevocably and unconditionally waive (and agree not to plead or claim) any objection to the laying of venue of any Proceeding arising out of this Agreement or the transactions contemplated hereby in (i) the

45

Table of Contents

Delaware Court of Chancery, or (ii) any state appellate court therefrom within the State of Delaware (or, only if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware) or that any such Proceeding brought in any such court has been brought in an inconvenient forum. Each of the Parties hereto also agrees that any final and non-appealable judgment against a Party hereto in connection with any Proceeding shall be conclusive and binding on such Party and that such award or judgment may be enforced in any court of competent jurisdiction, either within or outside of the United States. A certified or exemplified copy of such award or judgment shall be conclusive evidence of the fact and amount of such award or judgment.

9.3 Waiver of Jury Trial. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY ACTION OR PROCEEDING TO ENFORCE OR TO DEFEND ANY RIGHTS UNDER THIS AGREEMENT SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

ARTICLE X GENERAL PROVISIONS

- 10.1 *Amendment and Modification*. This Agreement may be amended, modified or supplemented only by written agreement of the Parties hereto.
- 10.2 *Waiver of Compliance; Consents*. Except as otherwise provided in this Agreement, any failure of any of the Parties to comply with any obligation, covenant, agreement or condition in this Agreement may be waived by the Party or Parties entitled to the benefits thereof only by a written instrument signed by the Party or Parties granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.
- 10.3 *Notices*. Any notice, demand or communication required or permitted under this Agreement shall be in writing and delivered personally, by reputable overnight delivery service or other courier or by certified mail, postage prepaid, return receipt requested, and shall be deemed to have been duly given (a) as of the date of delivery if delivered personally or by overnight delivery service or other courier or (b) on the date receipt is acknowledged if delivered by certified mail, addressed as follows; *provided* that a notice of a change of address shall be effective only upon receipt thereof:

If to Seller to:

Regency GP Acquirer LP c/o GE Energy Financial Services

800 Long Ridge Road Stamford, Connecticut 06927

Telephone: (203) 316-7355 Facsimile: (203) 961-2606 Attention: Portfolio-Regency

46

Table of Contents

With a copy (not itself constituting notice) to:

Regency GP Acquirer LP

c/o GE Energy Financial Services

800 Long Ridge Road

Stamford, Connecticut 06927 Telephone: (203) 357-4151 Facsimile: (203) 357-6632 Attention: General Counsel

and

Latham & Watkins LLP

885 Third Avenue

New York, New York 10022 Telephone: (212) 906-1259 Facsimile: (212) 751-4864 Attention: Charles E. Carpenter If to the Buyer Parties to: Energy Transfer Equity, L.P.

3738 Oak Lawn Dallas, Texas 75219

Telephone: (832) 668-1210 or (214) 981-0763

Facsimile: (832) 668-1127 Attention: General Counsel

and

Vinson & Elkins LLP 2500 First City Tower 1001 Fannin, Suite 2500 Houston, Texas 77007 Telephone: (713) 758-3613 Facsimile: (713) 615-5725

Attention: Douglas E. McWilliams

10.4 *Assignment*. This Agreement shall be binding upon and inure to the benefit of the Parties and their successors and permitted assigns. No Party may assign or transfer this Agreement or any of its rights, interests or obligations under this Agreement without the prior written consent of the other Parties. Any attempted assignment or transfer in violation of this Agreement shall be null, void and ineffective.

10.5 *Third Party Beneficiaries*. This Agreement shall be binding upon and inure solely to the benefit of the Parties hereto and their respective successors and assigns. Except as

47

Table of Contents

provided in <u>Sections 8.1</u> and <u>8.2</u>, none of the provisions of this Agreement shall be for the benefit of or enforceable by any third party, including any creditor of any Party or any of their Affiliates. No such third party shall obtain any right under any provision of this Agreement or shall by reasons of any such provision make any claim in respect of any liability (or otherwise) against any other Party.

- 10.6 *Entire Agreement*. Except for the Confidentiality Agreement which shall survive the execution of this Agreement, this Agreement and the other transaction documents constitute the entire agreement and understanding of the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both oral and written, among the Parties or between any of them with respect to such subject matter.
- 10.7 *Severability*. Whenever possible, each provision or portion of any provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable Law but if any provision or portion of any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability will not affect any other provision or portion of any provision in such jurisdiction, and this Agreement will be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision or portion of any provision had never been contained herein.
- 10.8 *Representation by Counsel*. Each of the Parties agrees that it has been represented by independent counsel of its choice during the negotiation and execution of this Agreement and the documents referred to herein, and that it has executed the same upon the advice of such independent counsel. Each Party and its counsel cooperated in the drafting and preparation of this Agreement and the documents referred to herein, and any and all drafts relating thereto shall be deemed the work product of the Parties and may not be construed against any Party by reason of its preparation. Therefore, the Parties waive the application of any Law providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document.
- 10.9 *Disclosure Schedules*. The inclusion of any information (including dollar amounts) in any section of the Seller Disclosure Schedule or the Buyers Disclosure Schedule shall not be deemed to be an admission or acknowledgment by a Party that such information is required to be listed on such section of the Seller Disclosure Schedule or the Buyers Disclosure Schedule or is material to or outside the ordinary course of the business of such Party or the Person to which such disclosure relates. The information contained in this Agreement, the Exhibits and the Schedules is disclosed solely for purposes of this Agreement, and no information contained in this Agreement, the Exhibits or the Schedules shall be deemed to be an admission by any Party to any third Person of any matter whatsoever (including any violation of a legal requirement or breach of contract). The disclosure contained in one disclosure schedule contained in the Seller Disclosure Schedule or Buyer Disclosure Schedule may be incorporated by reference into any other disclosure schedule contained therein, and shall be deemed to have been so incorporated into any other disclosure schedule so long as it is readily apparent on its face that the disclosure is applicable to such other disclosure schedule.

48

Table of Contents

10.10 *Facsimiles; Counterparts*. This Agreement may be executed by facsimile signatures by any Party and such signature shall be deemed binding for all purposes hereof, without delivery of an original signature being thereafter required. This Agreement may be executed in one or more counterparts, each of which, when executed, shall be deemed to be an original and all of which together shall constitute one and the same document.

[Signature page follows.]

49

Table of Contents

IN WITNESS WHEREOF, each of the Parties has caused this Agreement to be executed by its respective duly authorized officers as of the date first above written.

REGENCY GP ACQUIRER LP

By: REGENCY GP HOLDO I LLC, its general partner

By: AIRCRAFT SERVICES CORPORATION, its managing member

By: /s/ Mark Mellana Mark Mellana, Authorized Signatory

ENERGY TRANSFER EQUITY, L.P.

By: LE GP, LLC, its general partner

By: /s/ John W. McReynolds John W. McReynolds, President and Chief Financial Officer

ETE GP ACQUIRER LLC

By: /s/ Martin Salinas, Jr. Martin Salinas, Jr., Chief Financial Officer

Signature Page to General Partner Purchase Agreement

EXHIBIT A

Acquired GP Interests is defined in the recitals to this Agreement.

Affiliate means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a specified Person. A Person shall be deemed to control another Person if such first Person possesses, directly or indirectly, the power to direct, or cause the direction of, the management and policies of such other Person, whether through the ownership of voting securities, by contract or otherwise.

Agreement is defined in the preamble to this Agreement.

Assignment of Interests is defined in Section 2.3(a)(i).

Audit Firm means the independent accounting firm regularly engaged by Seller to review their quarterly financial statements and provide an audit report with respect to their annual financial statements.

Business Day means any day that is not a Saturday, Sunday or other day on which commercial banks in the State of New York are authorized or obligated to be closed by applicable Laws.

Buyer is defined in the preamble to this Agreement.

Buyer Indemnities is defined in Section 8.1.

Buyer Parties is defined in the preamble to this Agreement.

Buyers Disclosure Schedule means the disclosure schedule to this Agreement prepared by the Buyer Parties and delivered to Seller on the Execution Date.

Claim is defined in Section 8.4(a).

Claim Notice is defined in Section 8.4(a).

Closing is defined in Section 2.2.

Closing Date is defined in Section 2.2.

Code means the Internal Revenue Code of 1986, as amended.

Competing Proposal is defined in Section 5.9(c).

Confidentiality Agreement means that certain Confidentiality and Non-Disclosure Agreement, dated as of February 8, 2010, by and among ETE, ETP and GE Energy Financial Services, Inc.

Contract means any written agreement, lease, license, note, evidence of indebtedness, mortgage, security agreement, understanding, instrument or other legally binding arrangement.

Control means, where used with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of Voting Interests, by contract or otherwise, and the terms Controlling and Controlled have correlative meanings.

Convertible Preferred Units is defined in the recitals to this Agreement.

Creditors Rights is defined in Section 3.2(b).

1

De Minimis Claim is defined in Section 8.3(a).

Deductible is defined in Section 8.3(b)(i).

Delaware LLC Act means the Delaware Limited Liability Company Act, as amended from time to time.

Delaware LP Act means the Delaware Revised Uniform Limited Partnership Act, as amended from time to time.

Environmental Laws means any and all Laws pertaining to prevention of pollution, protection of the environment (including natural resources), remediation of contamination and workplace health and safety.

ERISA means the Employee Retirement Income Security Act of 1974, as amended.

ERISA Affiliate means, with respect to any entity, trade or business, any other entity, trade or business that is a member of a group described in Section 414(b),(c), (m) or (o) of the Code or Section 4001(b)(l) of ERISA that includes the first entity, trade or business, or that is a member of the same controlled group as the first entity, trade or business pursuant to section 4001(a)(14) of ERISA.

ETC II is defined in the recitals to this Agreement.

ETC III is defined in the recitals to this Agreement.

ETE is defined in the preamble to this Agreement.

ETE Benefit Plans is defined in Section 4.15.

ETE Common Units means a common unit representing a limited partner interest in ETE.

ETE Credit Agreement means that certain Credit Agreement dated February 8, 2006 between ETE and Wachovia Bank, National Association, as administrative agent as amended from time to time.

ETE Entities means ETE and all Subsidiaries of ETE.

ETE Financial Statements are defined in Section 4.9(c).

ETE Material Adverse Effect means any Material Adverse Effect in respect to the ETE Entities taken as a whole.

ETE Partnership Agreement means the Third Amended and Restated Agreement of Limited Partnership of Energy Transfer Equity, L.P., as amended from time to time.

ETE SEC Documents is defined in Section 4.9(a).

ETP is defined in the recitals to this Agreement.

ETP Common Units means a common unit representing a limited partner interest in ETP.

ETP GP Interest is defined in Section 4.5(h).

ETP GP LLC means Energy Transfer Partners, L.L.C., a Delaware limited liability company.

ETP GP LP means Energy Transfer Partners GP, L.P., a Delaware limited partnership.

A-2

ETP GP LP Interests is defined in Section 4.5(h).

ETP Incentive Distribution Rights means the Incentive Distribution Rights as such term is defined in the ETP Partnership Agreement.

ETP Partnership Agreement means the Amended and Restated Agreement of Limited Partnership of Energy Transfer Partners, L.P. as amended from time to time.

ETP Redemption Agreement is defined in the recitals to this Agreement.

Exchange Act means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

Execution Date is defined in the preamble to this Agreement.

Expiration Date is defined in Section 8.3(d)(i).

FERC means the Federal Energy Regulatory Commission of the United States of America.

Fraud means actual fraud involving a knowing and intentional misrepresentation of a material fact.

GAAP means generally accepted accounting principles in the United States of America.

GECC Names and Marks means the names or marks of GECC or any of its Affiliates, including GE (in block letters or otherwise), the GE monogram, General Electric Company and General Electric, either alone or in combination with other words and all marks, trade dress, logos, monograms, domain names and other source identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words; provided, however, that GECC Names and Marks shall not include the names or marks of Regency or any of its Subsidiaries, including Regency or Regency Energy Partners, either alone or in combination with other words and all marks, trade dress, logos, monograms, domain names and other source identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words.

Governance Policies means policies designed to address conflicts of interest, non-overlapping director positions, confidential information and other related matters.

Governmental Authority means any executive, legislative, judicial, regulatory or administrative agency, body, commission, department, board, court, tribunal, arbitrating body or authority of the United States or any foreign country, or any state, local or other governmental subdivision thereof.

Hazardous Substances means each substance, waste or material regulated, defined, designated or classified as a hazardous waste, hazardous substance, hazardous material, pollutant, contaminant or toxic substance under any Environmental Law.

Indemnified Party is defined in Section 8.3(e).

Indemnifying Party is defined in Section 8.3(a).

Indemnitees is defined in Section 8.2.

Intellectual Property means patents, trademarks, copyrights, and trade secrets.

A-3

Table of Contents

Knowledge means (a) with respect to Seller, the actual knowledge of James F. Burgoyne, Daniel R. Castagnola, Mark T. Mellana and Brian P. Ward, and (b) with respect to the Buyer Parties, the actual knowledge of John W. McReynolds and Sonia Aube.

Law means any law, statute, code, ordinance, order, rule, rule of common law, regulation, judgment, decree, injunction, franchise, permit, certificate, license or authorization of any Governmental Authority.

LEGP Interest is defined in Section 4.5(e).

LEGPLLC is defined in Section 4.5(e).

Lien means, with respect to any property or asset, (i) any mortgage, pledge, security interest, lien or other similar property interest or encumbrance in respect of such property or asset, and (ii) any easements, rights-of-way, restrictive covenants, rights, leases and other encumbrances on title to real or personal property (whether or not of record).

Losses is defined in Section 8.1.

Master Services Agreement is defined in the recitals to this Agreement.

Material Adverse Effect means, with respect to any Person, any change, event or development that is materially adverse to the business, assets, financial condition, or operations of such Person and its Subsidiaries, taken as a whole; provided, however, that, a Material Adverse Effect shall not be deemed to have occurred as a result of any of the following changes, events or developments (either alone or in combination): (a) any change in general economic, political or business conditions (including any effects on the economy arising as a result of acts of terrorism); (b) any change in oil or natural gas commodity prices; (c) any change affecting the natural gas transportation industry generally but which does not have a materially disproportionate impact on the business of such Person and its Subsidiaries; (d) any change in accounting requirements or principles imposed by GAAP or any change in Law after the Execution Date but which does not, in each case, have a materially disproportionate impact on the business of such Person and its Subsidiaries; (e) any change resulting from the execution of this Agreement or the announcement of the transactions contemplated hereby; or (f) any change resulting from compliance by such Person with the terms of this Agreement or from any action by such Person that is expressly permitted by this Agreement.

MEP Contribution is defined in the recitals to this Agreement.

MEP Contribution Agreement is defined in the recitals to this Agreement.

Operating Subsidiaries means Regency Gas Services LP, a Delaware limited partnership and all other Subsidiaries of Regency.

Organizational Documents means, with respect to any Person, the articles of incorporation, certificate of incorporation, certificate of limited partnership, bylaws, limited liability company agreement, operating agreement, partnership agreement, stockholders agreement and all other similar documents, instruments or certificates executed, adopted or filed in connection with the creation, formation or organization of such Person, including any amendments thereto.

Owned Units are defined in Section 4.5(i).

Party and **Parties** are defined in the preamble of this Agreement.

A-4

Table of Contents

Permits means all permits, approvals, consents, licenses, franchises, exemptions and other authorizations, consents and approvals of or from Governmental Authorities.

Permitted Liens means, with respect to any Person, (a) statutory Liens for current Taxes applicable to the assets of such Person or assessments not yet delinquent or the amount or validity of which is being contested in good faith and for which adequate reserves have been established in accordance with GAAP; (b) mechanics, carriers, workers, repairers, landlords and other similar liens arising or incurred in the ordinary course of business of such Person relating to obligations as to which there is no default on the part of such Person, (c) Liens as may have arisen in the ordinary course of business of such Person, none of which are material to the ownership, use or operation of the assets of such Person; (d) any state of facts which an accurate on the ground survey of any real property of such Person would show, and any easements, rights-of-way, restrictions, restrictive covenants, rights, leases, and other encumbrances on title to real or personal property filed of record that do not materially detract from the value of or materially interfere with the use and operation of any of the assets of such Person; (e) statutory Liens for obligations that are not delinquent, (f) Liens encumbering the fee interest of those tracts of real property encumbered by Rights-of-Way, (g) legal highways, zoning and building laws, ordinances and regulations, that do not materially detract from the value of or materially interfere with the use of the assets of such Person in the ordinary course of business and (h) any Liens with respect to assets of such Person, which, together with all other Liens, do not materially detract from the value of such Person or materially interfere with the present use of the assets owned by such Person or the conduct of the business of such Person.

Person means any natural person, corporation, limited partnership, general partnership, limited liability company, joint stock company, joint venture, association, company, estate, trust, bank trust company, land trust, business trust, or other organization, whether or not a legal entity, custodian, trustee-executor, administrator, nominee or entity in a representative capacity and any Governmental Authority.

Preceding Quarter is defined in Section 2.4(b).

Pre-Closing Tax is defined in Section 5.12(a).

Proceeding means any civil, criminal or administrative actions, suits, investigations or other proceedings.

Redemption and Exchange is defined in the recitals to this Agreement.

Regency is defined in the recitals to this Agreement.

Regency Benefit Plans is defined in Section 3.17(a)(ii).

Regency Common Unit means a common unit representing a limited partner interest in Regency.

Regency Credit Agreement means the Fifth Amended and Restated Credit Agreement dated as of March 4, 2010, by and among Regency, Wachovia Bank, National Association as Administrative Agent and the lenders party thereto, as amended from time.

Regency Entities means Regency, RGPLP, RGPLLC and the Operating Subsidiaries, collectively.

Regency Financial Statements is defined in Section 3.10(c).

A-5

Regency GP Entities means RGPLP and RGPLLC.

Regency GP Interest is defined in Section 3.5(e).

Regency GP LP Interests is defined in Section 3.5(e).

Regency Incentive Distribution Rights means the Incentive Distribution Rights as such term is defined in the Regency Partnership Agreement.

Regency Material Adverse Effect means any Material Adverse Effect in respect of the Regency Entities taken as a whole.

Regency Material Contracts is defined in Section 3.13(b).

Regency Partnership Agreement means that certain Amended and Restated Agreement of Limited Partnership of Regency Energy Partners LP, dated as of February 3, 2006, between RGPLP, as the General Partner, and Seller, as the Organizational Limited Partner, together with any other Persons who become Partners in the Partnership or parties thereto as provided therein, as amended by that Amendment No. 1 to Agreement of Amended and Restated Agreement of Limited Partnership of Regency Energy Partners LP, dated as of August 15, 2006, between RGPLP, as the General Partner, as further amended by that Amendment No. 2 to Amended and Restated Agreement of Limited Partnership of Regency Energy Partners LP, dated as of September 21, 2006, between RGPLP, as the General Partner.

Regency SEC Documents is defined in Section 3.10(a).

Regency Series A Unit is defined in Section 3.5(g).

Registration Rights Agreement is defined in Section 2.3(a)(vii).

Release means any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping, or disposing.

Representatives is defined in Section 5.3.

Responsible Officer means, with respect to any Person, any vice-president or more senior officer of such Person.

RGPLLC is defined in the recitals to this Agreement.

RGPLP is defined in the recitals to this Agreement.

Rights-of-Way means easements, rights-of-way and similar real estate interests.

RIGS Contribution is defined in the preamble to this Agreement.

RIGS HPC is defined in the preamble to this Agreement.

SEC is defined in Section 3.10(a).

Securities Act means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

Seller is defined in the preamble to this Agreement.

Seller Indemnitees is defined in Section 8.2.

A-6

Seller Disclosure Schedule means the disclosure schedule to this Agreement prepared by Seller and delivered to the Buyer Parties on the Execution Date.

Subsidiary means, with respect to any Person, any corporation, limited liability company, partnership, association or other business entity of which a majority of the Voting Interests are at the time owned or Controlled directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof; *provided* that for purposes of this Agreement, RIGS HPC and any of its Subsidiaries shall be deemed to be Subsidiaries of Regency.

Tax means (a) any tax, charge, fee, levy, penalty or other assessment imposed by any United States federal, state, local or foreign taxing authority, including any excise, property, income, sales, transfer, margin, franchise, payroll, withholding, social security or other tax, including any interest, penalties or additions attributable thereto, whether disputed or not; and (b) any liability for the payment of any amounts of the type described in clause (a) as a result of being a member of a consolidated, combined or unitary group for any period; and (c) any liability of for the payment of any amounts of the type described in clause (a) or (b) as a result of the operation of law or any express or implied obligation to indemnify any other Person.

Tax Return means any return, report, information return, declaration, claim for refund or other document (including any related or supporting information or schedules) supplied or required to be supplied to any authority with respect to Taxes and including any supplement or amendment thereof.

Termination Date is defined in Section 7.1(c).

Third Amendment is defined in the recitals to this Agreement.

Transaction Agreements means, collectively, this Agreement, the ETP Redemption Agreement and MEP Contribution Agreement.

Transfer Taxes is defined in Section 5.12(b).

Treasury Regulations means the regulations (including temporary regulations) promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Code. All references herein to sections of the Treasury Regulations shall include any corresponding provision or provisions of succeeding, similar or substitute, temporary or final Treasury Regulations.

Voting Interests of any Person as of any date means the equity interests of such Person pursuant to which the holders thereof have the general voting power under ordinary circumstances to elect at least a majority of the board of directors, managers, general partners or trustees of such Person (regardless of whether, at the time, equity interests of any other class or classes shall have, or might have, voting power by reason of the occurrence of any contingency) or, with respect to a partnership (whether general or limited), any general partner interest in such partnership.

A-7

new roman">

First quarter 2009

15 4%

Second quarter 2009

185 50%

Third quarter 2009

144 40% *

Fourth quarter 2009

174 52% *

* Forecasted

Upon proper qualification, the Company accounts for its fuel derivative instruments as cash flow hedges, as defined in SFAS 133. Under SFAS 133, all derivatives designated as hedges that meet certain requirements are granted special

hedge accounting treatment. Generally, utilizing the special hedge accounting, all periodic changes in fair value of the derivatives designated as hedges that are considered to be effective, as defined, are recorded in "Accumulated other comprehensive income (loss)" ("AOCI") until the underlying jet fuel is consumed. See Note 6 for further information on "AOCI." The Company is exposed to the risk that periodic changes will not be effective, as defined, or that the derivatives will no longer qualify for special hedge accounting. Ineffectiveness, as defined, results when the change in the fair value of the derivative instrument exceeds the change in the value of the Company's expected future cash outlay to purchase and consume jet fuel. To the extent that the periodic changes in the fair value of the derivatives are not effective, that ineffectiveness is recorded to "Other (gains) losses, net" in the statement of operations. Likewise, if a hedge ceases to qualify for hedge accounting, any change in the fair value of derivative instruments since the last period is recorded to "Other (gains) losses, net" in the statement of operations in the period of the change; however, in accordance with SFAS 133, any amounts previously recorded to "AOCI" would remain there until such time as the original forecasted transaction occurs at which time these amounts would be reclassified to "Fuel and oil" expense. In a situation where it becomes probable that a hedged forecasted transaction will not occur, any gains and/or losses that have been recorded to "AOCI" would be required to be immediately reclassified into earnings. The Company did not have any such situations occur for the three or six months ended June 30, 2009 or 2008.

Ineffectiveness is inherent in hedging jet fuel with derivative positions based in other crude oil related commodities. Due to the volatility in markets for crude oil and related products, the Company is unable to predict the amount of ineffectiveness each period, including the loss of hedge accounting, which could be determined on a derivative by derivative basis or in the aggregate for a specific commodity. This may result, and has resulted, in increased volatility in the Company's financial results. Factors that have and may continue to lead to ineffectiveness and unrealized gains and losses on derivative contracts include: significant fluctuation in energy prices, the number of derivative positions the Company holds, significant weather events affecting refinery capacity and the production of refined products, and the volatility of the different types of products the Company uses in hedging. The number of instances in which the Company has discontinued hedge accounting for specific hedges and for specific refined products, such as unleaded gasoline, has increased recently, primarily due to the foregoing factors. However, even though these derivatives may not qualify for SFAS 133 special hedge accounting, the Company continues to hold the instruments as it believes they continue to afford the Company the opportunity to minimize jet fuel costs.

SFAS 133 is a complex accounting standard with stringent requirements, including the documentation of a Company hedging strategy, statistical analysis to qualify a commodity for hedge accounting both on a historical and a prospective basis, and strict contemporaneous documentation that is required at the time each hedge is designated by the Company. As required by SFAS 133, the Company assesses the

Table of Contents

effectiveness of each of its individual hedges on a quarterly basis. The Company also examines the effectiveness of its entire hedging program on a quarterly basis utilizing statistical analysis. This analysis involves utilizing regression and other statistical analyses that compare changes in the price of jet fuel to changes in the prices of the commodities used for hedging purposes.

All cash flows associated with purchasing and selling derivatives are classified as operating cash flows in the unaudited Condensed Consolidated Statement of Cash Flows. The following table presents the location of all assets and liabilities associated with the Company's hedging instruments within the unaudited Condensed Consolidated Balance Sheet (in millions):

		Asset De	erivatives Fair Value	Liability I	Derivatives
	Balance Sheet Location	Fair Value at 6/30/09	at 12/31/08	Fair Value at 6/30/09	Fair Value at 12/31/08
Derivatives designated as hedging instruments under SFAS 133					
Fuel derivative contracts (gross)*	Accrued liabilities	\$92	\$94	\$71	\$19
Fuel derivative contracts (gross)*	Other deferred liabilities	112	40	142	522
Interest rate derivative contracts	Other assets	50	83	-	-
Interest rate derivative contracts	Other deferred liabilities	-	-	-	3
Total derivatives designated as hedging					
instruments under SFAS 133		\$254	\$217	\$213	\$544
Derivatives not designated as hedging					
instruments under SFAS 133					
Fuel derivative contracts (gross)*	Accrued liabilities	\$403	\$387	\$596	\$708
Fuel derivative contracts (gross)*	Other deferred liabilities	308	266	864	530
Total derivatives not designated as					
hedging instruments under SFAS					
133		\$711	\$653	\$1,460	\$1,238
Total derivatives		\$965	\$870	\$1,673	\$1,782

^{*} Does not include the impact of cash collateral deposits provided to counterparties. See discussion of credit risk and collateral following in this Note.

In addition, the Company also had the following amounts associated with fuel derivative instruments and hedging activities in its unaudited Condensed Consolidated Balance Sheet (in millions):

Table of Contents

	Balance Sheet Location	June 30, 2009	December 31, 2008
Cash collateral deposits provided	Offset against Other		
to counterparty - noncurrent	deferred liabilities	374	240
Cash collateral deposits provided	Offset against Accrued		
to counterparty - current	liabilities	51	-
Due to third parties for settled fuel contracts	Accrued liabilities	18	16
Net unrealized losses from fuel	Accumulated other		
hedges, net of tax	comprehensive loss	750	946

The following tables present the impact of derivative instruments and their location within the unaudited Condensed Consolidated Statement of Operations for the three and six months ended June 30, 2009 and 2008 (in millions):

Derivatives in SFAS 133 Cash Flow Hedging Relationships

		Amount of (Gain) Loss					Amount of (Gain) Loss				Amount of (Gain) Loss						
		Recog	gnized	in A	OCI on		Reclassified from AOCI					Recognized in Income					
		Deri	ivative	(Effe	ective			into I	ncome	e (Ef	fective		on Derivatives				
			Porti	ion)					Portio	on)(a)		(ineffective portion) (b)				
							Three months ended June				Three months						
	Th	hree months ended June 30,				30,					ended June 30,						
	20	09		20	800		20	09		20	80		20	09		20	80
Fuel derivative																	
contracts	\$	(42) *	\$	(1,493) *	\$	96	*	\$	(284) *	\$	(25)	\$	20
Interest rate																	
derivatives		(20) *		-			-			-			-			-
Total	\$	(62)	\$	(1,493)	\$	96		\$	(284)	\$	(25)	\$	20

^{*} Net of tax

⁽a) Amounts related to fuel derivative contracts and interest rate derivatives are included in Fuel and oil and Interest expense, respectively.

⁽b) Amounts are included in Other (gains) losses, net.

Table of Contents

Derivatives in SFAS 133 Cash Flow Hedging Relationships

							Amount of (Gain) Loss					Amount of (Gain) Loss					
		Amo	unt of (Gair	n) Loss		Reclassified from AOCI				Recognized in Income on						
		Reco	gnized	in A	OCI on		into Income (Effective				Derivatives (ineffective				fective		
	D	erivati	ve (Eff	ectiv	e Portion	1)			Portio	n)(a))		portion) (b)				
													Six months ended June				
	,	Six mo	onths er	nded	June 30,		S	ix mon	ths en	ded	June 30),			30),	
	20	009		20	800		20	09		20	08		20	009		20	80
Fuel derivative																	
contracts	\$	10	*	\$	(1,923) *	\$	206	*	\$	(454) *	\$	(9)	\$	26
Interest rate																	
derivatives		(25) *		-			-			-			-			-
			•														
Total	\$	(15)	\$	(1,923)	\$	206		\$	(454)	\$	(9)	\$	26
			,		` '	,					•	,					

^{*} Net of tax

Derivatives not in SFAS 133 Cash Flow Hedging Relationships Amount of (Gain) Loss Recognized in Income on Derivatives

Location of (Gain) Loss Recognized Three months ended June 30, in Income on 2009 2008 Derivatives Fuel derivative Other (gains) 19 losses, net contracts (381)

Derivatives not in SFAS 133 Cash Flow Hedging Relationships Amount of (Gain) Loss Recognized in Income on Derivatives

Location of (Gain) Loss
Recognized in
Six months ended June 30,
Income
on Derivatives

\$ (8) \$ (365)

⁽a) Amounts related to fuel derivative contracts and interest rate derivatives are included in Fuel and oil and Interest expense, respectively.

⁽b) Amounts are included in Other (gains) losses, net.

Fuel derivative	Other (gains)
contracts	losses, net

The Company also recorded expense associated with premiums paid for fuel derivative contracts that settled/expired during the three months ended June 30, 2009 and 2008, respectively, of \$37 million and \$14 million, and during the six months ended June 30, 2009 and 2008, respectively, of \$69 million and \$27 million. These amounts are excluded from the Company's measurement of effectiveness for related hedges.

The fair value of the derivative instruments, depending on the type of instrument, was determined by the use of present value methods or standard option value models with assumptions about commodity prices based on those observed in underlying markets. Included in the Company's total net unrealized losses from fuel hedges as of June 30, 2009, are approximately \$280 million in unrealized losses, net of taxes, that are expected to be realized in earnings during the twelve months following June 30, 2009. In addition, as of June 30, 2009, the Company had already recognized cumulative net losses due to ineffectiveness and derivatives that do not qualify for hedge accounting totaling \$43 million, net of taxes. These losses were recognized in the three months ended June 30, 2009, and prior periods, and are reflected in "Retained earnings" as of June 30, 2009, but the underlying derivative instruments will not expire/settle until subsequent periods of 2009 or future periods.

Interest rate swaps

The Company is party to interest rate swap agreements related to its \$385 million 6.5% senior unsecured notes due 2012, its \$350 million 5.25% senior unsecured notes due 2014, its \$300 million 5.125% senior unsecured notes due 2017, and its \$100 million 7.375% senior unsecured debentures due 2027. The primary objective for the Company's use of these interest rate hedges is to better match the repricing of its assets and liabilities. Under each of these interest rate swap agreements, the Company pays the London InterBank Offered Rate (LIBOR) plus a margin every six months on the notional amount of the debt, and receives payments based on the fixed stated rate of the notes every six months until the date the notes become due. These interest rate swap agreements qualify as fair value hedges, as defined by SFAS 133. In addition, these interest rate swap agreements qualify for the "shortcut" method of accounting for hedges, as defined by SFAS 133. Under the "shortcut" method, the hedges are assumed to be perfectly effective, and, thus, there is no ineffectiveness to be recorded in earnings.

The Company also entered into interest rate swap agreements concurrent with its entry into a twelve-year, \$600 million floating-rate term loan agreement during 2008, and a ten-year, \$332 million floating-rate term loan agreement during May 2009. Under these swap agreements, which are accounted for as cash flow hedges, the interest rates on the term loans are effectively fixed for their entire term at 5.223 percent and 6.64 percent, respectively, and ineffectiveness is required to be measured each reporting period. The fair values of the interest rate swap agreements, which are adjusted regularly, have been aggregated by counterparty for classification in the unaudited Condensed Consolidated Balance Sheet.

Credit risk and collateral

The Company's credit exposure related to fuel derivative instruments is represented by the fair value of contracts with a net positive fair value to the Company at the reporting date. These outstanding instruments expose the Company to credit loss in the event of nonperformance by the counterparties to the agreements. However, the Company has not experienced any significant credit loss as a result of counterparty nonperformance in the past. To manage credit risk, the Company selects and will periodically review counterparties based on credit ratings, limits its exposure to a single counterparty, and monitors the market position of the fuel hedging program and its relative market position with each counterparty. At June 30, 2009, the Company had agreements with all of its counterparties containing early termination rights and/or bilateral collateral provisions whereby security is required if market risk exposure exceeds a specified threshold amount or credit ratings fall below certain levels. Based on the Company's current agreements with two of these counterparties, cash deposits are required to be posted whenever the net fair value of derivatives associated with those counterparties exceed specific thresholds. If the threshold is exceeded, cash is either posted by the counterparty if the value of derivatives is an asset to the Company, or posted by the Company if the value of derivatives is a liability to the Company.

Under one of the Company's counterparty agreements, as amended, until January 1, 2010, if the Company becomes obligated to post collateral for obligations in amounts of up to \$300 million and in excess of \$700 million, the Company is required to post cash collateral; however, if the Company becomes obligated to post collateral for obligations in amounts between \$300 million and \$700 million, the Company has pledged 20 of its Boeing 737-700 aircraft as collateral in lieu of cash. At June 30, 2009, the fair value of fuel derivative instruments with this counterparty was a net liability of \$302 million, and the Company had posted \$300 million in cash collateral deposits with the counterparty, with the remaining \$2 million secured by pledged aircraft. If the fair value of fuel derivative instruments with this counterparty were in a net asset position, the counterparty would be required to post cash collateral to the Company on a dollar-for-dollar basis for amounts in excess of \$40 million. This agreement does not contain any triggers that would require additional cash to be posted by the Company outside of further changes in the

fair value of the fuel derivative instruments held with the counterparty. However, if the fair value of fuel derivative

Table of Contents

instruments with this counterparty were in a net asset position, and the counterparty's credit rating were to be lowered to specified levels, the counterparty could be required to post cash collateral to the Company on a dollar-for-dollar basis related to the first \$40 million of assets held.

Under another of the Company's counterparty agreements, the Company is obligated to post collateral related to fuel derivative liabilities as follows: (i) if the obligation is up to \$125 million, the Company posts cash collateral, (ii) if the obligation is between \$125 million and \$625 million, the Company has pledged 29 Boeing 737-700 aircraft as collateral in lieu of cash, and (iii) if the obligation exceeds \$625 million, the Company must post cash or letters of credit as collateral. The Company pledged 29 of its Boeing 737-700 aircraft to cover the collateral posting band in clause (ii). As of June 30, 2009, the fair value of fuel derivative instruments with this counterparty was a net liability of \$399 million, and the Company had posted \$125 million in cash collateral deposits to this counterparty, with the remaining \$274 million secured by pledged aircraft. This agreement also provides for the counterparty to post cash collateral to the Company on a dollar-for-dollar basis for any net positive fair value of fuel derivative instruments in excess of \$150 million held by the Company from that counterparty. This agreement does not contain any triggers that would require additional cash to be posted by the Company outside of further changes in the fair value of the fuel derivative instruments with this counterparty were in a net asset position, and the counterparty's credit rating were to be lowered to specified levels, the counterparty would be required to post cash collateral to the Company on a dollar-for-dollar basis related to the first \$150 million of assets held.

As of June 30, 2009, other than as described above, the Company did not have any fuel hedging agreements with counterparties in which cash collateral is required to be posted based on the Company's current investment grade credit rating. However, additional fuel hedging agreements contain a provision whereby each party has the right to terminate and settle all outstanding fuel contracts if the other party's credit rating falls below investment grade. Upon this occurrence, the party in a net liability position could subsequently be required to post cash collateral if a mutual alternative agreement could not be reached. At June 30, 2009, the Company's estimated fair value of fuel derivative contracts with one counterparty, containing this provision, was a liability of \$101 million, including \$29 million that will settle by the end of 2009.

Due to the Company's current fuel hedging agreements with counterparties, in the Company's judgment, it does not have significant exposure to future cash collateral requirements. As an example, even if market prices for the commodities used in the Company's fuel hedging activities were to decrease by 50 percent from market prices as of June 30, 2009, given the Company's current fuel hedge portfolio and its investment grade credit rating, it would not have had to provide additional cash collateral to its current counterparties.

The Company classifies its cash collateral provided to counterparties in accordance with the provisions of FIN 39-1. FIN 39-1 requires an entity to select a policy of how it records the offset rights to reclaim cash collateral associated with the related derivative fair value of the assets or liabilities of such derivative instruments. Entities may either select a "net" or a "gross" presentation. The Company has elected to present its cash collateral utilizing a net presentation, in which cash collateral amounts held or provided have been netted against the fair value of outstanding derivative instruments. The Company's policy differs depending on whether its derivative instruments are in a net asset position or a net liability position. If its fuel derivative instruments are in a net asset position with a counterparty, cash collateral amounts held are first netted against current derivative amounts (those that will settle during the twelve months following the balance sheet date) associated with that counterparty until that balance is zero, and then any remainder would be applied against the fair value of noncurrent outstanding derivative instruments (those that will settle beyond one year following the balance sheet date.) If its fuel derivative instruments are in a net liability position with a counterparty, cash collateral amounts provided are first netted against noncurrent derivative amounts associated with that counterparty until that balance is zero, and then any remainder would be applied against

the fair value of current outstanding derivative instruments. At June 30, 2009, of the \$425 million in cash collateral deposits posted with counterparties under its bilateral collateral provisions, \$374 million has been netted against noncurrent fuel derivative instruments within "Other deferred liabilities" and \$51 million has been netted against current fuel derivative instruments within "Accrued liabilities" in the unaudited Condensed Consolidated Balance Sheet.

Comprehensive income

6. COMPREHENSIVE INCOME (LOSS)

Comprehensive income (loss) includes changes in the fair value of certain financial derivative instruments, which qualify for hedge accounting, and unrealized gains and losses on certain investments. The differences between net income (loss) and comprehensive income (loss) for the three and six months ended June 30, 2009 and 2008, were as follows:

(In millions)	Tl 200		onths e	ended 200	June 30 08	,
Net income	\$	54		\$	321	
Unrealized gain (loss) on derivative instruments,						
net of deferred taxes of \$86 and \$753		138			1,209	
Other, net of deferred taxes of \$14 and (\$1)		22			(2)
Total other comprehensive income		160			1,207	
Comprehensive income	\$	214		\$	1,528	
(In millions)	200		hs end	200	08	
Net income	\$	(37)	\$	355	
Unrealized gain (loss) on derivative instruments,						
net of deferred taxes of \$122 and \$904		196			1,469	
Other, net of deferred taxes of \$16 and (\$7)		26			(11)
Total other comprehensive income (loss)		222			1,458	

A rollforward of the amounts included in "AOCI," net of taxes, is shown below for the three and six months ended June 30, 2009:

1,813

185

		Accumulated
	Fuel	other
	hedge	comprehensive
(In millions)	derivatives Other	income (loss)
Balance at March 31, 2009	\$(888) \$(34) \$ (922)
Second quarter 2009 changes in value	42 22	64
Reclassification to earnings	96 -	96
Balance at June 30, 2009	\$(750) \$(12) \$ (762)

		Fuel ledge			ccumulat other nprehens	
(In millions)		ivatives	Other		come (los	
Balance at December 31, 2008		(946) \$	(38) \$	(984)
2009 changes in value	·	(10)	26	, .	16	
Reclassification to earnings		206	-		206	
Balance at June 30, 2009	\$	(750) \$	(12) \$	(762)

7. ACCRUED LIABILITIES

(In millions)	J 200	une 30,	D 200	ecember 31, 08
Retirement plans	\$	97	\$	86
Aircraft rentals		109		118
Vacation pay		181		175
Advances and deposits		14		23
Fuel derivative contracts		120		246
Deferred income taxes		138		36
Workers compensation		124		122
Other		246		206
Accrued liabilities	\$	1,029	\$	1,012

8. POSTRETIREMENT BENEFITS

The following table sets forth the Company's periodic postretirement benefit cost for each of the interim periods identified:

(In millions)	Ti 200		nonths er	nded 200		30,
Service cost	\$	4		\$	4	
Interest cost		1			1	
Amortization of prior service cost		1			1	
Recognized actuarial gain		(1)		(1)
Net periodic postretirement benefit cost	\$	5		\$	5	

	Si	x moi	nths ende	ed Ju	ne 30,	
(In millions)	200)9		200	98	
Service cost	\$	7		\$	7	
Interest cost		2			2	
Amortization of prior service cost		1			1	
Recognized actuarial gain		(1)		(1)
Net periodic postretirement benefit cost	\$	9		\$	9	

9. FINANCING TRANSACTIONS

On April 29, 2009, the Company entered into a term loan agreement providing for loans to the Company aggregating up to \$332 million, to be secured by mortgages on 14 of the Company's 737-700 aircraft. On May 6, 2009, the Company borrowed the full \$332 million and secured the loan with the requisite 14 aircraft mortgages. The loan matures on May 6, 2019, and is repayable quarterly in installments of principal beginning August 6, 2009. The loan bears interest at the LIBO Rate (as defined in the term loan agreement) plus 3.30 percent, and interest is payable quarterly, beginning August 6, 2009. Concurrent with its entry into the term loan agreement, the Company entered into an interest rate swap agreement that effectively fixes the interest rate on the term loan for its entire term at 6.315 percent. The Company used the proceeds from the term loan for general corporate purposes, including the repayment of the Company's revolving credit facility.

During May 2009, the Company fully repaid the \$400 million it had previously borrowed in 2008 under its available \$600 million revolving credit facility. Therefore, as of June 30, 2009, the entire \$600 million of the Company's revolving credit facility was available for borrowing.

10. COMMITMENTS AND CONTINGENCIES

During the first quarter and early second quarter of 2008, the Company was named as a defendant in two putative class actions on behalf of persons who purchased air travel from the Company while the Company was allegedly in violation of FAA safety regulations. Claims alleged by the plaintiffs in these two putative class actions include breach of contract, breach of warranty, fraud/misrepresentation, unjust enrichment, and negligent and reckless operation of an aircraft. The Company believes that the class action lawsuits are without merit and intends to vigorously defend itself. Also in connection with this incident, during the first quarter and early second quarter of 2008, the Company received four letters from Shareholders demanding the Company commence an action on behalf of the Company against members of its Board of Directors and any other allegedly culpable parties for damages resulting from an alleged breach of fiduciary duties owed by them to the Company. In August 2008, Carbon County Employees Retirement System and Mark Cristello filed a related Shareholder derivative action in Texas state court naming certain directors and officers of the Company as individual defendants and the Company as a nominal defendant. The derivative action claims breach of fiduciary duty and seeks recovery by the Company of alleged monetary damages sustained as a result of the purported breach of fiduciary duty, as well as costs of the action. A Special Committee appointed by the Independent Directors of the Company has been evaluating the Shareholder demands.

The Company is from time to time subject to various other legal proceedings and claims arising in the ordinary course of business, including, but not limited to, examinations by the Internal Revenue Service (IRS).

The Company's management does not expect that the outcome in any of its currently ongoing legal proceedings or the outcome of any proposed adjustments presented to date by the IRS, individually or collectively, will have a material adverse effect on the Company's financial condition, results of operations, or cash flow.

During 2008, the City of Dallas approved the Love Field Modernization Program, an estimated \$519 million project to reconstruct Dallas Love Field with modern, convenient air travel facilities. Pursuant to a Program Development Agreement with the City of Dallas, the Company is managing this project, and initial construction is expected to commence during late 2009, with completion scheduled for October 2014. Bonds are expected to be issued at a later date by the Love Field Airport Modernization Corporation (a "local government corporation" under Texas law formed by the City of Dallas) that will provide funding for this project, with repayment of the bonds being made through recurring ground rents, fees, and other revenues collected by the airport.

11. FAIR VALUE MEASUREMENTS

The Company adopted SFAS 157 as of January 1, 2008. SFAS 157 establishes a three-tier fair value hierarchy, which prioritizes the inputs used in measuring fair value. These tiers include: Level 1, defined as observable inputs such as quoted prices in active markets; Level 2, defined as inputs other than quoted prices in active markets that are either directly or indirectly observable; and Level 3, defined as unobservable inputs in which little or no market data exists, therefore requiring an entity to develop its own assumptions.

As of June 30, 2009, the Company held certain items that are required to be measured at fair value on a recurring basis. These included cash equivalents, short-term investments, certain noncurrent investments, interest rate derivative contracts, fuel derivative contracts, and available-for-sale securities. Cash equivalents consist of short-term, highly liquid, income-producing investments, all of which have maturities of 90 days or less, including money market funds, U.S. Government obligations, and obligations of U.S. Government backed agencies. Short-term investments consist of short-term, highly liquid, income-producing investments, which have maturities of greater than 90 days but less than one year, including U.S. Government obligations, obligations of U.S. Government backed agencies, and certain auction rate securities. Derivative instruments are related to the Company's fuel hedging program and interest rate hedges. Noncurrent investments consist of certain auction rate securities, primarily those collateralized by student loan portfolios, which are guaranteed by the U.S. Government. Other available-for-sale securities primarily consist of investments associated with the Company's Excess Benefit Plan.

The Company's fuel derivative instruments consist of over-the-counter (OTC) contracts, which are not traded on a public exchange. These contracts include both swaps as well as different types of option contracts. See Note 5 for further information on the Company's derivative instruments and hedging activities. The fair values of swap contracts are determined based on inputs that are readily available in public markets or can be derived from information available in publicly quoted markets. Therefore, the Company has categorized these swap contracts as Level 2. The Company determines the value of option contracts utilizing a standard option pricing model based on inputs that are either readily available in public markets, can be derived from information available in publicly quoted markets, or are quoted by financial institutions that trade these contracts. In situations where the Company obtains inputs via quotes from financial institutions, it verifies the reasonableness of these quotes via similar quotes from another financial institution as of each date for which financial statements are prepared. The Company also considers counterparty credit risk and its own credit risk in its determination of all estimated fair values. The Company has consistently applied these valuation techniques in all periods presented and believes it has obtained the most accurate information available for the types of derivative contracts it holds. Due to the fact that certain of the inputs used to determine the fair value of option contracts are unobservable (principally implied volatility), the Company has categorized these option contracts as Level 3.

The Company's interest rate derivative instruments also consist of OTC swap contracts. The inputs used to determine the fair values of these contracts are obtained in quoted public markets. The Company has consistently applied these valuation techniques in all periods presented.

The Company's investments associated with its Excess Benefit Plan consist of mutual funds that are publicly traded and for which market prices are readily available.

All of the Company's auction rate security instruments are reflected at estimated fair value in the unaudited Condensed Consolidated Balance Sheet. At June 30, 2009, approximately \$109 million of these instruments are classified as available for sale securities and \$83 million are classified as trading securities. The \$83 million classified as trading securities are subject to an agreement the Company entered into in December 2008, as discussed below, and are included in "Short-term investments" in the unaudited Condensed Consolidated Balance Sheet. In periods when an

auction process successfully takes place every 30-35 days, quoted market prices would be readily available, which would qualify as Level 1 under SFAS

157. However, due to events in credit markets beginning during first quarter 2008, the auction events for most of these instruments failed, and, therefore, the Company has subsequently determined the estimated fair values of these securities utilizing a discounted cash flow analysis or other type of valuation model. In addition, during fourth quarter 2008, the Company performed a valuation of its auction rate security instruments and considered these valuations in determining estimated fair values of other similar instruments within its portfolio. The Company's analyses consider, among other items, the collateralization underlying the security investments, the expected future cash flows, including the final maturity, associated with the securities, and estimates of the next time the security is expected to have a successful auction or return to full par value. These securities were also compared, when possible, to other securities not owned by the Company, but with similar characteristics.

In association with this estimate of fair value, the Company has recorded a temporary unrealized decline in fair value of \$11 million, with an offsetting entry to "AOCI." The Company currently believes that this temporary decline in fair value is due entirely to liquidity issues, because the underlying assets for the majority of these auction rate securities held by the Company are almost entirely backed by the U.S. Government. In addition, for the \$109 million in instruments classified as available for sale, these auction rate securities represented approximately five percent of the Company's total cash, cash equivalent, and investment balance at June 30, 2009. Considering the relative significance of these securities in comparison to the Company's liquid assets and other sources of liquidity, the Company has no current intention of selling these securities nor does it expect to be required to sell these securities before a recovery in their cost basis. For the \$83 million in instruments classified as trading securities, the Company is party to an agreement with the counterparty that allows the Company to put the instruments back to the counterparty at full par value in June 2010. In conjunction with this agreement, the Company has applied the provisions of SFAS 159, "The Fair Value Option for Financial Assets and Financial Liabilities" to this put option. Part of this agreement also contains a line of credit in which the Company can borrow up to \$83 million as a loan from the counterparty that would be secured by the auction rate security instruments from that counterparty. There were no borrowings under this provision as of June 30, 2009. Both the put option and the auction rate instruments are being marked to market through earnings each period; however, these adjustments offset and had minimal impact on net earnings for the three and six months ended June 30, 2009. At the time of the first failed auctions during first quarter 2008, the Company held a total of \$463 million in securities. Since that time, the Company has been able to sell \$260 million of these instruments at par value in addition to the \$83 million subject to the agreement to be settled at par in June 2010.

During first quarter 2009, the Company also entered into a \$46 million line of credit agreement with another counterparty secured by approximately \$92 million (par value) of its remaining auction rate security instruments purchased through that counterparty. This agreement allows the Company the ability to draw against the line of credit secured by the auction rate security instruments from that counterparty. As of June 30, 2009, the Company had no borrowings against that available line of credit. The Company remains in discussions with its other counterparties to determine whether mutually agreeable decisions can be reached regarding the effective repurchase of its remaining securities. The Company has continued to earn interest on virtually all of its auction rate security instruments. Any future fluctuation in fair value related to these instruments that the Company deems to be temporary, including any recoveries of previous temporary write-downs, would be recorded to "AOCI." If the Company determines that any future valuation adjustment was other than temporary, it would record a charge to earnings as appropriate.

The following items are measured at fair value on a recurring basis subject to the disclosure requirements of SFAS 157 at June 30, 2009:

Fair Value Measurements at Reporting Date Using Quoted Prices

Significant Unobservable

Table of Contents 97

in

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A

				Active Iarkets for Identical		gnifican Other bservabl				
				Assets		Inputs			Inputs	
Description	Jun	e 30, 2009		(Level 1)	(Level 2)		(Level 3))
Assets				(in mi	llions)					
Cash equivalents	\$	946	\$	946	\$	-		\$	-	
Short-term investments		1,252		1,222		-			103	
Noncurrent investments (a)		89		-		-			89	
Interest rate derivatives		50		_		50			-	
Fuel derivatives (b)		915		-		338			577	
Other available-for-sale										
securities		33		25		-			8	
Total assets	\$	3,285	\$	2,193	\$	388		\$	777	
Liabilities										
Fuel derivatives (b)	\$	(1,678))		\$	(964)	\$	(714)
() T	41	1', 10 1	1.0	"1 4 1D 1	C1 4					

⁽a) Included in "Other assets" in the unaudited Condensed Consolidated Balance Sheet.

The following table presents the Company's activity for assets measured at fair value on a recurring basis using significant unobservable inputs (Level 3) as defined in SFAS 157 for the three months ended June 30, 2009:

⁽b) In the unaudited Condensed Consolidated Balance Sheet, amounts are presented as a net liability, and are also net of \$425 million in cash collateral provided to counterparties.

Fair Value Measurements Using Significant Unobservable Inputs (Level 3)

Fuel Rate Other Securities (in millions) Derivatives (a) Securities Total Balance at December 31, 2008 \$ (864) \$ 200 \$ 8 \$ (656) Total gains or (losses) (realized or unrealized) Included in earnings 473 473 Included in other comprehensive income (23) (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the chapter in unrealized raise or					A	uction					
(in millions) Balance at December 31, 2008 \$ (864) \$ 200 \$ 8 \$ (656) Total gains or (losses) (realized or unrealized) Included in earnings Included in other comprehensive income (23) (23) Purchases and settlements (net) Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the			Fuel			Rate		(Other		
Balance at December 31, 2008 \$ (864) \$ 200 \$ 8 \$ (656) Total gains or (losses) (realized or unrealized) Included in earnings 473 - 473 Included in other comprehensive income (23) - (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the					Se	curities					
Total gains or (losses) (realized or unrealized) Included in earnings 473 - 473 Included in other comprehensive income (23) - (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	(in millions)	Г	erivatives			(a)		Sec	curities	Total	
unrealized) Included in earnings 473 - 473 Included in other comprehensive income (23) - (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	Balance at December 31, 2008	\$	(864)	\$	200		\$	8	\$ (656)
Included in earnings Included in other comprehensive income (23) - (23) Purchases and settlements (net) Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	Total gains or (losses) (realized or										
Included in other comprehensive income (23) (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	unrealized)										
income (23) (23) Purchases and settlements (net) 277 (8) - 269 Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	Included in earnings		473			-			-	473	
Purchases and settlements (net) Balance at June 30, 2009 \$ (137) \$ 192 \$ (8) - 269 Balance at June 30, 2009 \$ (137) The amount of total gains or (losses) for the period included in earnings attributable to the	Included in other comprehensive										
Balance at June 30, 2009 \$ (137) \$ 192 (b) \$ 8 \$ 63 The amount of total gains or (losses) for the period included in earnings attributable to the	income		(23)		-			-	(23)
The amount of total gains or (losses) for the period included in earnings attributable to the	Purchases and settlements (net)		277			(8)		-	269	
(losses) for the period included in earnings attributable to the	Balance at June 30, 2009	\$	(137)	\$	192	(b)	\$	8	\$ 63	
(losses) for the period included in earnings attributable to the											
period included in earnings attributable to the	The amount of total gains or										
attributable to the	(losses) for the										
	period included in earnings										
change in unrealized gains or	attributable to the										
change in unicanzed gains of	change in unrealized gains or										
losses relating to	losses relating to										
assets still held at June 30, 2009 \$ 466 \$ - \$ 466	assets still held at June 30, 2009	\$	466		\$	-		\$	-	\$ 466	

- (a) Includes those classified as short-term investments and noncurrent investments.
- (b) Includes \$83 million classified as trading securities.

All settlements from fuel derivative contracts that are deemed "effective," as defined in SFAS 133, are included in "Fuel and oil" expense in the period the underlying fuel is consumed in operations. Any "ineffectiveness" associated with derivative contracts, as defined in SFAS 133, including amounts that settled in the current period (realized), and amounts that will settle in future periods (unrealized), is recorded in earnings immediately, as a component of "Other (gains) losses, net." See Note 5 for further information on SFAS 133 and hedging.

Gains and losses (realized and unrealized) included in earnings related to other investments for the three and six months ended June 30, 2009, are reported in "Other operating expenses."

The carrying amounts and estimated fair values of the Company's long-term debt and fuel derivative contracts at June 30, 2009 were as follows:

(In millions)	C	Carrying value		imated r value
		value	Tan	varue
10.5% Notes due				
2011	\$	400	\$	432
French Credit				
Agreements due				
2012		24		24
6.5% Notes due 2012		400		409

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A

5.25% Notes due		
2014	373	347
5.75% Notes due		
2016	300	270
5.125% Notes due		
2017	333	288
French Credit		
Agreements due		
2017	84	84
Term Loan		
Agreement due 2019	332	336
Term Loan		
Agreement due 2020	570	481
Pass Through		
Certificates	458	473
7.375% Debentures		
due 2027	114	107
Fuel derivative		
contracts*	(763)	(763)

^{*} Does not include the impact of cash collateral deposits provided to counterparties. See Note 5.

The estimated fair values of the Company's publicly held long-term debt were based on quoted market prices.

Table of Contents

12. TAX RATE

The Company's effective tax rate was a negative 7.3 percent in second quarter 2009. This low rate was due to a \$21 million adjustment (\$.03 per share, diluted) necessary to achieve a revised year-to-date tax rate of 35.0 percent based on the Company's currently forecasted results for the full year 2009.

13. EARLY RETIREMENT OFFER

On April 16, 2009, the Company announced Freedom '09, a one-time voluntary early out program offered to eligible Employees, in which the Company offered cash bonuses, medical/dental coverage for a specified period of time, and travel privileges based on work group and years of service. The purpose of this voluntary initiative and other initiatives is to right-size headcount in conjunction with the Company's current plans to reduce its capacity by five percent in 2009, and to help reduce costs. Virtually all of the Company's Employees hired before March 31, 2008 were eligible to participate in the program.

Participants' last day of work will fall between July 31, 2009 and April 15, 2010, as assigned by the Company based on the operational needs of particular work locations and departments, determined on an individual-by-individual basis. The Company did not have a target or expectation for the number of Employees expected to accept the package.

Employees electing to participate in Freedom '09 were required to notify the Company of their election by June 19, 2009. However, Employees had until July 16, 2009 to rescind their election and remain with the Company. Following the deadline to rescind such election, a total of 1,404 Employees have remained as participants in Freedom '09, consisting of the following breakdown among workgroups: 439 from Customer Support and Services, 464 from Ground Operations and Provisioning, 113 Flight Attendants, 20 Pilots, 91 from Maintenance, and 277 Managerial and Administrative Employees. The Company expects the total cost incurred for Freedom '09 to be approximately \$70 million. Due to the Company's mandatory service period requirement for Employees electing Freedom '09, the Company is required to spread the cost of the program over the period worked, once the period for revocation has passed. The Company may need to replace a small number of the positions with newly hired Employees to meet operational demands; however, the Company expects that most of the positions will not be filled based on the Company's recent capacity reductions.

14. SALE AND LEASEBACK TRANSACTIONS

On April 2, 2009, the Company closed the first tranche of a two tranche sale and leaseback transaction with a third party aircraft lessor for the sale and leaseback of six of the Company's Boeing 737-700 aircraft. In the first tranche, the Company sold three of its Boeing 737-700 aircraft for approximately \$105 million and immediately leased the aircraft back for approximately 12 years. On May 19, 2009, the Company sold an additional three of its Boeing 737-700 aircraft for approximately the same amount and upon similar terms as in the first tranche. These two sale and leaseback transactions resulted in deferred gains of \$20 million, which will be amortized over the terms of the leases.

All of the leases from these sale-leasebacks are accounted for as operating leases. Under the terms of the lease agreements, the Company will continue to operate and maintain the aircraft. Payments under the lease agreements will be reset every six months based on changes in the six-month LIBO rate. The lease agreements contain standard termination events, including termination upon a breach of the Company's obligations to make rental payments and upon any other material breach of the Company's obligations under the leases, and standard maintenance and return condition provisions. Upon a termination of the lease upon a breach by the Company, the Company would be liable for standard contractual damages, possibly including damages suffered by the lessor in connection with remarketing the aircraft or while the aircraft is not leased to another party.

15. SUBSEQUENT EVENTS

On July 1, 2009, the Company entered into a term loan agreement providing for loans to the Company aggregating up to \$124 million, to be secured by mortgages on five of the Company's 737-700 aircraft. Subsequently, the Company borrowed the full \$124 million and secured this loan with the requisite five aircraft mortgages. The loan matures on July 1, 2019, and is repayable semi-annually in installments of principal beginning January 1, 2010. The loan bears interest at a fixed rate of 6.84 percent, and interest is payable semi-annually, beginning January 1, 2010. The Company used the proceeds from the term loan for general corporate purposes.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Comparative Consolidated Operating Statistics

Relevant Southwest comparative operating statistics for the three and six months ended June 30, 2009 and 2008 are as follows:

Inree	months	enaea	June 30,

	2009	2008	Change
Revenue passengers carried	22,676,171	23,993,342	(5.5)%
Enplaned passengers	26,505,438	27,550,957	(3.8)%
Revenue passenger miles (RPMs)			
(000s)	19,683,479	19,811,541	(0.6)%
Available seat miles (ASMs) (000s)	25,552,927	26,335,085	(3.0)%
Load factor	77.0 %	75.2 %	1.8 pts
Average length of passenger haul			
(miles)	868	826	5.1 %
Average aircraft stage length (miles)	647	636	1.7 %
Trips flown	289,573	303,432	(4.6)%
Average passenger fare	\$110.52	\$114.48	(3.5)%
Passenger revenue yield per RPM			
(cents)	12.73	13.86	(8.2)%
Operating revenue yield per ASM			
(cents)	10.24	10.89	(6.0)%
Operating expenses per ASM (cents)	9.76	10.12	(3.6)%
Fuel costs per gallon, including fuel			
tax	\$1.95	\$2.42	(19.4)%
Fuel consumed, in gallons (millions)	371	388	(4.4)%
Full-time equivalent Employees at			
period-end*	35,296	34,936	1.0 %
Aircraft in service at period-end**	543	535	1.5 %

^{*} Headcount is defined as "Active" fulltime equivalent Employees for both periods presented.

Six months ended June 30,

	2009	2008	Change
Revenue passengers carried	42,435,861	45,498,163	(6.7)%
Enplaned passengers	49,555,428	52,259,572	(5.2)%
Revenue passenger miles (RPMs)			
(000s)	36,575,108	37,403,700	(2.2)%
Available seat miles (ASMs) (000s)	49,724,602	51,528,522	(3.5)%
Load factor	73.6 %	72.6 %	1.0 pts
Average length of passenger haul			
(miles)	862	822	4.9 %
Average aircraft stage length (miles)	641	632	1.4 %
Trips flown	568,708	598,222	(4.9)%
Average passenger fare	\$112.13	\$113.42	(1.1)%

^{**} Excludes any aircraft that have been removed from service and are held for sale or for return to the lessor.

Edgar Filing: Regency Energy Partners LP - Form SC 13D/A

Passenger revenue yield per RPM				
(cents)	13.01	13.80	(5.7)%
Operating revenue yield per ASM				
(cents)	10.00	10.48	(4.6)%
Operating expenses per ASM (cents)	9.85	9.91	(0.6))%
Fuel costs per gallon, including fuel tax	\$1.97	\$2.28	(13.6)%
Fuel consumed, in gallons (millions)	721	761	(5.3)%
Full-time equivalent Employees at				
period-end*	35,296	34,936	1.0	%
Aircraft in service at period-end**	543	535	1.5	%

^{*} Headcount is defined as "Active" fulltime equivalent Employees for both periods presented.

^{**} Excludes any aircraft that have been removed from service and are held for sale or for return to the lessor.

Table of Contents

Material Changes in Results of Operations

Summary

During second quarter 2009, Southwest recorded net income of \$54 million, or \$.07 income per share, diluted, versus the Company's second quarter 2008 net income of \$321 million, or \$.44 per share, diluted. More than 75 percent of the reduction in net income versus the same prior year period was due to a difference in recorded adjustments related to derivative contracts the Company utilizes in attempting to hedge against jet fuel price increases. In second quarter 2008, the Company recorded significant unrealized gains from marking to market derivatives used for hedging purposes, but that do not qualify for special hedge accounting, as defined in SFAS 133. See Note 5 to the unaudited condensed consolidated financial statements for further information on the Company's hedging activities and accounting associated with derivative instruments. Primarily as a result of second quarter 2008 increases in prices for fuel derivatives that have or will subsequently settle after June 30, 2008, that were ineffective, as defined, or that did not qualify for special hedge accounting, the Company recorded \$361 million in net gains, which are included in "Other (gains) losses, net." In second quarter 2009, the Company recorded a total of only \$6 million in net gains associated with fuel derivatives that were ineffective, as defined, or that did not qualify for special hedge accounting, in "Other (gains) losses, net." The majority of the remainder of the decrease in net income versus second quarter 2008 was due to a decline in demand for domestic air travel as a result of the current U.S. and global recession. This decline in demand resulted in fewer full-fare passengers and more fare discounting, which depressed yields, and occurred despite the Company, as well as most other airlines, reducing capacity versus the prior year. The Company experienced a 5.9 percent decrease in passenger revenue yield per available seat mile (ASM) versus second quarter 2008.

The Company's operating expenses declined 6.4 percent versus second quarter 2008, the majority of which was attributable to lower fuel prices. Overall, for second quarter 2009, the Company's average jet fuel cost per gallon (including related fuel taxes) decreased 19.4 percent compared to second quarter 2008, inclusive of gains and/or losses from fuel contract settlements and related SFAS 133 adjustments included in "Fuel and oil" expense. Cash settlements associated with fuel hedging were a loss of \$60 million in second quarter 2009 versus cash settlement gains of \$511 million in second quarter 2008. However, despite this disparity, overall fuel expense declined year-over-year primarily due to the dramatic decline in physical jet fuel prices. For second quarter 2009, the Company had operating income of \$123 million compared to second quarter 2008 operating income of \$205 million as the reduction in revenues exceeded the benefit of lower Fuel and oil expenses.

The Company continues its diligent cost control efforts and remains committed to maintaining its competitive cost advantage to sustain its strong low fare brand. In addition to other cost containment measures, the Company has a hiring freeze in place, as well as a pay freeze for all Company officers and senior management. Also, during second quarter 2009, the Company announced an early out option available for the vast majority of its Employees. This voluntary separation program provides cash bonuses, health care coverage for a specified period of time, and certain extended flight privileges to eligible Employees who elect early retirement under the program. The program was offered due to overstaffing created by the Company's prior decisions to reduce its capacity during 2009. A total of 1,404 Employees elected to accept this early out offer, which will result in approximately \$70 million in nonrecurring cost to the Company in future periods. The Company currently expects annual savings in subsequent years from the program to eventually exceed the cost incurred. See Note 13 to the unaudited condensed consolidated financial statements for further information. Despite the Company's overall reduction in capacity during 2009, it continues to add flights and new markets through a continual flight schedule optimization, which involves trimming unproductive and less popular flights and reallocating capacity to fund market growth opportunities. During the first half of 2009, the Company began service to Minneapolis-St. Paul (in March) and New York's LaGuardia airport (in June). During the second half of the year, the Company expects to begin service to Boston's Logan International Airport (in August)

and to Milwaukee International Airport (in November.)

Table of Contents

For the six months ended June 30, 2009, the Company had a net loss of \$37 million, or \$.05 loss per share, diluted, versus first half 2008 net income of \$355 million, or \$.48 per share, diluted. Results in each year were also impacted by adjustments related to derivative contracts the Company utilizes in attempting to hedge against jet fuel price increases. Due to the significant unrealized adjustments recorded to "Other (gains) losses, net," which is below the operating income line, the Company believes operating income provides a better indication of the Company's financial performance in both years than does net income. For the six months ended June 30, 2009, the Company had operating income of \$73 million versus first half 2008 operating income of \$293 million. The decline of \$220 million or 75.1 percent primarily was attributable to a 7.9 percent decrease in operating revenues as a result of lower passenger yields, which more than offset realized savings from lower fuel prices.

In second quarter 2009, the Company received eight new Boeing 737-700s and retired four Boeing 737-300 aircraft from service. The Company has only two additional deliveries of new Boeing 737-700s scheduled during the remainder of 2009, but also plans to sell, retire and/or return from lease ten of its existing Boeing 737 aircraft during the remainder of 2009. Overall, the Company currently expects to end 2009 with 535 aircraft, which is a net reduction of two aircraft for the year, and to fly approximately five to six percent fewer ASMs than it flew in 2008. Based on current plans, the Company expects its third quarter 2009 ASM capacity to decrease approximately six to seven percent versus third quarter 2008. The Company's cautious strategy is designed to enable it to match flights with expected demand in the current economic environment. However, the Company believes it has retained the flexibility to enable it to begin growing again once economic conditions improve.

Comparison of three months ended June 30, 2009, to three months ended June 30, 2008

Revenues

Consolidated operating revenues decreased by \$253 million, or 8.8 percent, primarily due to a \$241 million, or 8.8 percent, decrease in Passenger revenues. The majority of the overall decrease in Passenger revenues was due to an 8.2 percent decrease in Passenger yield per Revenue Passenger Mile (RPM yield), as full fare bookings were down versus the prior year and the Company increased the amount of fare discounting and fare sales in response to the decline in demand for air travel amid current domestic economic conditions. As a result of the Company's fare discounting efforts and overall 3.0 percent reduction in ASM capacity (ASMs), load factors increased 1.8 points to 77.0 percent in second quarter 2009. The overall decline in operating revenues, combined with the lower capacity led to a 6.0 percent decline in operating revenue yield per ASM (unit revenue).

The Company has recently implemented several programs and processes that it believes will create substantial opportunities for revenue growth, and has more planned for the fall, including the launch of a new and improved southwest.com, and several related products and initiatives. However, demand for business travel remains weak, and the Company continues to stimulate traffic with more discounted and promotional fares. Unless demand rebounds significantly, the Company expects third quarter 2009 unit revenues to decline year-over-year more than the second quarter 2009 decline of six percent due to more difficult comparisons.

Consolidated freight revenues decreased by \$8 million, or 21.6 percent, primarily due to fewer shipments as a result of the ongoing worldwide recession. The Company expects a comparable decrease in consolidated freight revenues for third quarter 2009 compared to third quarter 2008. Other revenues decreased by \$4 million, or 4.7 percent, compared to second quarter 2008. The majority of the decrease was due to lower charter revenues. Second quarter 2008 charter revenues were unusually high due to bankruptcies

and discontinuation of service by some of the Company's competitors. The Company expects Other revenues for third quarter 2009 to improve versus third quarter 2008, due to a number of revenue improvement initiatives, such as fees recently announced for unaccompanied minors and pets, as well as an increase in fees already charged for Customers checking a third bag.

Operating expenses

Consolidated operating expenses for second quarter 2009 decreased \$171 million, or 6.4 percent, compared to second quarter 2008, versus a 3.0 percent decrease in capacity compared to second quarter 2008. Historically, except for changes in the price of fuel, changes in operating expenses for airlines are typically driven by changes in capacity, or ASMs. The following presents Southwest's operating expenses per ASM for second quarter 2009 and second quarter 2008 followed by explanations of these changes on a per-ASM basis and/or on a dollar basis (in cents, except for percentages):

	Three months ended June 30,		Per ASM	Percent	
	2009	2008	Change	Change	
Salaries, wages, and benefits	3.38	3.19	.19	6.0	
Fuel and oil	2.84	3.59	(.75)	(20.9)	
Maintenance materials					
and repairs	.74	.73	.01	1.4	
Aircraft rentals	.19	.14	.05	35.7	
Landing fees and other rentals	.70	.60	.10	16.7	
Depreciation	.59	.56	.03	5.4	
Other operating expenses	1.32	1.31	.01	.8	
Total	9.76	10.12	(.36)	(3.6)	

Operating expenses per ASM for the three months ended June 30, 2009, were 9.76 cents, a 3.6 percent decrease compared to 10.12 cents for second quarter 2008. The decrease primarily was due to a decline in fuel and oil expense. The Company's fuel cost per gallon, net of hedging, declined 19.4 percent versus second quarter 2008. Excluding fuel and oil, the Company's operating expense per ASM increased versus second quarter 2008 primarily due to higher wage rates. In addition, the decline in capacity versus second quarter 2008 has caused many of the Company's fixed costs to be spread over a smaller quantity of ASMs. On a dollar basis, the majority of the \$171 million overall decrease in operating expenses was due to a \$219 million decline in Fuel and oil expense, the majority of which was due to a lower fuel cost per gallon. Partially offsetting this decrease was a \$24 million increase in salaries, wages, and benefits and a \$20 million increase in landing fees and other rentals. Excluding fuel and third quarter 2009 charges associated with Freedom '09, the Company's voluntary out program, and based on current cost trends and lower available seat miles, the Company expects third quarter 2009 unit costs to increase approximately 7 percent from third quarter 2008's 6.67 cents. Furthermore, the Company is not immune to the effects of the debilitating economic environment. Based on weak travel demand and fuel price volatility, and excluding the impact of Freedom '09 and any ineffectiveness or mark to market adjustments associated with fuel derivative instruments, the Company cannot predict a profitable third quarter 2009. See Note 13 to the unaudited condensed consolidated financial statements for more information on Freedom '09.

Salaries, wages, and benefits expense per ASM for the three months ended June 30, 2009, increased 6.0 percent compared to second quarter 2008, and on a dollar basis increased \$24 million. The majority of the increase per ASM was due to wage rate increases. These rate increases were a result of both completed and ongoing labor contract negotiations with various unionized Employee workgroups and rate increases associated with promotions and increased seniority of existing Employees. This increase was partially offset by a decrease in profitsharing expense versus second quarter 2008. The Company's profitsharing accruals are based on year-to-date income before taxes, primarily excluding unrealized gains and losses from fuel derivative contracts; therefore, profitsharing expense for second quarter 2009 was \$12 million, versus \$37 million in second quarter 2008. On a dollar basis, there was a \$31 million increase in salaries, primarily as a result of higher wage rates, and an \$18 million increase in health benefits expense. These were partially offset by the \$25 million decline in profitsharing expense. Based on current trends and considering ongoing labor negotiations, the Company expects third quarter 2009 salaries, wages, and benefits expense per ASM, excluding charges associated with the Company's voluntary out program, to increase from third quarter 2008's 3.25 cents per ASM.

The Company's Pilots, totaling approximately 5,600 active Employees, are subject to an agreement between the Company and the Southwest Airlines Pilots' Association ("SWAPA"), which became amendable during September 2006. During first quarter 2009, the Company and SWAPA came to a tentative agreement on a new contract extending to 2011. The tentative agreement was voted down by SWAPA membership during second quarter 2009, and the Company has restarted negotiations with SWAPA.

The Company's Customer Service and Reservations Agents, totaling approximately 4,700 active Employees, are subject to an agreement between the Company and the International Association of Machinists and Aerospace Workers, AFL-CIO ("IAM"), which became amendable in November 2008. During second quarter 2009, the Company and IAM came to a tentative agreement on a new contract extending to 2012. The tentative agreement was approved by IAM membership during second quarter 2009.

The Company's Flight Attendants, totaling approximately 9,300 active Employees, are subject to an agreement between the Company and the Transportation Workers of America, AFL-CIO Local 556 ("TWU 556"), which became amendable in June 2008. During first quarter 2009, the Company and TWU 556 came to a tentative agreement on a new contract extending to 2012. The tentative agreement was approved by TWU 556 membership during second quarter 2009.

Fuel and oil expense for the three months ended June 30, 2009, decreased \$219 million, and on a per ASM basis decreased 20.9 percent, primarily due to lower average prices. Excluding hedging, but including related fuel taxes in both years, the Company's average fuel cost per gallon in second quarter 2009 was \$1.63 versus \$3.64 in second quarter 2008. However, the Company had a worse performance from its fuel hedging program in second quarter 2009 versus the same prior year period. As a result of these positions, and overall lower physical prices for crude oil, jet fuel, and related products compared to second quarter 2008, the Company had hedging losses reflected in Fuel and oil expense totaling \$119 million, while second quarter 2008 hedging gains recorded in Fuel and oil expense were \$475 million.

Including the effects of hedging activities, the Company's average fuel cost per gallon in second quarter 2009 was \$1.95, which was 19.4 percent lower than second quarter 2008.

As of June 30, 2009, the Company had fuel derivative instruments in place for approximately 30 percent of its expected third quarter 2009 jet fuel consumption on an economic basis, the majority of which effectively cap prices in the low \$70 per barrel range of crude oil. In addition to these positions, the Company also had unsettled fuel derivative instruments relating to fourth quarter 2009 through 2013 whereby it has previously fixed some losses that will impact earnings in these future periods. The Company's current "economic hedge" position for third quarter 2009 excludes these previously "fixed" fuel contracts.

As a result of these previously fixed losses, the Company expects to pay higher than market prices for fuel for the remainder of 2009 through 2013. In addition, as a result of previous hedges that have been "undesignated" as defined in SFAS 133, the Company has significant amounts "frozen" in AOCI that will be recognized in earnings in future periods when the underlying fuel derivative contracts settle. As discussed in Note 6 to the unaudited condensed consolidated financial statements, the Company has deferred losses in AOCI of \$750 million, net of tax, related to fuel derivative contracts. The estimated fair market value (as of June 30, 2009) of the Company's net fuel derivative contracts for the remainder of 2009 through 2013 reflects a net liability of approximately \$338 million, including the effect of \$425 million in cash collateral that had been provided to counterparties as of June 30, 2009, which has been netted against the Company's liability in the unaudited Condensed Consolidated Balance Sheet. The following table displays the Company's estimated fair value of remaining fuel derivative contracts (not considering the impact of the \$425 million in cash collateral provided to counterparties) as well as the amount of deferred losses in AOCI at June 30, 2009, and the expected future periods in which these items are expected to settle and/or be recognized in earnings (in millions):

		Amount
	Fair value	of
	(liability)	(losses)
	of fuel	deferred
		in AOCI
	derivative	at June
	contracts	30,
	at June	2009 (net
Year	30, 2009	of tax)
2009	\$ (92)	\$ (153)
2010	\$ (160)	\$ (224)
2011	\$ (221)	\$ (157)
2012	\$ (155)	\$ (109)
2013	\$ (135)	\$ (107)
Total	\$ (763)	\$ (750)

Based on this liability at June 30, 2009 (and precluding any other subsequent changes to the fuel hedge portfolio), the Company's jet fuel costs per gallon are expected to exceed market (i.e., unhedged prices) during each of these periods. This is based primarily on expected future cash settlements associated with fuel derivatives, but excludes any SFAS 133 impact associated with the ineffectiveness of fuel hedges or fuel derivatives that are marked to market value because they did not qualify for special hedge accounting. See Note 5 to the unaudited condensed consolidated financial statements for further information. Based on the Company's derivative position and market prices as of July 20, 2009, the Company is currently estimating its third quarter 2009 jet fuel cost per gallon to decline significantly year-over-year to approximately \$2.15 per gallon, including fuel related taxes, but excluding the effects of any

ineffectiveness and mark to market adjustments from the Company's fuel hedging program. For full year 2009, the Company currently estimates its jet fuel cost per gallon, including fuel related taxes, but excluding the effects of any ineffectiveness and mark to market adjustments from the Company's fuel hedging program, to be less than \$2.00.

The Company has also continued its efforts to conserve fuel and, by the end of 2009, expects to complete the installation of Aviation Partners Boeing Blended Winglets on a total of 112 of its 737-300 aircraft (all 737-700 aircraft have already been equipped with winglets). This and other fuel conservation efforts resulted in an approximate 1.4 percent decrease in the Company's fuel burn rate per ASM for second quarter 2009 versus second quarter 2008.

Maintenance materials and repairs expense for the three months ended June 30, 2009, was approximately flat on a dollar basis compared to second quarter 2008, but increased 1.4 percent on a per-ASM basis compared to second quarter 2008. The increase on a per-ASM basis primarily was associated with a larger decrease in ASMs than the decrease in expense. In late June 2008, the Company transitioned to a new engine repair agreement related to its 737-700 aircraft fleet and expense is now based on flight hours associated with 737-700 engines. Considering the new agreement, the Company expects Maintenance materials and repairs per ASM for third quarter 2009 to be in the high .70s cents per ASM range, based on currently scheduled airframe maintenance events and projected engine hours flown.

Aircraft rentals per ASM for the three months ended June 30, 2009, increased 35.7 percent compared to second quarter 2008, and, on a dollar basis, increased \$9 million. Both of these increases primarily were due to the Company's recent sale and leaseback transactions involving a total of 16 Boeing 737-700 aircraft. See Note 14 to the unaudited condensed consolidated financial statements for further information. As a result of these transactions, the Company expects aircraft rentals per ASM in third quarter 2009 to increase from the .19 cents experienced during second quarter 2009.

Landing fees and other rentals for the three months ended June 30, 2009, increased \$20 million on a dollar basis, and increased 16.7 percent on a per ASM basis compared to second quarter 2008. The majority of these increases were due to higher space rentals in airports as a result of higher rates charged by those airports for gate and terminal space. A portion of these higher rates charged by airports is due to the fact that other airlines reduced capacity at a faster pace than the Company's 3.0 percent reduction, resulting in the Company incurring a higher percentage of total airport-related costs. As a consequence, the Company currently also expects Landing fees and other rentals per ASM in third quarter 2009 to be higher than the .70 cents per ASM recorded in second quarter 2009, primarily due to these higher rates.

Depreciation expense for the three months ended June 30, 2009, increased by \$2 million on a dollar basis compared to second quarter 2008, and increased 5.4 percent on a per-ASM basis. Both of these increases primarily were due to capitalized software costs associated with various Company projects. On a dollar basis, this increase was mostly offset by the Company's execution of sale and leasebacks of 16 Boeing 737-700 aircraft over the past twelve months. See Note 14 to the unaudited condensed consolidated financial statements. For third quarter 2009, the Company expects Depreciation expense per ASM to increase versus second quarter 2009's .59 cents as a result of an anticipated decline in ASMs.

Other operating expenses per ASM for the three months ended June 30, 2009, were flat compared to second quarter 2008, and on a dollar basis, decreased \$6 million. On a dollar basis, the decrease was related to a decline in bad debts related to revenues from credit card sales. For third quarter 2009, the Company currently expects Other operating expenses per ASM to be higher than third quarter 2008's 1.33 cents due to higher expected advertising expense.

Through the 2003 Emergency Wartime Supplemental Appropriations Act, the federal government has continued to provide renewable, supplemental, first-party war-risk insurance coverage to commercial carriers, at substantially lower premiums than prevailing commercial rates and for levels of coverage not available in the commercial market. The government-provided supplemental coverage from the Wartime Act is currently set to expire on August 31, 2009. Although another extension beyond this date is expected, if such coverage is not extended by the government, the Company could incur substantially higher insurance costs or unavailability of adequate coverage in future periods.

Other

Interest expense for the three months ended June 30, 2009, increased \$15 million, or 46.9 percent, compared to second quarter 2008, primarily due to the Company's borrowing under its \$600 million term loan in May 2008, its December 2008 issuance of \$400 million of secured notes, and the Company's borrowing under its \$332 million term loan in May 2009. As a result of these transactions, the Company also expects higher year-over-year interest expense for third quarter 2009. See Note 9 to the unaudited condensed consolidated financial statements.

Capitalized interest for the three months ended June 30, 2009, decreased \$1 million, or 16.7 percent, compared to the same prior year period primarily due to a decline in interest rates and a decrease in progress payment balances for scheduled future aircraft deliveries.

Interest income for the three months ended June 30, 2009, decreased by \$2 million, or 40.0 percent, compared to the same prior year period, primarily due to a decrease in rates earned on invested cash and short-term investments. In second quarter 2008, the Company's cash and cash equivalents and short-term investments included a significant amount of collateral deposits received from a counterparty of the Company's fuel derivative instruments. Although these amounts were not restricted in any way, the Company was required to remit the investment earnings from these amounts back to the counterparty. See Item 3 of Part I for further information on collateral deposits and Note 5 to the unaudited condensed consolidated financial statements for further information on fuel derivative instruments.

Other (gains) losses, net, primarily includes amounts recorded in accordance with the Company's hedging activities and SFAS 133. The following table displays the components of Other (gains) losses, net, for the three months ended June 30, 2009 and 2008:

	Three m June 30,	nonths ended	
(In millions)	2009	2008	
Mark-to-market impact from fuel contracts settling in future			
periods - included in Other (gains) losses, net	\$20	\$(369)
Ineffectiveness from fuel hedges settling in future periods -			
included in Other (gains) losses, net	(24) 14	
Realized ineffectiveness and mark-to-market (gains) or			
losses - included in Other (gains) losses, net	(2) (6)
Premium cost of fuel contracts included in Other (gains) losses, net	37	14	
Other	3	2	
	\$34	\$(345)

Based on the Company's current fuel derivative contracts position, for the expense related to amounts excluded from the Company's measurements of hedge effectiveness (i.e., the premium cost of option and collar derivative contracts), the Company expects expense of approximately \$30 million relating to these items in third quarter 2009.

The Company's effective tax rate was a negative 7.3 percent in second quarter 2009 compared to 39.3 percent in second quarter 2008. The lower rate in second quarter 2009 primarily was due to a \$21 million adjustment (\$.03 per share, diluted) necessary to achieve a revised year-to-date tax rate of 35.0 percent based on the Company's currently forecasted results for the full year 2009.

Comparison of six months ended June 30, 2009, to six months ended June 30, 2008

Revenues

Consolidated operating revenues decreased by \$427 million, or 7.9 percent, primarily due to a \$403 million, or 7.8 percent, decrease in Passenger revenues. Over 70 percent of the overall decrease in Passenger revenues was due to a 5.7 percent decrease in Passenger yield per Revenue Passenger Mile (RPM yield), as full fare bookings were down versus the prior year and the Company increased the amount of fare discounting and fare sales in response to the decline in demand for air travel amid current domestic economic conditions. The majority of the remainder of the Passenger revenue decrease was due to the 3.5 percent reduction in capacity (ASMs) versus the prior year. These declines were partially offset by a slight increase in load factors for the six months ended June 30, 2009 versus the same prior year period, as a result of the Company's fare discounting efforts and overall reduction in capacity. The overall decline in operating revenues combined with the lower capacity led to a 4.6 percent decline in operating revenue yield per ASM (unit revenue).

Consolidated freight revenues decreased by \$13 million, or 18.3 percent, primarily due to fewer shipments as a result of the ongoing worldwide recession. Other revenues decreased by \$11 million, or 6.6 percent, compared to the first half of 2008. The majority of the decrease was due to lower charter revenues. Charter revenues for the six months ended June 30, 2008, were unusually high due to bankruptcies and discontinuation of service by some of the Company's competitors.

Operating expenses

Consolidated operating expenses for first half 2009 decreased \$207 million, or 4.1 percent, compared to first half 2008, versus a 3.5 percent decrease in capacity compared to the first half of the prior

Table of Contents

year. Historically, except for changes in the price of fuel, changes in operating expenses for airlines are typically driven by changes in capacity, or ASMs. The following presents Southwest's operating expenses per ASM for the six months ended June 30, 2009 and 2008 followed by explanations of these changes on a per-ASM basis and/or on a dollar basis (in cents, except for percentages):

	Six months ended June 30,		Per ASM	Percent	
	2009	2008	Change	Change	
Salaries, wages, and benefits	3.42	3.18	.24	7.5	
Fuel and oil	2.86	3.38	(.52)	(15.4)	
Maintenance materials					
and repairs	.75	.65	.10	15.4	
Aircraft rentals	.19	.15	.04	26.7	
Landing fees and other rentals	.69	.64	.05	7.8	
Depreciation	.60	.57	.03	5.3	
Other operating expenses	1.34	1.34	-	-	
Total	9.85	9.91	(.06)	(.6)	

Operating expenses per ASM for the six months ended June 30, 2009, were 9.85 cents, slightly lower compared to 9.91 cents for first half 2008. This decrease primarily was due to a decline in fuel costs, which was mostly offset by higher wages, as a result of higher wage rates, and higher maintenance materials and repairs expense. In addition, the decline in capacity versus first half 2008 has caused the Company's fixed costs to be spread over a smaller quantity of ASMs. On a dollar basis, the majority of the \$207 million overall decrease was due to a \$322 million decline in Fuel and oil expense. Approximately 74 percent of this decrease was due to a lower fuel cost per gallon and the remainder was due to the reduction in fuel consumption. The decline in Fuel and oil expense was partially offset by increases in salaries, wages, and benefits and maintenance materials and repairs expense.

Salaries, wages, and benefits expense per ASM for the six months ended June 30, 2009, increased 7.5 percent compared to first half 2008, and on a dollar basis increased \$60 million. These increases primarily were due to wage rate increases. These rate increases were a result of both completed and ongoing labor contract negotiations with various unionized Employee workgroups and rate increases associated with promotions and increased seniority of existing Employees. This increase was partially offset by a \$37 million, or 74.7 percent decrease in profitsharing expense versus first half 2008.

Fuel and oil expense for the six months ended June 30, 2009, decreased \$322 million, and on a per ASM basis decreased 15.4 percent, primarily due to lower average prices, excluding the impact of hedging. Excluding hedging, but including related fuel taxes in both years, the Company's average fuel cost per gallon in first half 2009 was \$1.60 versus \$3.29 in first half 2008. Primarily as a result of a difference in the fuel derivative portfolio held by the Company in 2009 versus 2008, and overall lower physical prices for crude oil, jet fuel, and related products compared to first half 2008, the Company had hedging losses reflected in Fuel and oil expense totaling \$265 million (of which \$125 million was cash settlements paid to counterparties and \$140 million was unrealized losses associated with derivative contracts settling in the first half of 2009), while first half 2008 hedging gains recorded in Fuel and oil expense were \$766 million. Including the effects of hedging activities, the Company's average fuel cost per gallon in first half 2009 was \$1.97, which was 13.6 percent lower than first half 2008. The Company's fuel conservation efforts resulted in an approximate 2.1 percent decrease in the Company's fuel burn rate per ASM for first half 2009 versus first half 2008.

Maintenance materials and repairs expense for the six months ended June 30, 2009, increased \$40 million or 12.0 percent on a dollar basis compared to first half 2008, and increased 15.4 percent on a per-ASM basis compared to first half 2008. On both a dollar and a per-ASM basis, the increases compared to first half 2008 were due to higher engine costs related to the Company's 737-700 aircraft. For most of the first half of 2008, these aircraft engines had been accounted for on a time and materials basis, and there were relatively few repair events for these engines during that period. This was due to the fact that the 737-700 is the newest aircraft type in the Company's fleet, and there were not yet a significant number of engines on these aircraft that were due for their first major overhaul. In late June 2008, the Company transitioned to a new engine repair agreement for these aircraft and expense is now based on flight hours associated with 737-700 engines. The expense for 737-700 engines recognized in the first half of 2009 associated with the current agreement significantly exceeded the expense recognized in first half 2008, when repairs were still being accounted for on a time and materials basis.

Aircraft rentals per ASM for the six months ended June 30, 2009, increased 26.7 percent compared to first half 2008, and, on a dollar basis, increased \$17 million. Both of these increases primarily were due to the Company's recent sale and leaseback transactions involving a total of 16 Boeing 737-700 aircraft. See Note 14 to the unaudited condensed consolidated financial statements for further information.

Landing fees and other rentals for the six months ended June 30, 2009, increased \$15 million on a dollar basis, and on a per ASM basis was 7.8 percent higher than first half 2008. The majority of the increases on both a dollar and a per ASM basis were due to higher space rentals in airports as a result of higher rates charged by those airports for gate and terminal space. A portion of these higher rates charged by airports were due to the fact that other airlines reduced capacity at a faster pace than the Company, resulting in the Company incurring a higher percentage of total airport-related costs.

Depreciation expense for the six months ended June 30, 2009, increased by \$7 million on a dollar basis compared to first half 2008, and increased 5.3 percent on a per-ASM basis. The increase on a dollar basis primarily was due to the Company's net addition of eight Boeing 737s to its fleet over the past twelve months. This included the purchase of 16 new 737-700s from Boeing, net of six 737-300s returned from lease and two owned 737-300 aircraft retired from service. In addition, the Company executed sale and leasebacks of 16 737-700 aircraft. See Note 14 to the unaudited condensed consolidated financial statements for information on some of these recent transactions. The increase on a per-ASM basis primarily was due to the increase in the Company's fleet size combined with a decrease in ASMs as a result of the Company's decision to slow its growth given current economic conditions.

Other operating expenses per ASM for the six months ended June 30, 2009, were flat compared to first half 2008, but on a dollar basis decreased \$24 million. On a dollar basis, the decrease primarily was related to a decline in bad debts

related to revenues from credit card sales.

Other

Interest expense for the six months ended June 30, 2009, increased \$32 million, or 53.3 percent, compared to first half 2008, primarily due to the Company's borrowing under its \$600 million term loan in May 2008, its December 2008 issuance of \$400 million of secured notes, and the Company's borrowing under its \$332 million term loan in May 2009.

Capitalized interest for the six months ended June 30, 2009, decreased \$3 million, or 21.4 percent, compared to the same prior year period primarily due to a decline in interest rates and a decrease in progress payment balances for scheduled future aircraft deliveries.

Interest income for the six months ended June 30, 2009, decreased by \$4 million, or 33.3 percent, compared to the same prior year period, primarily due to a decrease in rates earned on invested cash and short-term investments.

Other (gains) losses, net, primarily includes amounts recorded in accordance with the Company's hedging activities and SFAS 133. The following table displays the components of Other (gains) losses, net, for the six months ended June 30, 2009 and 2008:

	30,	nths ended June	2
(In millions)	2009	2008	
Mark-to-market impact from fuel contracts settling in future			
periods - included in Other (gains) losses, net	\$18	\$(373)
Ineffectiveness from fuel hedges settling in future periods -			
included in Other (gains) losses, net	(11) 19	
Realized ineffectiveness and mark-to-market (gains) or			
losses - included in Other (gains) losses, net	(23) 17	
Premium cost of fuel contracts included in Other (gains) losses, net	69	27	
Other	4	3	
	\$57	\$(307)

The Company's effective tax rate was 35.0 percent in first half 2009 compared to 39.3 percent in first half 2008. The lower rate in first half 2009 primarily was due to the Company's lower expected financial results for 2009 and the related impact that permanent tax differences have on these projections.

Liquidity and Capital Resources

Net cash provided by operating activities was \$135 million for the three months ended June 30, 2009, compared to \$2.3 billion provided by operating activities in the same prior year period. For the six months ended June 30, 2009, net cash provided by operations was \$420 million compared to \$3.3 billion for the same prior year period. The operating cash flows for the first half of 2008 were largely impacted by counterparty deposits received associated with the Company's fuel hedging program. There was an increase in counterparty deposits of \$2.4 billion for the six months ended June 30, 2008, versus a decrease (outflow) of \$185 million during the six months ended June 30,

2009. Counterparty deposits are netted against the fair value of the fuel derivative instruments to which they relate in the unaudited Condensed Consolidated Balance Sheet—see Note 5 to the unaudited condensed consolidated financial statements. The fluctuations in these deposits in both years have been due to large changes in the fair value of the Company's fuel derivatives portfolio. See also Item 3 of Part I for further information. Cash flows from operating activities for both years were also impacted by changes in Air traffic liability. For the six months ended June 30, 2009, there was a \$244 million increase in Air traffic liability, as a result of bookings for future travel. This compared to the prior year's \$372 million increase in Air traffic liability. Net cash provided by operating activities is primarily used to finance capital expenditures and provide working capital.

Net cash flows used in investing activities during the three months ended June 30, 2009, totaled \$377 million compared to \$1.3 billion used in investing activities in the same prior year period. For the six months ended June 30, 2009, net cash used in investing activities was \$1.0 billion compared to \$1.4 billion used in investing activities for the same 2008 period. Investing activities for the first half of both years consisted of payments for new 737-700 aircraft delivered to the Company and progress payments for future aircraft deliveries, as well as changes in the balance of the Company's short-term investments and noncurrent investments. During the six months ended June 30, 2009, the Company's short-term and noncurrent investments increased by a net \$743 million, versus a net increase of \$802 million during the same prior year period.

Net cash provided by financing activities during the three months ended June 30, 2009, was \$43 million compared to \$599 million provided by financing activities for the same period in 2008. For the six months ended June 30, 2009, net cash provided by financing activities was \$172 million versus \$529 million provided by financing activities for the same 2008 period. During the six months ended June 30, 2009, the Company raised \$381 million from the sale and leaseback of eleven 737-700 aircraft and borrowed \$332 million under a term loan agreement. See Notes 9 and 14 to the unaudited condensed consolidated financial statements for further information. Also during the six months ended June 30, 2009, the Company repaid the \$400 million and \$91 million it had borrowed during 2008 under its revolving credit agreement and a credit line borrowing, respectively. During the six months ended June 30, 2008, the Company borrowed \$600 million under a term loan agreement entered into during May 2008, and repurchased \$54 million of its Common Stock, representing a total of 4.4 million shares.

The Company has a "well-known seasoned issuer" universal shelf registration statement, effective April 3, 2009, to register an indeterminate amount of debt or equity securities for future sales. The Company intends to use the proceeds from any future securities sales off this shelf for general corporate purposes. The Company has not issued any securities under this shelf registration statement to date.

Contractual Obligations and Contingent Liabilities and Commitments

Southwest has contractual obligations and commitments primarily for future purchases of aircraft, payment of debt, and lease arrangements. Through the first six months of 2009, the Company purchased eleven new 737-700 aircraft from Boeing and is scheduled to receive two more 737-700 aircraft from Boeing during the remainder of 2009. The Company also retired five of its older 737-300 aircraft from service (three leased and two owned) during the first six months of 2009. The Company also completed the sale and

leaseback of eleven of its previously owned 737-700 aircraft during the first half of 2009. However, these transactions have no impact on the Company's future aircraft commitments with Boeing. See Note 14 to the unaudited condensed consolidated financial statements for further information on recent sale and leaseback transactions. Based on recent economic events and announced industry capacity reductions, the Company continues to evaluate its plans with regards to planned aircraft retirements and future deliveries from Boeing. The Company currently plans to reduce its fleet by ten additional aircraft during the remainder of 2009 through a combination of lease returns, aircraft sales, and/or retirements, resulting in a fleet totaling 535 Boeing 737 aircraft as of December 31, 2009. As of July 20, 2009, Southwest's firm orders and options to purchase new 737-700 aircraft from Boeing are reflected in the following table:

	The	Boeing Con	npany	
	Di	Ontions	Purchase	T-4-1
	Firm	Options	Rights	Total
2009	13	-	-	13 *
2010	10	-	-	10
2011	10	10	-	20
2012	13	10	-	23
2013	19	4	-	23
2014	13	7	-	20
2015	14	3	-	17
2016	12	11	-	23
2017	-	17	-	17
Through				
2018	-	-	54	54
Total	104	62	54	220

^{*} Currently plan to reduce fleet by 15 aircraft, bringing 2009 to a net reduction of two aircraft.

The following table details information on the 543 aircraft in the Company's fleet that were in service as of June 30, 2009:

737 Type	Seats	Average Age (Yrs)	Number of Aircraft	Number Owned	Number Leased
-300	137	17.8	180	110	70
-500	122	18.2	25	16	9
-700	137	5.6	338	318	20
TOTALS		10.2	543	444	99

The Company has the option, which must be exercised two years prior to the contractual delivery date, to substitute -600s or -800s for the -700s. Based on the above delivery schedule, aggregate funding needed for firm aircraft commitments was approximately \$3.3 billion, subject to adjustments for inflation, due as follows: \$124 million

remaining in 2009, \$344 million in 2010, \$450 million in 2011, \$571 million in 2012, \$634 million in 2013, \$508 million in 2014, and \$693 million thereafter.

During October 2008, the Company elected to access \$400 million of the available \$600 million under its revolving credit facility. During May 2009, the Company repaid this borrowing, thus leaving the entire credit facility available for future borrowings. The Company has various options available to meet its capital and operating commitments, including cash on hand and short-term investments at June 30, 2009, of \$2.2 billion, internally generated funds, and its \$600 million revolving credit facility that expires in August 2010. Prior to its expiration, the Company anticipates that this revolving credit facility will either be renewed or replaced with a revolving credit facility that will provide sufficient capital resources to fund the Company's ongoing liquidity needs. However, continued disruption in U.S. and international credit markets may adversely affect the Company's ability to renew or replace this credit facility and any new or renewed facility may be under terms that are not as favorable as past credit agreements. As discussed in Note 15 to the unaudited condensed consolidated financial statements, on July 1, 2009, the Company borrowed \$124 million under a term loan agreement secured by five of the Company's 737-700 aircraft. The Company will also consider other various borrowing or leasing options to supplement cash requirements as necessary.

On July 22, 2009, Moody's downgraded the Company's senior unsecured debt rating from "Baa1" to "Baa3" and also lowered the ratings of the Company's Pass-through Enhanced Equipment Trust Certificates ("PTC" and "EETC"). The downgrade of the Company's senior unsecured debt rating was based on Moody's expectation of continuing weak fundamentals of the domestic airline sector. The downgrade of the Company's ratings on its PTC's and EETC's reflects the reduction in the Company's underlying credit quality, and with respect to the PTC's the elevated loan to value ratios resulting from the older vintage 737-300 aircraft that are pledged as collateral for these transactions. Also, on July 23, 2009, Fitch downgraded the Company's senior unsecured debt rating from "BBB+" to "BBB," based on the ongoing impact of the recession and the collapse in full-fare passenger demand on the Company's cash flow generation power and leverage at a time when jet fuel prices remain volatile and unpredictable. While the Company's credit rating remains "investment grade," these lower ratings will likely result in a slight increase in its borrowing costs on a prospective basis.

During 2008, the City of Dallas approved the Love Field Modernization Program, an estimated \$519 million project to reconstruct Dallas Love Field with modern, convenient air travel facilities. Pursuant to a Program Development Agreement with the City of Dallas, Southwest is managing this project, and initial construction is expected to commence during late 2009, with completion scheduled for October 2014. Bonds will be issued at a later date by the Love Field Airport Modernization Corporation (a "local government corporation" under Texas law formed by the City of Dallas) that will provide funding for this project, with repayment of the bonds being made through recurring ground rents, fees, and other revenues collected by the airport.

Table of Contents

In January 2008, the Company's Board of Directors authorized the repurchase of up to \$500 million of the Company's Common Stock. Repurchases may be made in accordance with applicable securities laws in the open market or in private transactions from time to time, depending on market conditions. The Company had repurchased 4.4 million shares for a total of \$54 million as part of this program through February 15, 2008; however, the Company has not repurchased any additional shares from that date through the date of this filing. The Company does not believe it is prudent to repurchase shares at the current time considering the current economic environment.

Fair value measurements

As discussed in Note 11 to the unaudited condensed consolidated financial statements, the Company uses the provisions of Statement of Financial Accounting Standards No. 157 (SFAS 157) in determining the fair value of certain assets and liabilities. As defined in SFAS 157, the Company has determined that it uses unobservable (Level 3) inputs in determining the fair value of its auction rate security investments, valued at \$192 million, a portion of its fuel derivative contracts, which totaled a net liability of \$137 million, and \$8 million in other investments, at June 30, 2009.

All of the Company's auction rate security instruments are reflected at estimated fair value in the unaudited Condensed Consolidated Balance Sheet. At June 30, 2009, approximately \$109 million of these instruments are classified as available for sale securities and \$83 million are classified as trading securities. In early 2008 and prior periods, due to the auction process which took place every 30-35 days for most securities, quoted market prices were readily available, which would have qualified as Level 1 under SFAS 157. However, due to events in credit markets beginning during first quarter 2008, the auction events for most of these instruments failed, and, therefore, the Company has determined the estimated fair values of these securities utilizing a discounted cash flow analysis or other type of valuation model as of June 30, 2009. In addition, the Company recently performed a valuation of its auction rate security instruments and considered these valuations in determining estimated fair values of other similar instruments within its portfolio. The Company's analyses consider, among other items, the collateralization underlying the security investments, the expected future cash flows, including the final maturity, associated with the securities, and estimates of the next time the security is expected to have a successful auction or return to full par value. These securities were also compared, when possible, to other securities not owned by the Company, but with similar characteristics. Due to these events, the Company reclassified these instruments as Level 3 during first quarter 2008.

In association with this estimate of fair value, the Company has recorded a temporary unrealized decline in fair value of \$11 million, with an offsetting entry to "Accumulated other comprehensive income (loss)." Given the quality and backing of the Company's auction rate securities held, the fact that the Company has not yet recorded a loss on the sale of any of these instruments, and the fact that it has been able to periodically sell instruments in the auction process, it believes it can continue to account for the estimated reduction in fair value of its remaining securities as temporary. These conclusions will also continue to be evaluated and challenged in subsequent periods. The Company currently believes that this temporary decline in fair value is due entirely to liquidity issues, because the underlying assets for the majority of securities are almost entirely backed by the U.S. Government. In addition, for the \$109 million in instruments classified as available for sale, these auction rate securities represented approximately five percent of the Company's total cash, cash equivalent, and investment balance at June 30, 2009.

Table of Contents

Considering the relative significance of these securities in comparison to the Company's liquid assets and other sources of liquidity, the Company has no current intention of selling these securities nor does it expect to be required to sell these securities before a recovery in their cost basis. For the \$83 million in instruments classified as trading securities, the Company has entered into an agreement with the counterparty that allows the Company to put the instruments back to the counterparty at full par value in June 2010. Part of this agreement also contains a line of credit in which the Company can borrow up to \$83 million as a loan from the counterparty that would be secured by the auction rate security instruments from that counterparty. There were no borrowings under this provision as of June 30, 2009. At the time of the first failed auctions during first quarter 2008, the Company held a total of \$463 million in securities. Since that time, the Company has been able to sell \$260 million of these instruments at par value, in addition to the \$83 million subject to the agreement to be sold at par in June 2010. The Company is also in discussions with other counterparties to determine whether mutually agreeable decisions can be reached regarding the effective repurchase of its remaining securities.

The Company determines the value of fuel derivative option contracts utilizing a standard option pricing model based on inputs that are either readily available in public markets, can be derived from information available in publicly quoted markets, or are quoted by its counterparties. In situations where the Company obtains inputs via quotes from its counterparties, it verifies the reasonableness of these quotes via similar quotes from another counterparty as of each date for which financial statements are prepared. The Company has consistently applied these valuation techniques in all periods presented and believes it has obtained the most accurate information available for the types of derivative contracts it holds. Due to the fact that certain inputs used in determining estimated fair value of its option contracts are considered unobservable (primarily volatility), as defined in SFAS 157, the Company has categorized these option contracts as Level 3.

As discussed in Note 5 to the unaudited condensed consolidated financial statements, any changes in the fair values of fuel derivative instruments are subject to the requirements of SFAS 133. Any changes in fair value of cash flow hedges that are considered to be effective, as defined, are offset within "Accumulated other comprehensive income (loss)" until the period in which the expected cash flow impacts earnings. Any changes in the fair value of fuel derivatives that are ineffective, as defined, or do not qualify for special hedge accounting, are reflected in earnings within "Other (gains)/losses, net," in the period of the change. Because the Company has extensive historical experience in valuing the derivative instruments it holds, and such experience is continually evaluated against its counterparties each period when such instruments expire and are settled for cash, the Company believes it is unlikely that an independent third party would value the Company's derivative contracts at a significantly different amount than what is reflected in the Company's financial statements. In addition, the Company also has bilateral credit provisions in some of its counterparty agreements, which provide for parties (or the Company) to provide cash collateral when the fair values of fuel derivatives with a single party exceed certain threshold levels. Since this cash collateral is based on the estimated fair value of the Company's outstanding fuel derivative contracts, this provides further validation to the Company's estimate of fair values.

Forward-looking statements

Some statements in this Form 10-Q may be "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Forward-looking statements are based on, and include statements about, Southwest's estimates, expectations, beliefs, intentions, and strategies for the future, and the assumptions underlying these forward-looking statements. Specific forward-looking statements can be identified by the fact that they do not relate strictly to historical or current facts and include, without limitation, statements related to the Company's (i) growth strategies and expectations; (ii) revenues, cost-cutting and other financial and operational strategies and initiatives and related expectations for future results of operations; (iii) expectations regarding liquidity, including anticipated needs for, and sources of, funds; (iv) plans and

expectations for

Table of Contents

managing risk associated with changing jet fuel prices; and (v) expectations and intentions relating to outstanding litigation. While management believes these forward-looking statements are reasonable as and when made, forward-looking statements are not guarantees of future performance and involve risks and uncertainties that are difficult to predict. Therefore, actual results may differ materially from what is expressed in or indicated by the Company's forward-looking statements or from historical experience or the Company's present expectations. Factors that could cause these differences include, among others:

- (i) continued unfavorable economic conditions, which could continue to impact the demand for air travel and the Company's ability to adjust fares;
- (ii) continued volatility in the price and availability of aircraft fuel and the impact of hedge accounting and any changes in the Company's fuel hedging strategies and positions;
- (iii) the Company's ability to timely and effectively prioritize its revenue and cost reduction initiatives and its related ability to timely and effectively implement, transition, and maintain the necessary information technology systems and infrastructure to support these initiatives;
 - (iv) the results of pending labor negotiations;
- (v) continued instability of the credit, capital, and energy markets, which could result in future pressure on credit ratings and could also negatively impact the Company's ability to obtain financing on acceptable terms and the Company's liquidity generally;
 - (vi) the impact of aircraft industry incidents and the economy on the future availability and cost of insurance;
- (vii) the impact of technological initiatives on the Company's technology infrastructure, including its point of sale, ticketing, revenue accounting, payroll and financial reporting areas;
- (viii) the extent and timing of the Company's investment of incremental operating expenses and capital expenditures to develop and implement its initiatives and its corresponding ability to effectively control its operating expenses;
- (ix) the Company's dependence on third party arrangements to assist with implementation of certain of its initiatives;
- (x) the impact of governmental regulations and inquiries on the Company's operating costs, as well as its operations generally, and the impact of developments affecting the Company's outstanding litigation;
 - (xi) competitor capacity and load factors; and
- (xii) other factors as set forth in the Company's filings with the Securities and Exchange Commission, including the detailed factors discussed under the heading "Risk Factors" in the Company's Annual Report on Form 10-K for the year ended December 31, 2008.

Caution should be taken not to place undue reliance on the Company's forward-looking statements, which represent the Company's views only as of the date this report is filed. The Company undertakes no obligation to update publicly or revise any forward-looking statement, whether as a result of new information, future events or otherwise.

Table of Contents

Item 3. Quantitative and Qualitative Disclosures About Market Risk

As discussed in Note 5 to the unaudited condensed consolidated financial statements, the Company uses financial derivative instruments to hedge its exposure to material increases in jet fuel prices. At June 30, 2009, the estimated gross fair value of outstanding contracts was a liability of \$763 million.

Outstanding financial derivative instruments expose the Company to credit loss in the event of nonperformance by the counterparties to the agreements. However, the Company does not expect any of the counterparties to fail to meet their obligations. The credit exposure related to these financial instruments is represented by the fair value of contracts with a positive fair value at the reporting date. To manage credit risk, the Company selects and periodically reviews counterparties based on credit ratings, limits its exposure to a single counterparty, and monitors the market position of the program and its relative market position with each counterparty. At June 30, 2009, the Company had agreements with all of its counterparties containing early termination rights and/or bilateral collateral provisions whereby security is required if market risk exposure exceeds a specified threshold amount or credit ratings fall below certain levels. At June 30, 2009, the Company had provided \$425 million in fuel derivative related cash collateral deposits under these bilateral collateral provisions to counterparties, but did not hold any cash collateral deposits from any of its counterparties as of that date. These collateral deposits are netted against the fair value of the Company's noncurrent derivative contracts in Other deferred liabilities in the unaudited Condensed Consolidated Balance Sheet. Cash flows as of and for a particular operating period are included as Operating cash flows in the unaudited Condensed Consolidated Statement of Cash Flows.

On July 22, 2009, Moody's downgraded the Company's senior unsecured debt rating from "Baa1" to "Baa3" and also lowered the ratings of the Company's Pass-through Enhanced Equipment Trust Certificates ("PTC" and "EETC"). The downgrade of the Company's senior unsecured debt rating was based on Moody's expectation of continuing weak fundamentals of the domestic airline sector. The downgrade of the Company's ratings on its PTC's and EETC's reflects the reduction in the Company's underlying credit quality, and with respect to the PTC's the elevated loan to value ratios resulting from the older vintage 737-300 aircraft that are pledged as collateral for these transactions. Also, on July 23, 2009, Fitch downgraded the Company's senior unsecured debt rating from "BBB+" to "BBB," based on the ongoing impact of the recession and the collapse in full-fare passenger demand on the Company's cash flow generation power and leverage at a time when jet fuel prices remain volatile and unpredictable. While the Company's credit rating remains "investment grade," these lower ratings will likely result in a slight increase in its borrowing costs on a prospective basis.

See Item 7A "Quantitative and Qualitative Disclosures About Market Risk" in the Company's Annual Report on Form 10-K for the year ended December 31, 2008, and Note 5 to the unaudited condensed consolidated financial statements in this Form 10-Q for further information about Market Risk.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

The Company maintains disclosure controls and procedures (as defined in Rule 13a-15(e) of the Securities Exchange Act) designed to provide reasonable assurance that the information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms. These include controls and procedures designed to ensure that this information is accumulated and communicated to the Company's management, including its Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure. Management, with the participation of the Chief Executive Officer and Chief Financial Officer, evaluated the effectiveness of the Company's disclosure controls and procedures as of June 30, 2009. Based on this evaluation,

the Company's Chief Executive Officer and Chief Financial Officer have concluded that the Company's disclosure controls and procedures were effective as of June 30, 2009, at the reasonable assurance level.

Table of Contents

Changes in Internal Control over Financial Reporting

During second quarter 2009, the Company implemented a new fuel derivative instrument management and valuation software which resulted in a material change in a component of the Company's internal control over financial reporting. Pre-implementation testing was conducted by management to ensure that internal controls surrounding the implementation process and the application itself were properly designed to prevent material financial statement errors. The Company's management has determined that the internal controls and procedures related to the financial reporting of fuel derivative instruments in the new system are effective as of the end of the period covered by this report.

Except as noted above, there were no changes in the Company's internal control over financial reporting (as defined in Rule 13a-15(f) under the Exchange Act) during the fiscal quarter ended June 30, 2009, that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

During the first quarter and early second quarter of 2008, the Company was named as a defendant in two putative class actions on behalf of persons who purchased air travel from the Company while the Company was allegedly in violation of FAA safety regulations. Claims alleged by the plaintiffs in these two putative class actions include breach of contract, breach of warranty, fraud/misrepresentation, unjust enrichment, and negligent and reckless operation of an aircraft. The Company believes that the class action lawsuits are without merit and intends to vigorously defend itself. Also in connection with this incident, during the first quarter and early second quarter of 2008, the Company received four letters from Shareholders demanding the Company commence an action on behalf of the Company against members of its Board of Directors and any other allegedly culpable parties for damages resulting from an alleged breach of fiduciary duties owed by them to the Company. In August 2008, Carbon County Employees Retirement System and Mark Cristello filed a related Shareholder derivative action in Texas state court naming certain directors and officers of the Company as individual defendants and the Company as a nominal defendant. The derivative action claims breach of fiduciary duty and seeks recovery by the Company of alleged monetary damages sustained as a result of the purported breach of fiduciary duty, as well as costs of the action. A Special Committee appointed by the Independent Directors of the Company has been evaluating the Shareholder demands.

The Company is from time to time subject to various other legal proceedings and claims arising in the ordinary course of business, including, but not limited to, examinations by the Internal Revenue Service (IRS).

The Company's management does not expect that the outcome in any of its currently ongoing legal proceedings or the outcome of any proposed adjustments presented to date by the IRS, individually or collectively, will have a material adverse effect on the Company's financial condition, results of operations, or cash flow.

Item 1A. Risk Factors

There have been no material changes to the factors disclosed in Item 1A. Risk Factors in our Annual Report on Form 10-K for the year ended December 31, 2008.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

(a) Recent Sales of Unregistered Securities

During the second quarter of 2009, Herbert D. Kelleher, Chairman Emeritus of the Company, exercised options to purchase unregistered shares of Southwest Airlines Co. Common Stock from the Company as follows:

Number of shares purchased	Option exercise price per share	Date of exercise	Aggregate proceeds to the Company
10,000	\$1.00	4/16/2009	\$10,000

The issuance of the above shares to Mr. Kelleher was exempt from the registration under the Securities Act of 1933, as amended (the "Act"), pursuant to the provisions of Section 4(2) of the Act because, among other things, of the limited number of participants in such transactions and the agreement and representation of Mr. Kelleher that he was acquiring such securities for investment and not with a view to distribution thereof. The issuance of such shares was not underwritten.

Item 3. Defaults upon Senior Securities

None

Item 4. Submission of Matters to a Vote of Security Holders

The Company's Annual Meeting of Shareholders was held in Dallas, Texas on Wednesday, May 20, 2009. The following matters were voted on by the Company's Shareholders at the Annual Meeting and received the following votes:

1. Proposal 1 - Election of nine Directors for one-year terms expiring in 2010.

Name of Nominee	Votes For	Votes Withheld	Broker Non-Votes
David W. Biegler	413,823,320	266,325,667	-
C. Webb Crockett	315,127,440	365,021,547	-
William H. Cunningham	410,523,787	269,625,200	-
John G. Denison	655,271,899	24,877,088	-
Travis C. Johnson	399,393,035	280,755,952	-
Gary C. Kelly	420,042,502	260,106,485	-
Nancy B. Loeffler	418,353,859	261,795,128	-
John T. Montford	399,077,148	281,071,839	-
Daniel D. Villanueva	646,557,851	33,591,136	-

2. Proposal 2 – A proposal to approve the Southwest Airlines Co. Amended and Restated 1991 Employee Stock Purchase Plan.

Votes For	Votes Against	Abstentions	Broker Non-Votes
586,195,297	3,947,529	3,843,304	86,162,857

3. Proposal 3 – A proposal to ratify the selection of Ernst & Young LLP as the Company's independent auditors for the fiscal year ending December 31, 2009.

Votes For Votes Against Abstentions Broker Non-Votes 667,882,990 11,403,849 862,148 -

4. Proposal 4 – A Shareholder proposal requesting that the Board of Directors take the necessary steps to reincorporate the Company in North Dakota.

 Votes For
 Votes Against
 Abstentions
 Broker Non-Votes

 116,538,447
 474,351,834
 3,095,849
 86,162,857

5. Proposal 5 – A Shareholder proposal requesting that the Board of Directors adopt principles for health care reform.

 Votes For
 Votes Against
 Abstentions
 Broker Non-Votes

 30,446,500
 465,960,873
 97,578,758
 86,162,856

Item 5. Other Information

None

Item 6. Exhibits

a) Exhibits

- 3.1 Restated Articles of Incorporation of Southwest (incorporated by reference to Exhibit 4.1 to Southwest's Registration Statement on Form S-3 (File No. 33-52155)); Amendment to Restated Articles of Incorporation of Southwest (incorporated by reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-O for the quarter ended June 30, 1996 (File No. 1-7259)); Amendment to Restated Articles of Incorporation of Southwest (incorporated by reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-O for the quarter ended June 30, 1998 (File No. 1-7259)); Amendment to Restated Articles of Incorporation of Southwest (incorporated by reference to Exhibit 4.2 to Southwest's Registration Statement on Form S-8 (File No. 333-82735)); Amendment to Restated Articles of Incorporation of Southwest (incorporated by reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-O for the quarter ended June 30, 2001 (File No. 1-7259)); Articles of Amendment to Articles of Incorporation of Southwest (incorporated by reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-O for the quarter ended June 30, 2007 (File No. 1-7259)).
- 3.2 Amended and Restated Bylaws of Southwest, effective January 15, 2009 (incorporated by reference to Exhibit 3.1 to Southwest's Current Report on Form 8-K dated January 15, 2009 (File No. 1-7259)).
- 10.1 Southwest Airlines Co. Amended and Restated Director Severance Plan
- 31.1 Rule 13a-14(a) Certification of Chief Executive Officer
- 31.2 Rule 13a-14(a) Certification of Chief Financial Officer
- 32.1 Section 1350 Certifications of Chief Executive Officer and Chief Financial Officer
- 101.INS XBRL Instance Document (1)
- 101.SCH XBRL Taxonomy Extension Schema Document (1)
- 101.CAL XBRL Taxonomy Extension Calculation Linkbase Document (1)
- 101.LAB XBRL Taxonomy Extension Labels Linkbase Document (1)
- 101.PRE XBRL Taxonomy Extension Presentation Linkbase Document (1)

Table of Contents

(1) Furnished, not filed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

SOUTHWEST AIRLINES CO.

July 23, 2009 By /s/ Laura Wright

Laura Wright

Chief Financial Officer

(On behalf of the Registrant and in her capacity as Principal Financial

and Accounting Officer)

EXHIBIT INDEX

3.1	Restated Articles of Incorporation of Southwest (incorporated by reference to
	Exhibit 4.1 to Southwest's Registration Statement on Form S-3 (File
	No. 33-52155)); Amendment to Restated Articles of Incorporation of Southwest
	(incorporated by reference to Exhibit 3.1 to Southwest's Quarterly Report on
	Form 10-Q for the quarter ended June 30, 1996 (File No. 1-7259));
	Amendment to Restated Articles of Incorporation of Southwest (incorporated by
	reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-Q for the
	quarter ended June 30, 1998 (File No. 1-7259)); Amendment to Restated Articles of
	Incorporation of Southwest (incorporated by reference to Exhibit 4.2 to Southwest's
	Registration Statement on Form S-8 (File No. 333-82735));
	Amendment to Restated Articles of Incorporation of Southwest (incorporated by
	reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-Q for the
	quarter ended June 30, 2001 (File No. 1-7259)); Articles of Amendment to
	Articles of Incorporation of Southwest (incorporated by
	reference to Exhibit 3.1 to Southwest's Quarterly Report on Form 10-Q for the
	quarter ended June 30, 2007 (File No. 1-7259)).
3.2	Amended and Restated Bylaws of Southwest, effective January 15,
	2009 (incorporated by reference to Exhibit 3.1 to Southwest's Current Report
	on Form 8-K dated January 15, 2009 (File No. 1-7259)).
10.1	Southwest Airlines Co. Amended and Restated Director Severance Plan
31.1	Rule 13a-14(a) Certification of Chief Executive Officer
31.2	Rule 13a-14(a) Certification of Chief Financial Officer
32.1	Section 1350 Certifications of Chief Executive Officer and Chief Financial
	Officer
101.INS	XBRL Instance Document (1)
101.SCH	XBRL Taxonomy Extension Schema Document (1)
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document (1)
101.LAB	XBRL Taxonomy Extension Labels Linkbase Document (1)
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document (1)

(1) Furnished, not filed.